The Corporate Centre in Japanese Management: An Analysis of its Size from an Information-Processing Perspective

A thesis submitted to the University of Manchester for the degree of Doctor of Business Administration in the Faculty of Business Administration

1996

Kimio Kase

Manchester Business School

ProQuest Number: 10948441

All rights reserved

INFORMATION TO ALL USERS The quality of this reproduction is dependent upon the quality of the copy submitted.

In the unlikely event that the author did not send a complete manuscript and there are missing pages, these will be noted. Also, if material had to be removed, a note will indicate the deletion.



ProQuest 10948441

Published by ProQuest LLC (2018). Copyright of the Dissertation is held by the Author.

All rights reserved. This work is protected against unauthorized copying under Title 17, United States Code Microform Edition © ProQuest LLC.

> ProQuest LLC. 789 East Eisenhower Parkway P.O. Box 1346 Ann Arbor, MI 48106 – 1346







Contents	
Contents	2
List of Tables	8
List of Figures	10
Abstract	11
Declaration	12
The Author	13
Dedication	14
Acknowledgements	15
CHAPTER 1: INTRODUCTION	17
1. 1. Corporate Centre and its Size	17
1. 2. Layout of this Thesis	23
1. 3. Justification of the Central Issue in This Research	24
CHAPTER 2: INFORMATION-PROCESSING AND THE SIZE OF THE CORPORATE CENTRE	30
2. 1. Introduction	30
2. 2. Examination of Different Approaches to Organisation: a Historical Framework for the Study of	
Organisation	32
2. 2. 1. Classical management	33
Auministrative theory Scientific management	35
Structuralist school	36
Neo-classical management and organization at theory	<u>उ</u> ठ 40
Human relations school	42 15
Behavioural school	40 48
2. 2. 2. Modern management and organisation theory	50

Management science Systems theory	54 57
2. 3. Contingency Theory: the Starting Point	63
2. 3. 1. Representative works of contingency theory	64
2. 3. 2. Criticisms of contingency theory: sociological explanations	69
2. 3. 3. Relevance of contingency theory to this study Easiness of drawing testable statements from contingency theory Applicability for practising managers	74 75 77
2. 4. Information-processing Need As a Contingent Factor Influencing the Size of the Corporate Centre	79
2.4.1. Uncertainty, equivocality, and turbulence	79
2. 4. 2. The size of the corporate centre and information- processing capabilities	82
2.4.3. Information-processing and contingency theory	85
2. 4. 4. Definition of the terminology	86
2.5. Summary	87
CHAPTER 3: VERTICAL INFORMATION FLOW AND T CORPORATE CENTRE	HE 89
3. 1. Introduction	89
3. 2. The Centre's Functions and Vertical Information Flow	90
3. 3. The Size of the Corporate Centre as a Function of Vertical Information Flow	96
3. 4. Sociological Explanations of the Size of the Corporate Centre Discussed in Chapter 2	107
3. 5. Previous Researches on the Factors Explaining the Centre's Size	109
3. 6. Summary	120
CHAPTER 4: HORIZONTAL INFORMATION FLOW AN THE CORPORATE CENTRE	D 123
4. 1. Introduction	123
4. 2. Network Formation for Information Flow	124

Complex ties and cross-cutting access	127
4. 3. Network Formation for Information Flow: Japanese	• .
Management	128
Mobility across jobs	128
Employment security	129
Teams, and complex ties and cross-cutting access	131
4. 4. Information-Processing Structure of Japanese	
Firms	134
4.4.1. Information-processing structure at the shop-	
floor level	135
Flexibility in job design	136
Inter-shop co-ordination	<i>13</i> 8
The kanban system	139
Efficiency of horizontal co-ordination	142
Centralised personnel administration	144
4. 4. 2. Generalisation of the information-processing	
structure	146
Centralised administration	14Ò
Horizontal co-ordination at middle management	147
4. 4. 3. Incentive scheme: the "ranking hierarchy"	151
Ranking hierarchy	152
Efficiency of the ranking hierarchy	154
4.5. The size of the corporate centre and horizontal	
information flow	159
4. 6. Implications of the Information-Processing and	
Information Flow Perspective for Practising Managers	161
Internationalisation process	161
Process re-engineering and corporate down-sizing	162
4.7. Summary	165
CHAPTER 5: RESEARCH ISSUES	168
5. 1. Introduction	168
5. 2. Scope of the Present Research	168
5. 3. Research Issues	170
5.3.1 First proposition	170
5. 3. 0. Second menoritien	170
5. 5. 2. Second proposition	175
5.4. Summary	174
CHAPTER 6 : RESEARCH DESIGN	176
6.1. Introduction	176
4	

6.2. Type of Research	176
6. 3. Population and Sampling Frame	179
6. 3. 1. Population	179
6. 3. 2. Sampling Frame	181
6. 4. Data Gathering Procedure	184
6. 5. Respondent Firms (Sample) The respondent firms Classification by total assets, turnover and full-time	189 <i>190</i>
employees	192
Industry classification	192 102
Comparison of industry classification and production technology	193 l 104
6 6 Penresentativeness of the respondent firms	· 105
6.7 Instrument for Data Gathering	195
6.8 Scales for the Measurement of Variables	197
6. 9 Data Analysis	199
6 10 Limitations of the Methodology Hood	204
6 11 Summons	200
0. 11. Summary	200
CHAPTER 7: PRESENTATION AND ANALYSIS OF T DATA	HE 210
7. 1. Introduction	210
7. 2. Descriptive Analysis of the Respondent Firms	213
7. 2. 1. The size of the corporate centre	213
7.2.2 Environmental elements	215
Business Environment	215
Business Environment Diversification Production Technologies	215 216 217
Business Environment Diversification Production Technologies The size of the corporate centre and environmental elements	215 216 217 218
Business Environment Diversification Production Technologies The size of the corporate centre and environmental elements 7. 2. 3. Organisational structure	215 216 217 218 220
Business Environment Diversification Production Technologies The size of the corporate centre and environmental elements 7. 2. 3. Organisational structure Divisional form	215 216 217 218 220 221
Business Environment Diversification Production Technologies The size of the corporate centre and environmental elements 7. 2. 3. Organisational structure Divisional form Functions performed by divisions	215 216 217 218 220 221 221 221
Business Environment Diversification Production Technologies The size of the corporate centre and environmental elements 7. 2. 3. Organisational structure Divisional form Functions performed by divisions Functions performed by the corporate centre The size of the corporate centre, and divisional and	215 216 217 218 220 221 221 221 223
 Business Environment Diversification Production Technologies The size of the corporate centre and environmental elements 7. 2. 3. Organisational structure Divisional form Functions performed by divisions Functions performed by the corporate centre The size of the corporate centre, and divisional and corporate functions 	215 216 217 218 220 221 221 223 225
 Business Environment Diversification Production Technologies The size of the corporate centre and environmental elements 7. 2. 3. Organisational structure Divisional form Functions performed by divisions Functions performed by the corporate centre The size of the corporate centre, and divisional and corporate functions 7. 2. 4 Summary of the descriptive analysis 	215 216 217 218 220 221 221 223 225 226
 Business Environment Diversification Production Technologies The size of the corporate centre and environmental elements 7. 2. 3. Organisational structure Divisional form Functions performed by divisions Functions performed by the corporate centre The size of the corporate centre, and divisional and corporate functions 7. 2. 4 Summary of the descriptive analysis 7. 3. Tests of the Hypotheses 	215 216 217 218 220 221 221 223 225 226 227

Hypothesis 1	228
Hypotheses 2 and 3	233
7. 3. 2. Second proposition	238
7. 4. Recapitulation and Evaluation of the Analysis	
Results	244
7. 5. Illustration of the Findings with Case Materials	245
7. 5. 1. Relationship between the corporate centre and	
shared customers Kawasaki Haguy Industries	247
7 5 0 Uigh planning and high control influences by	241
arge corporate centres	251
Matsushita Electric Industrial	252
7. 5. 3. Relationship between the corporate centre and	
shared product development	265
Kao Corporation Sharp Corporation	266
7.6. Companian with atlian analysis 1 studies	270
<i>Child</i> (1973c)	274 275
Cresap (1988)	279
Young (1995)	281
Discussions	282
•	
7.7. Summary	284
7.7. Summary CHAPTER 8: CONCLUSIONS AND IMPLICATIONS	284 287
7.7. Summary CHAPTER 8: CONCLUSIONS AND IMPLICATIONS 8.1. Introduction	284 287 287
 7. 7. Summary CHAPTER 8: CONCLUSIONS AND IMPLICATIONS 8. 1. Introduction 8. 2. Main Conclusions 	284 287 287 288
 7. 7. Summary CHAPTER 8: CONCLUSIONS AND IMPLICATIONS 8. 1. Introduction 8. 2. Main Conclusions 8. 2. 1. The size of the corporate centre and linkages 	284 287 287 288 289
 7. 7. Summary CHAPTER 8: CONCLUSIONS AND IMPLICATIONS 8. 1. Introduction 8. 2. Main Conclusions 8. 2. 1. The size of the corporate centre and linkages Two issues related to the first conclusion: centre size and 	284 287 287 288 289
 7. 7. Summary CHAPTER 8: CONCLUSIONS AND IMPLICATIONS 8. 1. Introduction 8. 2. Main Conclusions 8. 2. 1. The size of the corporate centre and linkages Two issues related to the first conclusion: centre size and linkages, and the nature of linkages 	284 287 287 288 289 289
 7. 7. Summary CHAPTER 8: CONCLUSIONS AND IMPLICATIONS 8. 1. Introduction 8. 2. Main Conclusions 8. 2. 1. The size of the corporate centre and linkages Two issues related to the first conclusion: centre size and linkages, and the nature of linkages 1) The centre's size and linkages 2) Linkages: shared product development and shared 	284 287 287 288 289 289 290
 7. 7. Summary CHAPTER 8: CONCLUSIONS AND IMPLICATIONS 8. 1. Introduction 8. 2. Main Conclusions 8. 2. 1. The size of the corporate centre and linkages Two issues related to the first conclusion: centre size and linkages, and the nature of linkages 1) The centre's size and linkages 2) Linkages: shared product development and shared customers 	284 287 287 288 289 289 290 291
 7. 7. Summary CHAPTER 8: CONCLUSIONS AND IMPLICATIONS 8. 1. Introduction 8. 2. Main Conclusions 8. 2. 1. The size of the corporate centre and linkages Two issues related to the first conclusion: centre size and linkages, and the nature of linkages 1) The centre's size and linkages 2) Linkages: shared product development and shared customers 8. 2. 2. Diversification, divisional functions and 	284 287 287 288 289 289 290 291
 7. 7. Summary CHAPTER 8: CONCLUSIONS AND IMPLICATIONS 8. 1. Introduction 8. 2. Main Conclusions 8. 2. 1. The size of the corporate centre and linkages Two issues related to the first conclusion: centre size and linkages, and the nature of linkages 1) The centre's size and linkages 2) Linkages: shared product development and shared customers 8. 2. 2. Diversification, divisional functions and corporate functions 	284 287 287 288 289 289 290 291 291
 7. 7. Summary CHAPTER 8: CONCLUSIONS AND IMPLICATIONS 8. 1. Introduction 8. 2. Main Conclusions 8. 2. 1. The size of the corporate centre and linkages Two issues related to the first conclusion: centre size and linkages, and the nature of linkages 1) The centre's size and linkages 2) Linkages: shared product development and shared customers 8. 2. 2. Diversification, divisional functions and corporate functions Three issues related to the second conclusion (diversification number of compariso and number 	284 287 288 289 289 290 291 292
 7.7. Summary CHAPTER 8: CONCLUSIONS AND IMPLICATIONS 8. 1. Introduction 8. 2. Main Conclusions 8. 2. 1. The size of the corporate centre and linkages Two issues related to the first conclusion: centre size and linkages, and the nature of linkages 1) The centre's size and linkages 2) Linkages: shared product development and shared customers 8. 2. 2. Diversification, divisional functions and corporate functions Three issues related to the second conclusion (diversification, number of corporate functions, and number of divisional functions) 	284 287 287 288 289 289 290 291 292
 7.7. Summary CHAPTER 8: CONCLUSIONS AND IMPLICATIONS 8. 1. Introduction 8. 2. Main Conclusions 8. 2. 1. The size of the corporate centre and linkages Two issues related to the first conclusion: centre size and linkages, and the nature of linkages 1) The centre's size and linkages 2) Linkages: shared product development and shared customers 8. 2. 2. Diversification, divisional functions and corporate functions Three issues related to the second conclusion (diversification, number of corporate functions, and number of divisional functions) 1) Diversification 	284 287 287 288 289 290 291 292 292 292
 7.7. Summary CHAPTER 8: CONCLUSIONS AND IMPLICATIONS 8. 1. Introduction 8. 2. Main Conclusions 8. 2. 1. The size of the corporate centre and linkages Two issues related to the first conclusion: centre size and linkages, and the nature of linkages The centre's size and linkages Linkages: shared product development and shared customers 8. 2. 2. Diversification, divisional functions and corporate functions Three issues related to the second conclusion (diversification, number of corporate functions, and number of divisional functions) Diversification Number of corporate functions 	284 287 288 289 289 290 291 292 292 292 293 294
 7.7. Summary CHAPTER 8: CONCLUSIONS AND IMPLICATIONS 8. 1. Introduction 8. 2. Main Conclusions 8. 2. 1. The size of the corporate centre and linkages Two issues related to the first conclusion: centre size and linkages, and the nature of linkages 1) The centre's size and linkages 2) Linkages: shared product development and shared customers 8. 2. 2. Diversification, divisional functions and corporate functions Three issues related to the second conclusion (diversification, number of corporate functions, and number of divisional functions) 1) Diversification 2) Number of corporate functions 3) Number of divisional functions 	284 287 288 289 289 290 291 292 292 292 293 294 294
 7.7. Summary CHAPTER 8: CONCLUSIONS AND IMPLICATIONS 8. 1. Introduction 8. 2. Main Conclusions 8. 2. 1. The size of the corporate centre and linkages Two issues related to the first conclusion: centre size and linkages, and the nature of linkages The centre's size and linkages Linkages: shared product development and shared customers 8. 2. 2. Diversification, divisional functions and corporate functions Three issues related to the second conclusion (diversification, number of corporate functions, and number of divisional functions) Diversification Number of corporate functions Number of divisional functions S. 2. 3. Planning and control influences 	284 287 288 289 289 290 291 292 292 293 294 294 294 294

1) The involvement of the centre in operations 2) Trade-off between planning and control influences	299 301
8. 3. Connection of the Results of this Research with the Literature	302
8.3.1. Linkages and the size of the corporate centre	302
8.3.2. Strategic management in Japanese firms	303
8. 3. 3. Information-processing and the corporate centre 8. 3. 4. Considerations of sociological explanations	305 <i>306</i>
8.3.5. Contribution of this research to management study	309
8. 4. Implications and Prescriptions for Corporate	212
Management	313
8. 4. 0. Implications for the intermetation of Israe	313
management	317
8. 5. Limitations of the Present Research	319
8. 6. Suggestions for Future Research	321
Bibliography	323
Annex 1a:	364
Annex 1b:	366
Annex 2a:	368
Annex 2b:	376

. 7 •

•

•

.

•

List of Tables

Table 1. 1.: Mean financial statement ratios for US andJapanese companies for fiscal years 1985 to 1988	21
Table 3. 1.: Key Factors influencing the size of the corporate centre	110
Table 3. 2.: Basic characteristics of the corporate centre units in six firms	112
Table 3. 3.: Four types of the corporate centre	118
Table 6. 1.: Sampling frame	183
Table 6. 2.: Information on the firms in the sampling frame	184
Table 6. 3.: Respondent firms	186
Table 6. 4.: Departments and posts of respondents (All responses)	188
Table 6. 5.: Respondent firms	191
Table 6. 6.: Information on the respondent firms	192
Table 6. 7.: Respondent firms by industry	193
Table 6. 8.: Routineness of production technology	195
Table 6. 9.: t-test for the sampling frame and the respondentfirms	196
Table 6. 10.: Measurement and operationalisation of the keyvariables	202
Table 7. 1.: Data on the size of the corporate centre	214
Table 7. 2.: Stability of business environment	216
Table 7. 3.: Diversification	217
Table 7. 4.: Production technologies	218
Table 7. 5.: Correlation between stability, diversification, and the size of the corporate centre	219
Table 7. 6.: Multiple regression on the size of the corporate centre and diversification	220

,

• .

221
222
224
225
226
227
229
232
233
235
237
239
241
243
297

List of Figures

Figure 2. 1.: Evolution of management and organisation theory	33
Figure 2. 2.: Framework of this research	86
Figure 3. 1.: Strategic management styles	101
Figure 3. 2.: Relationships between strategic management styles and co-ordinations	106
Figure 4. 1.: Comparison of the information and personnel management between Japanese and American firms	150
Figure 5. 1.: General framework and scope of the research	169
Figure 5. 2.: Relationship of the hypotheses	175
Figure 6. 1.: Production technology	194
Figure 7. 1.: Scattergram for planning and control influences	230
Figure 7. 2.: Planning and control influences	230
Figure 7. 3.: Scattergram for the size of the corporate centre and planning influence	234
Figure 7. 4.: Scattergram for the size of the corporate centre and control influence	236

Abstract

THE UNIVERSITY OF MANCHESTER

ABSTRACT OF THESIS submitted by Kimio Kase

for the Degree of Doctor of Business Administration and entitled "the Corporate Centre in Japanese Management: An Analysis of its Size from an Information-Processing Perspective"

Month and Year of Submission:

This thesis addresses the relationship between the size of the corporate centre, and the information-processing needs and capability resulting from vertical and horizontal information flows in Japanese firms.

It is contended that the corporate centre discharges the functions of planning and allocation of resources, control of performance, and central service provision. Of these three, the corporate centre uses planning and control functions to manage the organisation. Planning and control functions are found to be closely related to the vertical information flow.

A review of the literature on Japanese management demonstrates that Japanese firms are equipped with elements facilitating the creation of horizontal information flow. Linkages among divisions such as shared clients, production technology, etc., are discussed to determine the horizontal information flow. It is held that the horizontal information flow reflected in linkages is associated with the centralisation of administration.

The propositions and hypotheses derived from the foregoing argument have been empirically tested on the basis of fifty-five valid responses to the mail questionnaire. These fifty-five Japanese firms represent eight manufacturing industries.

The results obtained by multivariate analysis techniques, mainly, multiple regression analysis, indicate that vertical information, as measured in terms of planning and control influences, is not associated with the size of the corporate centre, whereas horizontal information flow as measured in terms of linkages, is significantly associated with it. The total number of employees and the number of divisional and corporate functions are found to be associated with the size of the corporate centre, as well. Japanese firms are also found to exert both high planning and high control influences to lower management, irrespective of industries.

Declaration

No portion of the work referred to in the thesis has been submitted in support of an application for another degree or qualification of this or any other university or other institute of learning.

The Author

Degrees

Bachelor of Arts in Foreign Studies (Spanish Studies), Tokyo University of Foreign Studies, Japan

Licenciado en Ciencias Empresariales, ICADE, Spain

Master in Business Administration, IESE, University of Navarra, Spain

Research Experience

Joint research on the corporate centre of Japanese firms — Manchester Business School, Ashridge Strategic Management Centre, University of Kobe, and Japan's Ministry of International Trade and Industry (1988/89)

Dedication

£

To Mercedes and my mother and

to the loving memory of Gabriel and my father.

Acknowledgements

I would like to express my gratitude to Dr Nigel Campbell, supervisor of this dissertation, for his support. His advice guided this research work, and his stimulating comments helped me to make my ideas clearer.

My heart-felt thanks also go to my former and present colleagues at the Doctoral Programme of Manchester Business School, especially to Dr Hernán Riquelme and Dr James Yu-Shan Liu for their kind advice and comments on the draft chapters. Without their help the writing of the present dissertation would have been a task difficult to achieve.

Manchester Business School has made it possible for me to carry out this research, above all, by allowing me to re-register for the doctoral programme after some years of interruption. Pauline Rashid and ' Barbara Beeby, the past and current Doctoral Programme Managers, should also be mentioned with my gratitude. Professor Gerry Lawson provided me with useful advice.

T Higashira of Price Waterhouse reviewed the translation of the questionnaire in Japanese. Three Japanese managers test-answered it. They include T Katsuta of Toshiba Corporation, K Kawaguchi of Taisho Pharmaceutical and T Yoshioka of Nissan Motor. Likewise, H Ikebe, I Inoue and K Motomura, all of whom are the second year MBA students at Manchester Business school test-answered the questionnaire. Additional measure of acknowledgement is due to John Sharp, who helped me to improve my writing in English. Price Waterhouse merits my special thanks for partly financing a research project which ultimately led to the present research work.

My final acknowledgements and thanks are to be addressed to Mercedes, my wife, who has accompanied me in my stay in Manchester and has held out her supporting hand to me whenever I needed it.

Manchester, June, 1994.

CHAPTER 1: INTRODUCTION

1. 1. Corporate Centre and its Size

It is commonly held that excessive overheads¹ in terms of staff should be avoided as they result in lower efficiency and performance. They mean complex systems and structures. As the number of overhead people increases arithmetically, the number of possible interactions among them increases geometrically. To cope with the complex communication needs generated by the size of overheads, complex systems are required (Peters and Waterman, 1982).

As a consequence, the organisation becomes paralysed, because the complex structure not only makes priorities less clear, but also automatically dilutes them. This leads to tardiness in responding to fast-changing conditions in the environment and the issues posed by the omnipresent existence of convoluted conditions. Every important strategy decision must go through several reviews. Middle management tends to stop ideas coming down and going up. Staff at the corporate centre tend to be in the home office checking on things rather than in the field solving problems.

Excessive overheads also mean higher indirect costs. Both the service and manufacturing sectors in Japan have 15 to 20 per cent excess

¹Overheads are "expenses arising from management, office supplies, interest on borrowed capital, etc., as distinct from business transactions (The Concise Oxford Dictionary of Current English, Sixth Edition: 785)".

white-collar work force (Hori, 1993). Traditionally, excess white-collar costs were subsidised by blue-collar labour productivity growth. Recently, however, the blue-collar subsidy to white-collar costs has been weakening on account of the levelling-out of productivity growth and of the appreciation of the yen, which drives production offshore² (Hori, 1993).

The issue of overheads has also been linked to corporate centre organisation, which comprises top management, planning and control staff, special staff like personnel, purchasing and accounting, and marketing staff. Peters and Waterman (1982) argue that the 'excellent' companies they studied have comparatively few people at the corporate level. The corporate centres of these companies have shed all the main functions, including product development, finance and personnel to lower management levels. Simplicity of form and lean staff at the corporate centre are concomitant in 'excellent' companies (Peter and Waterman, 1982).

However, in analysing whether there is excess in corporate personnel it is necessary to assess what value the corporate centre adds to the firm. Lorenz (1993) warns against the current trend in corporate management to cut the central personnel on the sheer ground that their services can be contracted out and the assumption that the resultant gain to the corporate centre by its down-sizing is always in the interests of shareholders. He asserts that indiscriminate

²The yen rose from $\frac{252}{\$1}$ at the beginning of 1985 to $\frac{125}{\$1}$ by the end of 1988. In June, 1994 the exchange rate is $\frac{101}{\$1}$.

streamlining of the corporate centre and its staff might be the best way to create confusion.

That it is unwise to "streamline" the corporate centre is supported by Japanese experience. There has always been a large staff at Japanese firms' corporate centre, regardless of whether the organisational structure is functional or divisional (Kono, 1984). The approximate number of employees in the corporate centre is, in the case of Toyota — 4000, Matsushita — 2000, Hitachi — 2000, Canon — 1500, whereas the corporate centres of ICI and GEC of UK accommodate only about 100 persons, and GE, Lockheed, and Northrop of USA may have fewer than 300 people in the corporate centre in Japanese firms is approximately 1000 persons, and the average percentage of the total number of employees is about 9 per cent according to the survey of forty-four Japanese manufacturing companies carried out by Kono in 1982 (Kono, 1984)³.

Moreover, this situation has been further accentuated in recent years (Morikawa, 1993). Japan's industries, for example, electronics, automobiles, semiconductors, telecommunications, and so on, have strengthened their corporate and headquarters functions. Between 1984 and 1992 these industries saw overhead expenses—sales, general and administrative expenses—increase on average from less than 14 per cent of sales to 17 per cent (Hori, 1993). The average number of

³In this survey, the average size of the corporate centre was 638, and 9.1 per cent of the total employees.

board members in Japanese firms increased from around 10 in 1954 to 20 in 1975. In some firms such as Nippon Steel or Mitsui & Co., the number of board members is as high as fifty people (Morikawa, 1993).

The foregoing argument may suggest that the overhead costs would make the Japanese profit margin lower than that in the West. However, Brown et al (1993) contend that the profit margin does not differ significantly between US and Japan. They found that Japanese firms have a higher percentage of cost of goods sold, which appears to be consistent with a low pricing strategy to achieve high market share. The lower percentage of selling and administrative expense for Japan offsets its higher percentage of cost of goods sold and equalises its profit margin with US companies (See Table 1. 1.). Japanese companies do not disaggregate selling and administrative expenses further, which prevents additional insights into their lower expenses. Brown et al (1993) surmise that the Japanese emphasis on human resources (open communications, work groups, consultative decision making) increases employee trust and reduces the need for elaborate control systems and their related costs. Thus, Japanese firms disagree with the commonly held conception of oversized staffing and the assumption that the value added by the corporate centre to the company management does not compensate for the cost it involves.

 Table 1. 1.: Mean financial statement ratios for US and Japanese companies for fiscal years 1985 to 1988

•	US	Japan
COST OF GOODS	0.65	0.75*
SOLD/SALES (%)		
SELLING & ADMIN.	0.28	0.19*
EXP./SALES (%)		
PROFIT MARGIN	3.2	3.3
(%)		

Note: An asterisk to the right of the mean Japanese ratio indicates the level of significance (p = .01) at which it differs from the mean US ratio. Source: Adapted from Brown *et al* (1994: 78)

Some authors have argued that the size of the corporate centre is positively associated with information-processing needs and capabilities (Haleblian and Finkelstein, 1993). Information-processing in an organisation is defined as including the gathering of data, the transformation of data into information, and the communication and storage of information in the organisation (Galbraith, 1973; Tushman and Nadler, 1978). Information-processing needs arise when there is an information gap, which means the difference between the amount of information required to perform the task and the amount of information already possessed by the organisation (Tushman and Nadler, 1978). Galbraith (1973) argues that information-processing capabilities relate to vertical and horizontal information flows, and that they depend on the educational level of the human resources, the management process employed in the organisation and the infrastructure (e.g., IT, MIS) through which this information is communicated.

Information-processing needs are related to the intensity of the corporate centre's intervention in decision-making processes in lower management (which might be called vertical information flow) and to the extent of cross-business co-ordination (which might be called horizontal information flow). This means that there should be a significant relationship between the size of the corporate centre and the vertical or horizontal information exchange between it and lower management⁴.

Therefore, it is necessary to investigate how information flows. This is seen in the way the corporate centre interacts with lower management (the vertical information flow), and the way it organises the organisational overlaps or "linkages" (which are, for example, shared clients, common product development, shared production technology, etc.) relating to the horizontal information flow. Unless value in the form of information-processing is added to company management by the corporate centre in this interaction and organisation, the corporate centre may be small.

Thus the focus of this research is directed to the relationship between the information flows (vertical and horizontal) and the size of the corporate centre.

⁴There are several essential issues in the above relationship existing between information flow and the size of the corporate centre. Technology, location and environment variables (Child, 1973), organisational interdependence (Aiken and Hage, 1968), work characteristics and sub-unit communication (Tushman, 1979), qualifications of personnel (Blau, 1968), corporate linkages (Boyd, 1990), etc. have all been studied as factors influencing organisation structures and size.

1. 2. Layout of this Thesis

To address the central issue of this research, namely, the relationship between the size of the centre and vertical and horizontal information flows, the later chapters are organised as follows:

Chapter 2 reviews literature on organisation in search of factors influencing organisation, especially the size of the corporate centre. Information-processing is identified as a key factor influencing the size of the corporate centre.

Chapter 3 discusses the relationship between the size of the corporate centre and the vertical information flow. That information flow links the corporate centre and lower management.

Chapter 4 complements the foregoing discussion and addresses the relationship between the size of the corporate centre and the horizontal and inter-divisional or inter-departmental information flow in Japanese firms.

Chapter 5 identifies the issues for the research and specifies them as four hypotheses, for the testing of which an empirical survey has been conducted on the basis of a mail questionnaire to major Japanese manufacturing firms.

Chapter 6 sets out the design and the methodology of the research. It describes in detail the type of research, the population, sampling frame, sample size, procedures for the gathering of data, instruments of data gathering, scales for measurement of variables, techniques for analysing the data, and the limitations of the methodology used.

Chapter 7 provides a statistical analysis of the data, and furnishes the findings of the study and discussion. Wherever possible, reference is made to the connection of the results of the research with the existing literature.

Chapter 8 submits the conclusions and implications for both academics and practising managers. Limitations of the research are also mentioned.

1. 3. Justification of the Central Issue in This Research

Holden and Burgess (1994: 4) note:

Japanese management practice, no matter how poorly understood in other countries, has become a universal byword for innovation, product quality, excellence in manufacturing and harmonious industrial relations.

Therefore, this research addresses Japanese management practice with a focus, as shown in the foregoing lay-out, upon the strategic management of Japanese firms in relation to their organisational design, with strategic management defined as the formulation,

implementation, and evaluation of managerial actions that enhance the value of a business enterprise (Teece, 1990).

Based on the argument in Section 1. 1, the research attempts to explain the relationship between information flows and the size of the corporate centre. It should be pointed out, however, that in formulating this "functionalist" relationship between the two variables, namely, information flows and the size of the corporate centre, alternative approaches for their explanation are not excluded.

These approaches relate to wider and sociological aspects of organisational behaviour such as political process, social constructions and dysfunctional uses of bureaucracy (Child, 1973b). They are discussed in the literature review in Chapters 3 and 4, and in Chapter 8. Two limitations that the approaches to these wider aspects imply for this research may be pointed out.

First, the secretive nature of Japanese firms makes the gathering of the information on them difficult. Section 6. 4, shows the reasons given by Japanese firms for declining to answer the questionnaire for this research.

The second limitation refers to the lack of objective measurements of the wider aspects of organisational behaviour. In consequence, a limited number of case studies or secondary data will be looked for. They will help us to understand the dynamic process at the corporate centre of Japanese firms reflecting a larger institutional environment which constrains Japanese management. This will be treated in Sections 7. 5. and 8. 3.

The focus of this research on Japanese management in relation to the size of the corporate centre and information flows may be justified on three grounds. First, despite the importance of the corporate centre in strategic management, there are few accepted theories of, and even less empirical research on, the corporate centre. Questions such as what the distinctive tasks of the centre are in strategic management and how the corporate centre relates to lower level management remain relatively unapproached by researchers. Techniques like those of portfolio planning are used in some firms (Haspeslagh, 1982), but are regarded, at best, as a useful aid rather than as a complete guide. There seems to have been little research on the corporate centre which has proved of practical value to senior managers (Goold and Campbell, 1987).

Some of the few researchers who have investigated the issues related to the corporate centre are Chandler, Williamson, and Goold and Campbell. Chandler (1962) studied both the emergence of divisional structure and the corporate centre detached from daily operations and essentially engaged in strategy formulation. Williamson (1975) argued that the corporate centre carries out functions that cannot be done as efficiently by the capital market. Goold and Campbell (1987) have focused their attention on two aspects of the role played by the corporate centre. The first connected with planning and allocating resources sets policy as a framework for guiding co-ordination and

operating decisions, appraising plans, and appraising resource requests. The second controls and audits performance.

Second, Japanese corporate centres have a large staff compared with their counterparts in US firms. However, as argued by Brown *et al* (1993), the overall performance of Japanese firms is not impaired by the seemingly heavy load of indirect staff at the corporate centre. The centre adds value to company management which compensates for its cost. It is therefore interesting to study what factors influence the size of the corporate centre and how the centre contributes value. Thus, in delving into the empirical aspect of the size of the corporate centre in Japanese firms, the research seeks to provide some, useful guidance for the organisational design for practising managers.

Third, in the face of today's intense competition on business organisations, firms seek drastic improvement in their key processes through process innovation (Davenport, 1993) or process re-engineering (Hammer, 1990). Japanese firms discovered and implemented process management long before the West, and it has enabled firms in a number of industries to develop efficient processes in key areas (Davenport, 1993). As processes and information flow are closely linked (Davenport, 1993), this research may help us to understand the implications of process innovation for the organisational design.

Likewise, the information flow perspective of this research may cast some light on the growing awareness in the strategic management literature that strategy formation is an emergent process (Mintzberg,

1990a, 1990b), since Kagono *et al* (1985) contend that the strategy formation in Japanese firms is process-based and, therefore, hinges on information flows.

However, for the gathering and processing of these information flows "contextual" skills, namely, familiarity with the firm's way of doing business, are required. This fact must make it difficult for Japanese firms to "internationalise" their management structure and to transfer their production technology, like lean production technology, to their overseas operations.

In summary, from the viewpoint of practising managers, regarding the centre's size and information flows the questions begging to be addressed are:

- why the Japanese corporate centre is large;
- what value the centre adds to compensate for its cost;
- what elements corporate restructuring should take into account;
- what the difficulties could be for Japanese firms' internationalisation process; and
- what could be the difficulties for the transfer of lean production technology.

They will be addressed in Chapter 8, based on the results of the research.

With regard to the characteristics of the research, suffice it to say that, rather than being exploratory or descriptive (Green and Tull, 1978;

Bailey, 1982), it seeks to propose causal relationships formulated as hypotheses. The empirical evidence for such causal relationships has been sought by means of a mail questionnaire sent to the corporate centres of 385 Japanese firms in Japan.

CHAPTER 2: INFORMATION-PROCESSING AND THE SIZE OF THE CORPORATE CENTRE

2. 1. Introduction

The preceding chapter identified the starting point of this research and defined its purpose. It attempts to analyse the factors influencing the size of the corporate centre (top management, planning and control staff, personnel, purchasing, accounting, etc., in the central organisation) in Japanese firms. It seeks to explicate the relationship between information flows resulting from the way the corporate centre manages the firm, and the size of the corporate centre.

Chapters 2, 3, and 4 examine existing literature. The first step in this chapter reviews organisation theory in search of explanations regarding the factors influencing organisation structure, especially the size of the corporate centre. The contingency approach is adopted as a point of departure. Criticism of contingency theory and alternative explanations are reviewed. Information-processing is deemed to be a contingent factor influencing the size of the corporate centre. Thereafter, in Chapter 3, there is a review of what authors have researched and written about the corporate centre and the vertical information flow. Taken in conjunction with the analysis contained in Chapter 4 on the horizontal information flow, this provides a solid basis for advancing to the identification of research issues in Chapter 5.

This chapter is divided into five sections. These revolve around the search for the factors influencing organisation structure, especially the size of the corporate centre.

Section 2. 2 examines different approaches to organisations within the context of historical framework. Two large groups of approaches are studied. First, classical management theory is reviewed. It includes administrative theory, scientific management, the structuralist school, and neo-classical management and organisational theory. This last includes the human relations school and the behavioural school. Second, modern management and organisation theory is reviewed. Management science and systems theory are studied in this category.

As an outgrowth from systems theory, contingency theory is reviewed in Section 2. 3. Its representative works are examined. Criticisms of contingency theory are discussed. Relevance of the theory to this study is analysed in the light of (1) the testability of the statements based on contingency theory and, therefore, its appropriateness for empirical studies, and (2) the applicability of its results for practising managers.

Contingent factors influencing organisational structure in general and the size of the corporate centre in particular are examined in Section 2. 4. It examines the need for information-processing as a factor influencing organisation structure and the size of the corporate centre. Theories on communications process are reviewed in relation to information-processing need.

Section 2. 5 summarises the discussion in Chapter 2.

2. 2. Examination of Different Approaches to Organisation: a Historical Framework for the Study of Organisation

Organisation theory is "the body of thinking and writing which addresses itself to the problem of how to organise. More specifically, it is the study of the structure, functioning and performance of organisations and the behaviour of groups and individuals within them." (Pugh, 1984: 9)

Organisation theory considers the behaviour of individuals, but in the aggregate. It is distinct from organisation behaviour, which is the micro approach to organisations, because the latter focuses on the individuals within organisations as the relevant units of analysis. Organisation behaviour examines concepts such as motivation, leadership style, and personality and is concerned with cognitive and emotional differences among people within organisations (Daft, 1995). The focus of the dissertation is on organisation theory.

The development of organisation theory and practice can be analysed by examining the evolution of society as it has passed from an agrarian through an industrial to a post-industrial structure. The rationale underlying this historical perspective emerges from an area of study within sociology referred to as the *sociology of knowledge*. This field study holds that various theories and explanations of the world are in some measure social products that

are influenced by wider cultural, economic, and technological forces (Bowditch and Buono, 1994).

Figure 2. 1. summarises the evolution of management and organisation theory.

Figure 2. 1.: Evolution of management and organisation theory

Period Nature of society Locus of work	1800s <u>Agrarian</u> Farm/home	1880-1930	1930-1960 Industrial Factory	1960s	1970-1990s Post-industrial Office
Nature of work	Physical: agriculture/crafts	Physica	al: manufacturing	Social: services	Abstract: computer mediated
Management and organisation theory	Prescientific	Classical Administrative Theory	<u>Neoclassical</u> Human Relations	Modem Management Science	
2*		Scientific Management	Behavioural	н	Systems Theory Contingency
_		Siructuralists			Theory
Focus of management control	Patterns of physic precise ways of	al movement and performing work	Patterns of so	cial interaction	Patterns of attention
Role of management	Control of employed	e behaviour	Maintain employee	e social system	Facilitate employee development

Source: Adapted from Bowditch and Buono (1990:8)

2. 2. 1. Classical management

Towards the end of the nineteenth century and into the beginning of the twentieth, the social, economic, and technological environment began to create the necessary conditions for the systematic study of management. An Industrial Revolution, initiated by the inventions and technological improvements of the eighteenth century, led to changes not only in the work role, but in the very nature of our society as well. Conceptually, such industrialisation occurs through two basic stages: (1) the development of an industrial infrastructure and (2) the creation of a capital goods sector. Essentially, an *industrial infrastructure* consists of such things as (a) nation-wide systems of transport, (b) sources of relatively cheap power, *e.g.*, coal and oil, (c) technological innovations (such as cotton gin and the steam engine), (d) modern communication systems, (e) networks of financial institutions, (f) preparation of the labour force through education and technical training (Etzioni, 1980).

The process of industrialisation changed the nature of work from farming and crafts to more complex forms of manufacturing and working with machines. It also led to the urbanisation of our society as the industrial process transferred the locus of work from the farm and the rural home to the factory and the city. As plants and factories were built in centralised locations, people began to migrate from rural areas to where the "new" jobs were located. Immigrants created urban enclaves in areas near the factories as well. As a result, the business firms and complex division of labour which evolved from this transition further prompted efforts to study seriously these new forms of organisation and their management.

The Classical school of management and organisation theory which evolved during this period was, thus, influenced by the wider social, economic, and technological forces that precipitated the transition from an agrarian to an industrial society. The dominant set of assumptions about human nature characterised people as rational, economic beings who would act to maximise their own self-interest.
Correspondingly, the focus on management practice which emerged dealt with how these new forms of industrial organisation could be structured, how work could be delegated and co-ordinated, and how people within these organisational structures could be motivated.

Administrative theory

One of the major orientations within the Classical school is referred to as Administrative Theory or the Universal Principles school of management. Based on deductive reasoning, this group examined certain forms of organisation, such as the church and the military, and concluded that there were basic dimensions of organisational structure and characteristics of management that were common to *all* organisations.

Fayol (1949) identified five basic functions of management: planning, organising, commanding, co-ordinating, and controlling. Based on these functions, he described how management should be carried out in 14 management principles which included:

(1) Division of work. This principle refers to the specialisation of tasks, and control of the number of objects under each worker or manager to improve effectiveness and efficiency.

(2) Authority and responsibility. This tenet gives the person in authority the right to give orders and the power to obtain obedience. Responsibility emerges directly from authority.

(3) Unity of command. This principle states that no one should have more than one boss.

(4) Remuneration. This principle specifies that pay should be fair and satisfactory to the employer and employee. No one should be under or over rewarded.

(5) *Esprit de corps*. This tenet proposes that morale and good feelings about the organisation are enhanced by effective face-to-face communication and team cohesiveness.

The focus of the Administrative Theory school was on the formal structure of organisations, and their top and middle management. The orientation was authoritarian, and posited one best way to approach all organisational and managerial problems. Employees were basically viewed as extensions of the organisation's structure and machinery.

Scientific management

In contrast to the emphasis on principles of organisation and management, the Scientific Management school focused on the measurement and structure of work itself. The aim of this approach was to create the most effective way to carry out work tasks (Taylor, 1987). Since people were viewed as rational economic beings, it was reasoned that they would act on their own interests (and subsequently the organisation's) if they understood "correct" work procedures and were rewarded for following them.

As practised by Taylor and his followers, Scientific Management focused on existing manufacturing and work procedures, improving and standardising them, while concurrently attempting to conserve

resources (Bowditch and Buono, 1994). His emphasis was that the real way to increase output was to "work smarter," not necessarily harder. By understanding the nature of jobs — to be deduced from study and analysis — the process of work could be made more effective. He did, however, have the needs of the individual worker at heart. In fact, he would refuse to take a factory as a client unless the owners agreed first to raise wages, sometimes even tripling them (Peterson, 1990). The four basic tenets of Scientific Management are:

(1) Develop the one best way to do each job.

(2) Select the best individual for the position.

(3) Ensure that work be carried out in prescribed fashion through training and by increasing wages for those workers who follow correct procedures.

(4) Divide work efforts among employees so that activities such as planning, organising, and controlling are the prime responsibilities of managers rather than of individual workers.

Although Taylor (1987) felt strongly that management rather than workers should choose the methods of carrying out organisational tasks, the basic dimensions of Scientific Management discussed above were qualified by an *exception principle*. Once output standards and routine work procedures were defined through the use of the scientific method and precise measurement, management should give its attention only to situation where standards or procedures were not within those norms.

In numerous instances, the gains and profits that emerged from Taylor's (1987) system were impressive; in others, such as the attempt to apply these principles to reform naval manufacturing, his efforts were deemed a failure. In these latter situations, however, the lack of success has been attributed to internal political battles, resistance to shifts in power, and a lack of incentives to become more efficient rather than Taylor's principles *per se* (Peterson, 1990).

Other developments from the Scientific Management school include Gilbreth's (1912) time and motion studies. Based on the principles of economy of motion, the Gilbreths took an analytic approach which measured body motions to discover the most efficient way to carry out a particular task. Henry Gantt, who, like Taylor, was interested in production efficiency, was another contributor to this school (Wren, 1979). In contrast to Taylor, Gantt focused on devising remuneration procedures that would provide fair pay for correctly doing the task and a bonus for completing it in a timely fashion (Wren, 1979). His main contribution is a production scheduling aid referred to as the Gantt chart, a way of plotting work in progress against the calendar.

Structuralist school

In contrast to the deductive approach taken by the Universalists in their study of organisations and management, the Structuralist school used an inductive approach in its study of organisations. Rather than creating a conceptual model of an organisational structure and then generalising to all organisations, members of this school examined many different organisations to determine empirically the common structural elements which could characterise how organisations actually operate.

Weber (1975) empirically discovered those basic aspects that characterised an ideal type of organisation, or what he referred to as a bureaucracy:

(1) Rules and procedures control organisational functions.

(2) A high degree of differentiation exists between organisational functions.

(3) Organisation of offices is hierarchically determined; no sub-unit is allowed to "drift," and each sub-unit reports directly to one higher-level office.

(4) An emphasis is placed on prescribed rules or established norms that regulate behaviour.

(5) Ownership and administration of an organisation are separate — the president does not own the company, the dean does not own the college.

(6) An administrator must not be able to control position or the "trappings" of an office.

(7) All administrative acts are recorded in writing.

Although Weber acknowledged that not all organisations were pure bureaucracies, it is clear that he thought bureaucracies were preferable to other forms of organisations. To Weber (1975), bureaucracy was one of the most characteristic and ubiquitous forms of administration in modern society. It was efficient, it consisted of (needed) specialists and experts, and since it dealt with

people on an impersonal basis it ensured that rules would be applied consistently in all situations.

Etzioni (1964) summarises Structuralists attempts to synthesise the classical and the human relations schools. It is in exploring the "harmony" view of the human relations writers that the Structuralist writers first recognise fully the organisation dilemma: the inevitable strains (1) between organisational needs and personal needs (2) between rationality and non-rationality, (3) between discipline and autonomy, (4) between formal and informal relations, (5) between management and workers, and (6) between ranks and divisions. The Structuralists see the organisation as a large, complex social unit in which social groups may interact. While these groups share some interests, they have other, incompatible interests. The various groups might co-operate in some spheres and compete in others. but they are not and could hardly become one big happy family as the human relations writers often imply. Their view is much in line with Weick's (1979) idea about designing the organisation to reduce uncertainty.

Although there were differences in the manner in which the Universalists and the Structuralists created their models of organisation, there are many aspects common to both. Overall, as an initial attempt to create a theory of organisation and management, Classical theory was a significant advance.

Classical theory advanced by such authors as Barnard (1938) and Urwick (1952) emphasised limited and prescribed communication channels, detailed role description and authoritative leadership

styles. The organisations in the classical theory coincide with the mechanistic organisations described by Burns and Stalker (1961). Mechanistic organisations break down the problems and tasks into "specialisms"; somebody at the top is responsible for seeing that one person's work is relevant to that of others, while the technical methods, duties and powers attached to each post are precisely defined (Burns, 1963). Burns and Stalker, however, contend that there are mechanistic and organic organisations.

While it is generally recognised today that Classical theory does not include important aspects of human behaviour and environmental influences on organisations and their management, the understanding of formal functions and structures that it provides is still an important part of our understanding of the visible and tangible aspects of organisations. The problem is that Classical theory fails to explain why there are differences in organisations.

While Classical management and organisation theory provides relevant insights into the nature of organisation, the value of the theory is limited by its rather narrow focus on the formal anatomy of the firm.

Neo-classical management and organisational theory

As early as the 1920s, a number of social critics began to point out the potentially harmful effects of trying to standardise people as well as jobs. Although many of the basic tenets of classical organisation and management theory (such as structure and stability, division of labour, and task specialisation) were not directly challenged, criticism was focused on those individual managers and theorists who appeared to treat employees as little more than mere appendages to machines (Koontz, 1961; Miles, 1975; Schein, 1980). When Taylor (1987) proposed his theory of Scientific Management, his work was met with antagonism and hostility. He defended his principles on the basis of a "mental revolution" that would take place in the attitudes of management and labour in which both sides would recognise the need for co-operation, and the importance of scientific investigation (rather than individual judgement and opinion) as the basis for structuring work assignments. Others, however, argued that while management might seek to standardise skills and methods, it could not expect perfectly standard, emotionless behaviour from its employees.

Studies during this period also began to draw attention to the possibility that co-workers could exert a greater influence on work behaviour than economic incentives offered by management. The recognition that workers had social needs led to a new set of assumptions about human nature. Rather than viewing people as rational and economic creatures, social considerations were now seen as the prime motivator of human behaviour and work performance. Since the mechanisation of work during the Industrial Revolution stripped jobs of much of their intrinsic meaning, people sought out meaning in their work through social relationships on the job. Management, it was argued, must therefore help people to satisfy these natural desires. Although these arguments may appear to be somewhat moralistic, they were tied to prescriptions for organisational effectiveness and efficiency. If managers did not respond to these socially oriented needs with greater consideration and warmth, lagging work performance and resistance to authority were viewed as likely outcomes.

In an attempt to compensate for the neglect of human interaction in the Classical school, Neo-classical theory introduced the behavioural sciences into management thought. The underlying rationale was that since management involves getting things done with and through people, the study of management must be centred on understanding inter-personal relations. Overall, the Neoclassical school of thought can be viewed as a critique of the Classical doctrine (Scott, 1961).

1. Each organisation should have a *structure*; however, human behaviour can disrupt the most carefully planned organisational activities. While the formal structure may represent how things are supposed to occur, the informal organisation which emerges in response to people's social needs dictates how things are actually done.

2. Although *division of labour* might make sense from the organisation's standpoint, some of the unintended outcomes

for workers are feelings of isolation and anonymity due to insignificant jobs.

3. While the scalar and functional principles might be theoretically valid, they deteriorate in practice due to the way in which these processes are carried out (*e.g.*, insufficient delegation, overlapping authority, gaps in authority).

4. Finally, a manager's *span of control* is a function of human factors and cannot be reduced to a precise, universally applicable ratio.

There are two main sources of Neo-classical theory: (1) the sociologists and social psychologists who were concerned with interaction and relations within groups, often referred to as the Human Relations school, and (2) the psychologists who focused on individual behaviour, or the Behavioural school.

Human relations school

The basic tenets of the Human Relations school (approximately 1930-1950) emerged from a group of studies during the mid-1920s and early 1930s known as the "Hawthorne Experiments". Based on the Classical theory tradition, these experiments were conducted in Western Electric's Hawthorn plant to assess the effect of working conditions on productivity (Miles, 1975; Schein, 1980). A number of related studies were undertaken which varied such aspects of the work environment as lighting, rest periods, length of the work day, and other facets of the work environment (Mayo, 1987).

Much to the surprise of the researchers, they found that, regardless of the group's productivity, output per worker increased. Since the conditions under which these individuals accomplished the work efforts did not explain the change in productivity, the researchers were forced to look at other factors in the work place. One observation was that, during the experiments, members of the work group had developed high morale which seemed positively to influence their job performance. Further analysis revealed three basic underlying factors which created the group's high morale:

1. The workers perceived themselves as "special" and important to management, because they were singled out for this research role.

2. The women developed good interpersonal relationships with each other and their supervisor, because they had considerable autonomy or freedom to decide their own division of the work and their own work place.

3. The social contact and easy relations which emerged in the group created a generally pleasant work environment.

Based on these preliminary findings, the researchers hypothesised that the nature of social relations among members of a work group, and between employees and their supervisor was an influential motivator of work performance in terms of both productivity and quality of work. In an effort to test this hypothesis, another group of studies was conducted in which three small groups of men who worked on switchboard equipment were studied to assess the relative effects of a complex wage-incentive plan. The researchers found that production norms — or standards which the group created for itself — were more powerful determinants of production than wage incentives. The workers established an implied norm of what was a "fair day's work", and those who over-produced or underproduced ("one must not be a 'rate-buster' or a 'chiseller") were brought into line (Mayo, 1987). These findings, as well as a number of other research efforts, confirmed the importance of the work group and pointed to the existence of an *informal organisation* within a firm's formal structure.

Over time these studies were popularised by a number of interpretations of the research itself and its implications for management. The Hawthorne Studies marked a significant turning point in the evolution of management theory, because they introduced a new way of thinking about people in organisations. It is not that the earlier scientific management theorists — Taylor (1987),

Gilbreths (1912), Gantt — misunderstood the human factor; workers were a factor, but they never held centre stage in this body of work (Greenwood and Wrege, 1986).

Human relations theory points to the need to understand workers as human beings if they are to become organisation members. Social processes which surround the formal management system namely, the informal organisation which is part of every organisation's infrastructure — are emphasised (Mayo, 1977; Roethlisberger and Dickson, 1939). Human relations theory also approaches organisation problems as a universal prescription for all situations and all organisations.

It is important to point out that the Hawthorne Studies and the emergence of the Human Relations school were not without their critics. A number of questions concerning research methodology and the way in which the experiment's outcomes were interpreted surfaced early on and linger to this day (Sykes, 1965). Even the basic rationale and intent of the experiments have been viewed with a degree of scepticism. For instance, the United Auto Workers quoted by Baritz (1960) was quite critical of the Hawthorne project:

At Hawthorne, Ma Bell, when she wasn't organising company unions, allowed Professor Mayo to carry out experiments with a group of women workers for some nine years...For these nine years about every kind of experiment a very bright Harvard professor could think of was tried on the women. Everything you do to white mice was done to them, except their spines and skulls were not split so the fluid could be analysed...What did make them produce and produce and produce with everincreasing speed was the expression of interest in their personal problems by the supervisor; interviews by psychiatrically trained social workers and (later on) the way they were paired off with friendly or unfriendly coworkers...Now obviously this is the greatest discovery since J.P. Morgan learned that you can increase profits by organising a monopoly, suppressing competition, raising prices and reducing production.

Such criticism, however, became submerged in the overwhelming support that was given to the Human Relations school. The emerging set of beliefs held that management could not treat people as if they were mere extensions of an organisation's structure and machinery (Miles, 1975; Schein, 1980).

Behavioural school

As an outgrowth of the interest in the Human Relations school and the focus on people *within* the work group, a growing number of theorists began to shift their attention to the individual and the nature of work itself (Miles, 1975; Schein, 1980; Bowditch and Buono, 1994). Beginning with the Neo-classical assumption that work had lost its meaning through the mechanisation accompanying the Industrial Revolution, this new school of thought argued that workers had become alienated from their work, because their jobs prevented them from fully using their skills and capabilities. While these theorists agreed that people had needs for acceptance, status, and recognition, they went beyond the human relations perspective and proposed that workers also wanted to gain personal satisfaction from their jobs by developing their abilities (and themselves) in the accomplishment of meaningful and worthwhile work (Daft, 1995).

The basic assumption underlying human behaviour were now perceived to be oriented towards personal growth, accomplishment, and inner development. If managers were to become truly effective, it was argued, they must go beyond simply providing fair pay and treatment and attempt to make organisational members feel important. In the interest of effective and efficient organisational performance, managers should develop jobs and organisational structures that allowed people an opportunity to develop their abilities and to experience personal growth (Bowditch and Buono, 1994).

This perspective is often referred to as the Behavioural school. It was developed during the early 1950s and was based on the work of such theorists as Maslow (1987), McGregor (1987), Likert (1987), Argyris (1987), Herzberg (1987), and McClelland (1987). There has been a continual development of the Behavioural perspective and it remains one of the bases of much management and supervisory education today. It can be viewed as a bridge between traditional Neo-classical theory and many of our contemporary analyses of management and organisation theory.

In summary, the Neo-classical school argues that there is a dimension of organisational reality that is not contained in the formal, structural orientation of the Classical school — the behaviour of groups and individuals within the organisation. The insights of the Neo-classical doctrine thus add to our understanding of the complexity of organisations. It has been criticised, however, for not integrating the varied facets of human behaviour which occur in organisations and, like the Classical school, for being short-

sighted and incomplete (Scott, 1961). Moreover, while the Neoclassical school began to realise that organisations were affected by a wider range of forces than previously thought, its view of relevant environmental forces is quite narrow by today's standard (Bowditch and Buono, 1994).

2. 2. 2. Modern management and organisation theory

Contemporary or "Modern" management and organisation theory is different from both the Classical and Neo-classical schools in that it emphasises conceptualisation and analysis, relies heavily on empirical research, and, perhaps most importantly, attempts to integrate the various elements that contribute to the whole organisation (Scott, 1961). The underlying theme is that organisations are systems composed of mutually interrelated and interdependent variables. This is not to suggest, however, that modern theory is a unified body of thought. Indeed, different theorists and researchers have their own particular emphases on what they consider an organisational system, or the more influential components of that system. The common thread is the attempt to analyse organisations in their totality (Bowditch and Buono, 1994).

The development of modern management and organisation theory has been influenced by a number of social, political, economic, and technological changes in the larger society. Many social observers, for example, argue that another transition from an industrial to a *post-industrial society* has begun, similar to the transitions society underwent when it evolved from an agrarian to industrial structure

(Bell, 1973). In contrast to the agriculturally and industrially based societies of earlier periods, post-industrial society is characterised by the following dimensions (Bell, 1973):

 A basic shift in the orientation of the labour force from goods-producing to services-rendering/information-processing
A gradual and steady rise in the influence of professional and technical occupations

3. A growing influence and centrality of theoretical knowledge as the source of innovation and policy formulation for society4. An increased need for the planning and control of technology and its growth

5. An emergence of mass integrated computer systems that will create a new "intellectual technology".

Although there is still some debate concerning whether our society is actually developing the type of post-industrial structure envisaged by such authors as Bell (1973), Little (1973) and Hill (1974), it is becoming clear that society is undergoing a transformation that will re-arrange the industrial and economic structure (Reich, 1991). For example, although routine production-type jobs are still found in high technology as well as in traditional, heavy industries, this type of work is increasingly being robotised and automated to the point where it currently constitutes only about 25 per cent of the US employment base. At the same time, while the high-level, symbolic and analytic-type service jobs — for instance, research scientists, design engineers, investment bankers, financial consultants, marketing strategists, and corporate planners — that prophets of the so-called information age predicted would transform the working

world have emerged, a far larger proportion of the US labour force has been relegated to lower level, in-person service jobs. These latter jobs involve the direct delivery of person-to-person services such as retail sales, hotel workers, waiters and waitresses, cashiers, home cleaning services, taxi drivers, and security guards. Combined with a general "flattening" of organisational structures, such shifts raise significant questions about our approaches to management and organisation (Bowditch and Buono, 1994).

In addition to the rapid and often volatile technological changes envisaged by post-industrialism, organisations are also confronted with myriad pressures that include: (1) growing competition in the world arena; (2) government regulations which have increasingly dealt with social controversies (such as equal opportunity, workplace safety, environmental protection); (3) resource scarcity; (4) increased labour-force diversity; (5) changing cultural norms; and (6) the more traditional business pressures from competitors, suppliers and supply and demand forces. Moreover, the growing influence of a number of new stakeholders - public interest groups, environmentalists, local community advocacy groups, consumer advocates, and so forth - further emphasises the expanding demands and expectations that are being placed on business firms today. Modern management and organisation theory acknowledges these demands and focuses on the organisation within its environment and the congruence or fit of organisational subsystems within that larger mosaic (Daft, 1995).

£

As the complexity of organisational environments began to dawn on social scientists, they began to challenge many of the basic

assumptions underlying human nature. The move towards increased empirical research produced inconsistent support for many of the generalised concepts (Bowditch and Buono, 1994). In other words, while these general theories were supported in some studies, other research efforts produced different results. The major impact of these trends has been the move towards more complicated models of human nature and the resultant implications about how to manage these individuals. A set of complex assumptions about people began to emerge (Schein, 1980).

1. Human needs fall into many categories and vary according to the individual's stage of development and total life situation.

2. Since needs and motives interact to form complex motivation patterns, it must be realised that for different people the same object or outcome (*e.g.*, money) can vary in importance and fulfil quite different needs.

3. People are capable of learning new motives through organisational experiences.

4. The same individual may display different needs in different organisations or in different groups within the same organisation.

5. Employees can become productively involved with organisations on the basis of many different types of motives.
6. Organisational members can respond to different kinds of managerial strategies, depending on their own motives and abilities and the nature of the task.

This shift in our underlying assumptions influenced a similar shift in the ways in which the focus and role of management are perceived. It is important to point out, however, that the fundamental tasks of management have remained essentially the same: (1) to ensure that people are capable of joint performance through common goals and values, (2) to create the "right" structure to facilitate such interaction, and (3) to provide a sufficient level of training and development to ensure that organisational members can perform and respond to organisational and environmental changes (Drucker, 1989).

The broad changes discussed above have influenced a number of developments in management and organisation theory: (1) the emergence of Management Science and Operational Research, which emphasises the use of computers and mathematical models in organisational decision making; (2) the development of Systems theory, which looks at the organisation and its environment in totality; (3) the birth of Contingency theory which takes a situational (rather than a one-best-way) view of organisational structure and managerial activity; and (4) the growing influence of Organisational Behaviour which has, for many, become almost synonymous with management (Bowditch and Buono, 1994).

Management science

The basic orientation of the Management Science school is to apply quantitative techniques to management and organisational problems, and to merge strategic concern for planning and

forecasting with administrative concern for organisational objectives and goal accomplishment. Although there were some earlier developments in the use of statistical theory and quantitative methods in business organisations, most people identify Word War II as the starting point of Management Science. During this period, the first "operational research" team (the analytic profession within Management Science) was used to develop a plan to enable the British to deploy their aircraft efficiently, while at the same time providing effective air reconnaissance against Nazi convoys in the English Channel and surrounding waters (Chase and Prentis, 1987). The approach uses quantitative models representing actual conditions and computers for simulations and calculations to facilitate decisions which were previously made almost solely by judgement and experience. Thus, by calculating the speeds and patterns followed by enemy ships, and the time and fuel required by its aircraft to cover a given sector, the Allied commanders were able to reduce the number of reconnaissance flights needed and actually increase the effectiveness of their surveillance efforts (Churchman, Ackoff and Arnoff, 1957).

Following the war, the late 1940s and 1950s saw the development of numerous mathematical techniques — ranging from queuing theory and mathematical programming to game theory — that have become a basic part of the operations management (OM) practitioner's "tool kit" (Chase and Prentis, 1987). During the 1960s, the field continued to expand as its dominant focus on relatively narrow optimisation techniques gave way to the importance of viewing the production process as part of an overall system. This broadened perspective was further influenced during the 1970s by the widespread

application of computers to operational problems. From a manufacturing perspective, perhaps the most significant breakthrough was materials requirements planning (MRP) which ties together in a computer program all the parts that go into an end product. The resulting program provides the necessary information for production controllers to plan production schedules and inventory purchases to meet product demands (Forrester, 1971). While this framework does begin to focus on a broader view of organisations than previous theories, it is often criticised for its oversimplified view of reality.

The changing competitive circumstances in an increasingly global business world have continued to require manufacturers to improve both their productivity and product quality. These pressures have led OM specialists to become more involved in the application of advanced technologies, such as Flexible Manufacturing Systems and Computer Integrated Manufacturing, as well as new manufacturing philosophies. Indeed, due to the success of Japanese industry during the 1980s and early 1990s, the management science expanded its focus to include such management practices as justin-time (JIT) production and total quality management (TQM) systems (Monden, 1983). Yet, as the adoption and implementation of these new technologies and quantitative approaches has shown, the problems confronted by managers and their firms are just as much social and organisational as they are scientific and technological (Kamm, 1987).

Systems theory

As researchers began to realise the interaction between the structural (mechanistic) and human (behavioural) dimensions of organisations and the influence of the external environmental forces, the concept of a business firm as a system began to dominate organisational theory. According to this perspective, an organisation is a system composed of sub-units or sub-systems that continually interact with and are mutually dependent on one another (von Bertralanffy, 1972). Actions that occur within one part of the system not only affect that particular unit (department, section, group, etc.) but can have a "ripple effect" through other organisational sub-systems as well. The implication of Systems theory, therefore, is that things do not simply happen, but rather they evolve from multiple pressures and can entail multiple outcomes.

Within this context, organisations are often referred to as multivariate social or sociotechnical systems, composed of at least four basic components (Nadler and Tushman, 1977, 1980).

1. A Task/Technological Sub-system that consists of the basic or inherent work to be done by the organisation, specific work activities or functions, and the tools or technologies which enable that work to be accomplished.

2. An Administrative/Structural Sub-system (formal organisation), which includes such things as task groupings (units, divisions, departments), work rules, and policies; authority systems (reporting relationships, power bases,

control procedures); the way in which jobs are designed; and the physical work environment (internal).

3. A Sub-system of Individuals, which involves the people who perform the various organisational tasks, and the nature and characteristics of these individuals in terms of their knowledge, skills, attitudes, values, expectations, and perceptions.

4. An Emergent Sub-system (informal organisation), which develops over time as people interact within the formal system and includes such implicit and unwritten arrangements as norms (standards of behaviour), intra-organisational statuses, competition and co-operation between groups, and other "nonprogrammed" activities and interactions.

These four sub-systems are highly interdependent on each other. A change in administrative policies, for example, will not only affect those policies but can also exert an influence on the task, people performing the task, and the informal system.

Systems theory pertains to dynamic, ongoing activities within an organisation. An important step for understanding organisations is to look at dimensions that describe specific organisation traits.

Daft (1995) argues that organisational dimensions fall into two types: structural and contextual. Structural dimensions provide labels to describe the internal characteristics of an organisation. They create a basis for measuring and comparing organisations. They include formalisation of documentation, specialisation of functions and roles, standardisation of procedures, hierarchy of

authority, complexity, centralisation, professionalisation and personnel ratios.

Contextual dimensions characterise the whole organisation, including its size, technology, environment, goals and strategy, and culture. They describe the organisational setting that influences the structural dimensions. Contextual dimensions can be confusing because they represent both the organisation and the environment as the context within which the structural dimensions occur.

Organisations are also referred to as open systems. Conceptually, open and closed systems are compared (Mink *et al*, 1979). Systems that are completely self-contained and do not involve any interaction with their environment are termed *closed systems*. Systems that interact with their environment and are influenced by external forces, by contrast, are defined as open systems. No system, however, is totally open or totally closed (Kast and Rosenzweig, 1972).

Organisations are open systems and are influenced by a multitude of environmental forces or *inputs* such as availability of raw material, changes in technology, competition, changing worker values, governmental policies, and so forth. The organisation transforms these inputs into its goods and services (*outputs*) within the constraints imposed by its environment. In a bank, for example, deposits can be perceived as an input that is transformed (through record keeping and bank policies) into consumer outputs (loans, mortgages, and so forth). This process is influenced by a number of external factors such as changes in the inflation rate, government

policy, and consumer attitudes among other events. Thus, organisations receive inputs from their environment and transform these inputs into usable outputs for their customer or clients within the context of their environment.

There are limits or *boundaries*, however, to the *openness* of organisational systems. These boundaries are based on the inputtransformation-output process that links the organisation to other systems; behaviour not linked with these functions lies outside the system. Although the notion of a boundary is somewhat arbitrary and varies from system to system, it is an important concept since open systems are not affected by every external force or change.

As part of the input-transformation-output process, an organisation receives *feedback* or information concerning its performance. The continuous flow of information between a system, its internal components and the external environment forms a *feedback loop* that enables the organisation to adapt to changing environmental conditions. Thus, the capacity to use this information to control the system and make necessary changes is crucial if organisations are to become self-correcting systems.

Another important aspect of systems is that they seek *equilibrium*, a steady state of regularity, consistency, or balance. When something occurs that puts the system out of balance, the system reacts to bring itself back into balance. This does not imply that organisations are, or should be, static and unchanging. Rather, as the feedback process discussed above indicates, organisational systems can use the information provided through their feedback loops to make

certain adjustments to changing conditions (adaptation), and reach and maintain for as long as appropriate a new balance point. This equilibrium point then becomes the steady state until another change indicates the need for further adaptation.

The primary advantage to System theory is that it provides a framework for thinking about organisations in more complex and dynamic terms than earlier management and organisational theories. The application of a Systems perspective is especially useful when assessing the need for major organisational change (Bowditch and Buono, 1994). The theory can guide managers to think about how change in one part of the organisation may affect other aspects of the organisation, and the (potential) effect of the external environment. A potential problem, however, is a tendency to be evaluative in analytic efforts; that is, perceiving a particular behaviour as being "good" or "bad". Systems analysis is based on the *functionality* of a particular event or behaviour, or the extent to which the event contributes to the maintenance of the system (Leighton, 1978). If a given event occurs that dislocates a particular system or sub-system and leads to difficulties, the event is dysfunctional to that system. If an event reinforces the system, it is *functional* to that system. Caution must be exercised, however, when determining functionality. Certain behaviours that may appear to be dysfunctional for task performance (e.g., game playing, socialising) may be quite functional for individuals and the informal system by fulfilling social needs and reinforcing certain norms and interaction patterns.

Contingency theory, or the contingency approach to organisation study, developed in parallel with system theory. Whereas system theory is more concerned with the universal prescription based on the interaction between different systems and/or sub-systems, contingency theory tries to pay heed to predicting efficient form in certain situations of a specific organisation.

Therefore, contingency theory addresses itself to the relational fit between an organisation and its context. There is no universal form of effective business, but the universality exists in the relationship between an organisation and its context to the effect that the more the fit, the better the efficiency across different organisations, industry sectors and cultures.

As the starting point of this dissertation, contingency theory will be discussed in detail in the following section.

2. 3. Contingency Theory: the Starting Point

The central thesis of Contingency theory is that there are no universal principles of management that can be applied uncritically in all situations. Organisation and management approaches must vary from one firm to the next, because it depends on the unique environmental conditions and internal factors which are inherent to each organisation. Contingency theory proposes that the best organisational design is bound up with the situational conditions (Takahashi, 1988). The central concept of contingency theory is congruency between the organisational design and the characteristics of the environment in which a firm operates. Thus, a situational approach to decisions about organisational structure and appropriate managerial behaviours is necessary.

A review of representative works on contingency theory comes next in the search for factors influencing organisational structure.

2. 3. 1. Representative works of contingency theory

Woodward (1958) investigated whether the principles of organisation, as laid down by management theory, correlate with business success. Her research shows that conformity with the "rules" of management does not necessarily result in success nor does non-conformity cause business failure. It shows, instead, that there are differences in organisation structure — for example, the existence or otherwise of line-staff patterns, the number of levels of management, etc. Hence, she concluded that firms using similar technical methods or production systems have a similar organisational structure. Her interpretation is 'that different technologies impose different kinds of demand on individuals and organisations, and that these demands have to be met through an appropriate form of organisation. Likewise, Woodward found that the size of the corporate centre was a more reliable measure of the "bigness" of a firm than its total personnel. "Bigness" refers to the characteristics of a large firm, including a well-defined and developed management structure, considerable financial resources. etc.

Burns and Stalker (1961) posit two "ideal types" of organisation: the one mechanistic, adapted to relatively stable conditions; the other, organic, adapted to conditions of change.

In mechanistic systems, adapted to relatively stable conditions, the problems and tasks which face the organisation as a whole are, typically, broken down into specialisms. Each individual carries out his or her assigned task as something apart from the overall

purpose of the company. The corporate centre is responsible for seeing that an individual's work is relevant to that of others. The technical methods, duties, and powers attached to each post are precisely defined, and a high value is placed on precision and demarcation. Interaction within the working organisation follows vertical lines — namely, between superiors and subordinates. How a person operates is prescribed by his, or her, functional role and governed by instructions and decisions issued and taken by superiors. This hierarchy of command reflects the assumption that only the person at the top knows all about the firm. Hence, this person is the only one who knows how human resources can properly be deployed. The management system operates as a simple control system with information flowing upwards through a succession of filters, while decisions and instructions flow downwards through a succession of amplifiers.

Burns and Stalker (1961) argue that organic systems adapt to unstable conditions, in which new and unfamiliar problems and requirements continually arise which cannot be broken down and distributed among specialist roles within a hierarchy. Therefore, jobs lose much of their formal definition, as responsibilities and functions have to be constantly re-defined through interaction with others participating in common tasks or in the solution of common problems. Each individual has to do his or her job with a knowledge of the overall purpose and situation of the firm. Interactions run horizontally as much as vertically and communication between people of different rank tends to resemble 'lateral' or horizontal consultation rather than 'vertical' command.

Lawrence and Lorsch (1967) suggest the importance of structural differentiation, such as division of work into organisational systems and the attendant problems for achieving integration of activities. They were among the first to apply contingency theory to organisations in proposing that organisational performance is related to the degree to which (a) organisational divisions are successfully integrated management structures and (b) processes are developed to fit the levels of uncertainty encountered in the environments of organisations.

Hage and Aiken (1967) argue that an organisation's structure can be described by dimensions measuring (a) the pattern of division of labour and (b) the vertical and horizontal distribution of power and influence within the organisation.

Four such dimensions were recognised as adequately representing organisation structure (Pugh *et al*, 1968). These are (1) the degree of formalisation, namely, the extent to which the authority and responsibility of every manager are clearly defined, (2) the degree of standardisation as in formalised job descriptions, performance evaluation system, etc., (3) degree of centralisation, and (4) the interdepartmental distribution of influence.

Regularities have emerged from different studies in the forms of relationships between contextual (situational) variables and characteristics of organisation structure, and also between features of the structure itself. In addition to this research on the corporate centre, there are two particularly influential arguments relevant to an explanation of variation in organisational structure. The first concerns the influence on structure of a physical organisational attribute: size. The second argument comes from the environment, that is, environmental conditions are posited as critical constraints upon the choice of effective structural form.

Perrow (1967) presented a technology-based perspective on organisations. Technology is meant to be "the actions that an individual performs upon an object or 'raw material' — living beings, human or otherwise, a symbol or an intimate object." Technology is considered the defining characteristic of organisations.

Child (1973a) examined size as a predictor of organisation structure with data from a sample of British business organisations. He found that, while the broad outlines of formal organisation structure are predictable with a high degree of confidence from a knowledge of organisation size, a comparison of size-structure regressions across different industries suggests that other variables must also be taken into account. Child asserts that while size, with technology, location and environmental variables, predicts complexity, the degree of complexity itself has a more direct relationship with formalisation than does size. Size, however, is the main predictor of decentralisation.

Environment conditions have also been singled out as critical constraints upon the choice of effective structural forms (Duncan, 1972; Randall, 1973). The degree of change and the heterogeneity

and range, which characterise environmental activities, are considered to be relevant to an organisation's operations. "Environmental stress", or the degree of threat to organisational decision-makers in the achievement of their goals arising from external conditions, hostility or even indifference, is said to be associated with organisational "slack" (Khandwalla, 1974).

In summary, there are three main emphases within the Contingency school (Fiedler, 1967; Lawrence and Lorsch, 1967; Thompson, 1967; Woodward, 1965).

1. One stream relies heavily on open systems planning (developed from General Systems theory) and stresses that there is no one "best" solution for all organisations, since each firm has to cope with its own unique set of technical, human, and market inputs.

2. A second major emphasis focuses on the formal design of organisations for different technologies and economic environments. Research, for instance, has indicated that in relatively predictable markets with simple, repetitive technologies the more formal, pyramidal organisation structures tend to be effective, while in more uncertain environments with complex technologies, a "flatter" or matrix form of organisation seems to be more effective.

3. The third major component of Contingency theory focuses on leadership style and behaviour. The main conclusion of this stream is that rather than taking one basic approach to

leadership in all situations, leadership style should vary according to such factors as the nature of leader-member relations, the degree of task structure, the power (positional and personal) a leader has, the expertise and willingness of subordinates to assume responsibility, and so forth.

2. 3. 2. Criticisms of contingency theory: sociological explanations

The criticisms of contingency theory lie in its failure to refer to some relevant facets of organisational behaviour. The theory implies a view of organisation which is insufficient from the standpoint of sociological explanation, and this inhibits it from identifying the full extent of opportunities for variety which are likely to exist in organisational design (Child, 1973b).

From the standpoint of sociological explanation, there are three objections to contingency theory. They relate to (1) the political process in organisations, (2) the inadequacy of regarding situational "contingencies" as functional imperatives, and (3) situational imperatives for organisational design.

The first objection to the contingency theory runs as follows. Its approach encourages managers to weigh the implications of the operating situation in which they find themselves against their preferred courses of action. The possibility that managers may have preferences for outcomes other than those indicated by an application of contingency theory leads to the first major objection against the theory. Decision making about organisation is not

simply a matter of accommodating to operational contingencies. It is equally a political process into which other considerations, particularly the expression of power holders' values, also enter (Child, 1973b; 1984).

Much of the social science literature on organisations makes little or no reference at all to political processes, since the social scientist has normally found himself able to research only the consequences of policy which has already been established, especially the extent to which the manifest objectives of the policy have been achieved (Child, 1973b).

Dubin (1962) in his review of available studies of decision making in organisations points out the considerable time which often elapses between the recognition of a problem and the decision regarding its solution. One of the reasons for this time-lag is the operation of the political factor. It would appear typical for an attempt to be made to secure some degree of "legitimation" for the proposal in the eyes of interested parties who have power to obstruct the implementation of any decision which they do not accept. During this process there is an effort to accommodate in the decision the disparate interests held by such individuals or groups. The political process in organisations therefore incorporates an important normative aspect, and it is the means through which the values of various groups impinge to a greater or lesser degree upon organisational decisions (Dubin, 1962).

This process operates in regard to the interface between situational factors and the structure of organisations, because political criteria
are brought to bear upon the way in which contingencies are interpreted (Child, 1973b). To refer to an empirical example, Norman (1971) finds in thirteen case studies of new product development in Swedish firms that the existing values and power structure in a firm played a critical role in predicting reaction to new ideas and information emanating from the environment.

The second objection to the contingency theory argues as follows. A recognition of the part played by the political system in organisations also draws attention to the inadequacy of regarding situational "contingencies" as functional imperatives for organisational design (Child, 1973b). The implication that the relationships between situational variables and organisational variables are "mechanistic" is open to qualification for two reasons. First, some variation appears to be possible in the design of organisation for a unit operating within a given situation, without apparently incurring serious consequences for its level of economic performance. Second, the type of situation in which a unit is operating may itself be open to some degree of deliberate manipulation and selection on the part of organisational decision makers, and therefore it does not necessarily represent a fixed point of reference for them.

Several studies (Rice, 1958; Trist *et al*, 1963; Miller and Rice, 1967) have recorded experiments which appear to demonstrate that, while the choice of organisation at operating levels is limited by technology, the limits do nevertheless allow room for a number of alternatives. In some of the experiments, substantial changes were made in the structure of working groups and in their degree of

autonomy within the overall technological rationale of the production processes. In most cases these changes had favourable results, both in terms of social criteria such as job satisfaction.

The third objection runs as follows. The foregoing line of argument does not deny that the relationships between situational and organisational variables reported in comparative studies may reflect pressures to establish organisational arrangements which cope with prevailing circumstances. It may, however, qualify the conclusion that these findings imply the presence of situational imperatives for organisational design. Extremely little is known about the nature of cause and effect in this whole area (Child, 1973b). However, a cause-effect relationship is quite likely in the case of the association between "bureaucratic" features and large size (Parkinson, 1958; 1980).

Parkinson (1958; 1980) attributed to dysfunctional uses of bureaucracy the rise of the proportion of indirect to direct employees within industry. Departments grow because managers prefer to multiply subordinates, not rivals. In turn, more subordinates make more work for each other. This empire building is more easily achieved in areas away from direct production, since in production a growth in numbers would often be constrained by technology and by established controls on work load, stock levels, and manning. Drake and Mitchell (1977) maintain that power dimensions are important for understanding people's behaviours in the decision process.

Child (1973c; 1984) argues that a political approach advanced by Parkinson (1958; 1980) would in a sense act as a counterbalance to

an exclusive reliance on contingency explanations. His view is that it should lead us to a study of the decisions which are in practice taken on the permissible sizes of divisions and departments within organisations. For Child (1973c), Parkinson (1958; 1980) made a serious point in drawing attention to such decision making as the essence of his Law.

What may well be happening in practice is that, as an organisation grows larger, it begins to employ "professional" managers who take some interest in the available management wisdom. This tends to recommend the establishment of routine system, clear job definitions, the use of delegation, and other bureaucratic features as a basis for orderly and effective operations. Hence as organisations grow larger, such precepts are increasingly applied (Child, 1973b).

The contingency approach not only specifies that organisation should be designed to suit situational circumstances, but it also implies that such circumstances may be taken as given. This latter presumption is not wholly acceptable for two reasons. First, Child (1973b) holds that if there are limits within which an organisation's size, technology, and mode of adjustment to the environment should fail if the organisation is to obtain a high level of economic performance, then these limits do not as yet seem capable of precise definition. The second reason is that the directorates of organisations do in practice over a period of time modify the circumstances in which their organisation is operating, and equally that they sometimes refuse to make such changes even when these become possible. The manipulation of situational variables in the

policy prescription. This is important if policy prescription is to be a main point of our research.

Sociological explanations such as social constructions and dysfunctional uses of bureaucracy contrast with the contingency approach in that (1) they are more appropriate to conduct *ex-post* analysis and provide descriptions of organisation on the basis of theoretical speculation, and (2) these analysis and descriptions are mainly aimed at an academic audience for their discussions.

Easiness of drawing testable statements from contingency theory

Contingency theory offers an appropriate framework for empirical investigation because of its consistency with a number of analytical paradigms. Predictive utility further enhances its appropriateness, and is essential to our research framework, in which an independent variable — information flow — is tested to predict a dependent variable — the centre's size. Thus, predictive utility makes contingency theory a better choice than other competing approaches.

Appropriateness for empirical investigations

Contingency theory was developed through empirical research. Consequently it is an appropriate framework for empirical investigation into organisations (Kagono *et al*, 1985). It is consistent with a number of analytical paradigms such as the information

processing model, and can incorporate many different empirical findings.

Predictive utility

Contingency theory seeks more than one type of clustering among variables. Different relationships are expected in different contexts and configurations. However, it goes somewhat further in searching not just for richly described types but for ones with *predictive utility* (Miller, 1981). They represent statistically significant clustering among variables such that, given a partial description of an organisation (or scenario), it becomes possible to identify its category and thereby to accurately predict many of its other features.

Two arguments against the feasibility of predictive utility may be adduced:

(1) An infinite number of adaptive patterns are possible when many variables are used. That is, there will be too many types, all or most of them being relatively rare.

(2) Inclusion of the time dimension and of strategy-making variables will be particularly troublesome in that they add complexity to the findings. Long time periods may allow intricate and therefore unique scenarios to develop.

Concerning the first argument, Miller (1981) alleges (1) natural selection, (2) adaptive structural functionalism and (3) holistic interdependence among internal factors in defence of contingency

theory¹. He argues that, due to these factors, the effective and applicable patterns which are important for practising managers are actually limited in their number.

Concerning the second argument, it may be contended that (1) strategy-making is closely tied to structure and environment and (2) adaptive scenarios take place over time and seem to follow common progressions². Thus, in the long run, effective organisational patterns will converge on a limited number rather than diverge.

Applicability for practising managers

As one of the major purposes of this research is to provide practising managers with practical and comprehensive suggestions, the adoption of the contingency approach is justified, since it will help to shed light on such emerging management questions as lean production management and lean management.

¹ Each type of firm has integrally related features: the nature of the technology influences spans of control, the number of levels in the hierarchy, the critical functional area, etc. (*e.g.*, Burns and Stalker, 1961; Woodward, 1965; Filley and House, 1969; Perrow, 1970; Mintzberg, 1979). In other words, adaptive patterns do not only have their variety limited by external constraints through selection and adaptive structural functionalism, they also must have an internally consistent set of attributes.

 $^{^2}$ Filley and House (1969) as well as Chandler (1962) have described several stages of development which firms pass through in a fixed sequence, regardless of their industry sectors. The sequence leads from a simple, one-man firm, to a functionally organised enterprise, to a divisionalised corporation. Each stage has its own complex set of structural parameters which change with size and age in a certain sequence. Penrose's (1980) study also points to the structured patterns which one finds in corporate history.

Practical use for managers

As Lupton (1971) points out, this approach encourages managers to search for an organisational design to suit their particular operational requirements instead of relying upon a universal precept:

It is of great practical significance whether one kind of managerial style or procedure for arriving at decisions or one kind of organisational structure is suitable for all organisations, or whether the managers in each organisation have to find the expedient that will best meet the particular circumstances of size, technology, product, competitive situation and so on. In practice managers do, indeed must, attempt to define the particular circumstances of the unit they manage, and to devise ways of dealing with these circumstances. (Lupton, 1971:125).

<u>Comprehensiveness</u>

The contingency approach offers a comprehensive framework which can account for interdependent relationships among many organisational phenomena, including environmental, organisational and strategic variables (Pennings, 1992). It seeks to look simultaneously at a large number of variables that collectively define a meaningful and coherent slice of organisational reality (Miller, 1981).

Kagono *et al* (1985) believe that a business organisation is a unified whole, whose components, namely, objectives, strategy, technology, organisational structure and process, as well as the personal predisposition of its members are interrelated and interdependent. Therefore, if the organisation is to cope effectively with its environment, it must develop an integral configuration among its components. Kagono *et al* (1985) contend that an understanding of the interrelationships among the organisational components as well as between the organisation and its environment is important.

2. 4. Information-processing Need As a Contingent Factor Influencing the Size of the Corporate Centre

The central concept of contingency theory is the congruency between the organisation design and the characteristics of the environment in which a firm operates. While intuitively appealing, this concept is elusive and undefined (Hrebiniak, 1981). It is pointed out that congruency could refer to a broad *gestalt* — a particular configuration of organisation and environment that is whole and complete as a picture. If so, no causality is indicated, since both environment and organisation are needed to complete the picture (Hrebiniak, 1981; Miller, 1981).

2. 4. 1. Uncertainty, equivocality, and turbulence

In order to explicate the concept of congruency between organisation design and environment, and to integrate the diverse organisation design/structure literature, concepts of uncertainty (Galbraith, 1974; Tushman, 1978), equivocality (Daft and Lengel, 1986) and turbulence (Dess and Beard, 1984; Haleblian and Finkelstein, 1993) have been introduced.

Tushman (1978) views organisations as information processing systems facing uncertainty. Uncertainty is defined as the difference between information possessed and information required to complete a task. The basic notion developed by Tushman is that sub-units — sets of groups or departments — face different amounts of work-related uncertainty and that, to be successful, they must match information processing capacity to information processing needs. Since different structures have different information processing capabilities, sub-units can deal with workrelated uncertainty through appropriate structural arrangements. Thus, the essence of organisational design is that, to deal most effectively with information processing requirements, a particular set of organisational arrangements must be chosen from a feasible set of structural alternatives (Tushman, 1978).

Daft and Lengel (1986) hold that, in addition to uncertainty, equivocality (defined as ambiguity due to the existence of multiple and conflicting interpretations of an organisation situation) influences organisational design. Their view agrees with Tushman's (1978) in that organisational structure determines both the amount (quantity) and the richness (quality) of information provided to managers.

Equivocality leads to the exchange of existing views among managers to define problems and resolve conflicts through the enactment of a shared interpretation that can direct future activities, whereas uncertainty leads to the acquisition of objective information about the world to answer specific questions (Daft and Lengel, 1986: 557).

In relation to uncertainty, structural design can facilitate the amount of information needed for co-ordination and control of the organisation. As to reducing equivocality, structural mechanisms have to facilitate the processing of "rich" information, that is, the "quality" information which helps to change existing views among managers within a time interval (Daft and Lengel, 1986).

Environments vary in their degree of turbulence or stability (Dess and Beard, 1984). Turbulence refers to volatility or unpredictable discontinuity in an environment (Aldrich, 1979; Dess and Beard, 1984; Keats and Hitt, 1988). Problems in a turbulent environment require a great quantity of information processing (Daft and Parks, 1988).

All these authors consider information processing in organisations as essential to counter uncertainty, equivocality or turbulence. They consider that information-processing needs are handled by organisational design³. Thus, these authors contend that information-processing is an essential factor in the design of "effective" organisation to cope with environment.

³Their view is much in line with Weick's (1979) idea about designing the organisation to reduce uncertainty.

2. 4. 2. The size of the corporate centre and informationprocessing capabilities

The corporate centre is considered to be the information-processing centre of an organisation in its relationship with its environment (Thompson, 1967). Team size helps to determine the corporate centre's information-processing capabilities. Therefore, "largeness" provides increased capabilities and viewpoints, which can be critical in turbulent environments (Haleblian and Finkelstein, 1993). In fact, Hambrick and D'Aveni (1992) found that firms nearing bankruptcy tended to have smaller corporate centres than survivors⁴.

Large groups can enhance problem-solving capabilities resulting from information-processing by increasing the number of (1) items of information that can be absorbed, (2) critical judgements available to correct errors in inference and analysis, (3) potential solution strategies, and (4) the range of perspectives brought to bear on a problem (Harrison, 1975; Shaw, 1981).

In the light of this relationship between the size of the corporate centre and its information-processing capabilities, it may be contended that the firm's environmental characteristics, through the information-processing needs arising from them, influence the size of the corporate centre.

^{*}Scholars have argued that large groups are superior to small ones because large groups have more capabilities and resources with which to solve group tasks (Hill, 1982; Jackson, 1992).

Due to the difficulty of directly measuring information-processing phenomena⁵ between sub-units of an organisation, a different approach to operationalising the information-processing perspective is needed. Galbraith (1973) contends that information-processing capabilities of an organisation are influenced by the design features including rules and programmes, hierarchical referral, goal setting, vertical information systems and lateral relations. He also argues that, when uncertainty (*i.e.*, information gap between the information required and the possessed) increases, exceptions must be referred up the hierarchical structure for decision-making, and, therefore, information capabilities will be enhanced by vertical information flow between the corporate centre and lower management. However, when uncertainty and informationprocessing needs are large, there must be lateral (i.e., horizontal) relations in addition to the vertical information flow, in order to share the information-processing capabilities which cannot be fully loaded in the centre.

Based on the above argument by Galbraith (1973) and also on account of difficulties to measure information flow directly, the present study uses information-processing as an intervening concept to aid in positing relationships between the size of the corporate centre and planning and control influences (vertical

⁵A number of empirical studies have used some form of information-processing approach (Burns and Stalker, 1961; Lawrence and Lorsch, 1967; Duncan, 1973; Van de Ven, Delbecq, and Koenig, 1976; Galbraith, 1977; Tushman, 1978b). The more micro-level studies of organisations, where the units of analysis are either individuals or small groups, have managed to measure directly such aspects of information processing as the frequency of oral communications between work groups (Tushman, 1978b), the extent to which policies and procedures, work plans, personal contact, and meetings are used to co-ordinate members of work teams (Van de Ven, Delbecq, and Koenig, 1976), and the structure of groups during decision-making (Duncan, 1973).

information flow, to be dealt with in Chapter 3) and linkages (horizontal information flow, to be dealt with in Chapter 4).

The "functionalist" interpretation of information-processing does not exclude other approaches like that of communications process which influences the organisation structure and, therefore, indirectly the size of the corporate centre. Masuda (1981) believes that informatics (or Information Technology) will change organisation structures from chain-of-command pyramids to matrices, and that hierarchies will be flatter, as the technostructure of experts becomes a larger part of the total workforce⁶. Daft (1995) predicts that the impact of Information Technology on the structure of organisations can be: (1) flatter organisation structure, (2) greater centralisation or decentralisation, depending on manager choices, (3) improved co-ordination, (4) fewer narrow tasks and more intellectual engagements and challenging work, and (5) larger professional staff ratio. Carney (1986) discusses the emerging paradigm for the understanding of organisational communications exemplified by the "soft" skills of personnel management and human resource development in Japanese firms. As noted by Pascale and Athos (1981), these skills are noticeable in the following areas:

• care taken to choose and develop the type of employee desired;

⁶ It is likely that informatics is not affecting organisation structures too much. PC applications like Lotus Notes link people together maintaining chain-of-command pyramids.

• developing skill specialities in key staff so as to make the organisation outstanding in its field;

• developing an appropriate top management style and an organisational culture to go with it;

• having a transcendent goal and philosophy to give a larger meaning to working for the organisation.

The discussions on organisation theory from a sociological standpoint will be further elaborated upon in Chapters 3 and 4.

2. 4. 3. Information-processing and contingency theory

This research is based on the contingency theory's argument that there is a congruency between organisation design and the characteristics of environment. Therefore it uses the following framework to explain information-processing and its influence on the size of the corporate centre. (See Figure 2. 2.)





* (c) and (d) are surrogates of (a) and (b)

** (a) and (c) influence (b) and (d).

The figure shows that the size of the corporate centre is contingent on the planning and control influences (See Chapters 3 and 4), which are, in their turn, contingent on information-processing.

2. 4. 4. Definition of the terminology

As a cornerstone of this research, information-related terms ought to be defined.

- Information means input or output of strategic decision-making.
- Information flow (vertical or horizontal) relates to the way information passes through different organisational units.

- Information-processing means a process which adds value by means of storing, analysing, transforming and creating information.
- Information-processing capability is concerned with the effectiveness and efficiency in the information-processing and is a combined effect of information structure organisation structure and dynamics as well as planning and control intensity and information resources.

2. 5. Summary

In conjunction with Chapters 3 and 4, this chapter forms part of the literature review.

Organisation theories including administrative, scientific management, structuralist, neo-classical, human relations, behavioural, management science, systems, and contingency theories were reviewed. Contingency theory is adopted as a starting point, since it makes it easy to draw testable statements which are amenable to empirical studies and its results are applicable for corporate management. Factors influencing organisational structure, including the size of the corporate centre, were examined. Information-processing was found to be a factor influencing the size of the corporate centre.

The next chapter discusses the functions performed by the corporate centre and how they relate to the vertical information flow.

Thereafter, the corporate centre and the horizontal information flow are discussed in Chapter 4.

information flow, and, therefore, influence the size of the corporate centre.

Section 3. 4 compares the alternative explanation of the centre's size based on sociological elements with the argument of this research based on information-processing capabilities. It is concluded that information-processing needs and capability are the main determinants of corporate centre size, although due recognition is made, in many situations, to the explanatory power of sociological elements.

Section 3. 5 examines the works of those authors who have analysed different types of corporate centre. These authors consider different dimensions to influence the type and size of corporate centre, while planning and control influences are found to be closely related to vertical information flow.

Section 3. 6. summarises the discussions in the chapter.

3. 2. The Centre's Functions and Vertical Information Flow

This section discusses the functions performed by the corporate centre and how they relate to the vertical information flow. It first addresses the emergence of the corporate centre as the cardinal point of corporate organisation and, thereafter, the functions it performs, namely, planning, allocating resources, controlling and auditing performance, and providing central services. The discharge of these

functions gives rise to vertical information flow and informationprocessing needs.

The literature on the "nature" of the corporate centre, namely, how the corporate centre has been formed, what functions it discharges in corporate management, what value added it creates, etc., is sparse. However, what emerges from the literature is that there are reasons for the existence of a corporate centre that takes on the role of intermediary between the business unit and the financial markets. Our review is centred on the works of Chandler, Williamson — two representative authors on the subject.

Chandler (1962) studied how four major US firms (Du Pont, General Motors, Standard Oil of New Jersey and Sears, Roebuck) changed from a functional structure to a divisional structure. These were the first firms to devise this "decentralised" form. Du Pont and General Motors began to fashion their new structure shortly after World War I. Standard Oil of New Jersey started its reorganisation in 1925, and Sears, Roebuck in 1929. There were some firms that initiated comparable changes between 1925 and 1932, but these administrative reorganisations proved to be less creative innovations than those of the afore-mentioned four major firms. Therefore du Pont, General Motors, and Standard Oil of New Jersey were selected for study. Sears was also chosen because its activities were complex and because information about the firm was more readily available. All four of these firms first adopted a "functionally departmentalised, centralised structure". The centre's role in a functionally organised firm is to co-ordinate and integrate the output of the several functional departments to changing market demands, to provide central services, and to allocate the future use, as well as to appraise the present performance, of the resources of the enterprise. Chandler (1962: 293) states:

The executives at all four companies came to realise...that systematic policy formulation and allocation of resources called for carefully defined budgeting and capital appropriation procedures. This planning also required information on future financial and general economic conditions as well as on anticipated market demand.

Chandler argues that, when the administrative load on the senior executive officers increased to such an extent that they were unable to handle their entrepreneurial responsibilities efficiently, the innovators built the divisional structure with (a) a corporate centre whose executives would concentrate on entrepreneurial activities and (b) autonomous, fairly self-contained operating divisions whose managers would handle operational activity. Thus, the corporate centre focused more on planning and strategy. Budgeting and capital approval processes were also strengthened to permit central control. The centre reduced its involvement in co-ordination, limiting its role to forming links between divisions and leaving the functional co-ordination to the general managers of divisions. The experience of all four firms clearly shows how concentration on operational activities interfered with, and inhibited, long-term planning, including the designing of a new structure to meet the administrative needs of a new strategy. Only when the executives withdrew from day-to-day tasks did they produce useful organisational plans or proposals. The executives in the corporate centre, freed from all but entrepreneurial responsibilities, could fully devote themselves to planning and appraisal (Chandler, 1962).

In short, decentralising enabled the centre focus attention on adding value to the divisions through activities relating to planning and the allocation of resources, control and audit of performance, and by providing some central services. In the functional structure, the centre had carried out these activities, but had become overburdened with coordination and operating issues.

Chandler thus explained why the divisional structure has been more successful than the functional structure in companies with diverse businesses. What he did not explain is whether the centre of a divisionalised company is more effective in allocating resources and exercising control than are the financial markets.

Williamson (1975, 1981, 1991) filled this explanation gap. He supports Chandler's view that the divisional structure could be more effective than the functional structure, but he also considers why businesses within a large diversified firm perform better than they would as separate firms, each responding to the pressures of the outside financial markets. Williamson (1975, 1981, 1991) calls his way of focusing on the study of organisations the "transaction cost" approach, which claims that the economising of transaction cost is central to the study of organisation.

Williamson (1975) argues that under the divisional structure, because of efficient and effective control by the corporate centre, control and internal efficiency are both enhanced within a large diversified firm, as compared with the case in which divisions are separate firms. He likens this control to the capital market control of management, calling the divisional structure a miniature capital market.

Williamson (1975) holds that the corporate centre is more efficient than the capital market for three reasons. First, it is an internal rather than external control mechanism with authority, expertise and lowcost access to the requisite data, which permits the corporate centre to make detailed evaluation of the performance of each of its operating parts. Second, the corporate centre can undertake fine-tuning as well as discreet adjustments. Taken together, these two factors permit the corporate centre to intervene early in a selective, preventive way. Finally, the costs of intervention by the corporate centre are relatively low.

Chandler and Williamson argue that the corporate centre of divisional structure can be more effective than either a functional structure or the capital markets in managing the diversified portfolio of businesses. The centre's functions as intermediary between the businesses and

the financial market are found by Chandler to be planning and allocating resources, controlling and auditing performance, and providing some central services¹.

These basic functions mentioned by Chandler (1962) relate to the vertical co-ordination described by Hungenberg (1993), who asserts that the centre's intervention in decision-making processes at the business unit level will define the intensity of this vertical co-ordination and thus its information flow. Different types of corporate centres in terms of their strategic focus will then have different staff size to carry out the necessary information-processing.

In summary, in the discharge of the centre's functions outlined above, *i.e.*, planning and allocating resources, controlling and auditing performance, and provision of centre services, a need for informationprocessing for vertical co-ordination arise. This need varies as the corporate centre varies its strategic focus.

¹The functions of the corporate centre have also been separately studied by other authors. Steiner (1968), Child (1972), Vancil and Lorange (1975), Hofer and Schendel (1978), Christensen *et al* (1978), Bower and Doz, (1979), Andrews, (1980), Andrews (1983), Burgelman (1983), Hambrick and Mason (1984), Galbraith and Kazanjian (1986), Ansoff (1987), Hamel and Prahalad (1989), Chakravarthy and Lorange (1989), Hax and Majluf (1991), and Daniel (1992) are some of them.

3. 3. The Size of the Corporate Centre as a Function of Vertical Information Flow

Before addressing the size of the corporate centre and the vertical information flow, this section examines how strategic management style influences information flow.

Based on the results of a cross-section interview survey, Goold and Campbell (1987) define two dimensions of the corporate centre's influence on lower management, namely, planning influence and control influence, which are two of the three functions described in the preceding section.

As a background to these two dimensions of influence, Goold and Campbell point out that there are basic conflicts or tensions among different layers of management, *i.e.*, between the profit centre, the business unit, the division, the group and the corporate headquarters or centre.

The five fundamental tensions encountered by Goold and Campbell (1987: 164) are:

- clear responsibilities versus multiple perspectives
- detailed planning versus entrepreneurial decisions
- strong leadership versus business autonomy
- long-term strategic versus short-term financial objectives

• tight control versus the flexible pursuit of long-term objectives.

In order to see how the centre copes with these tensions, Goold and Campbell focus on the way the centre influences managers lower down thereby affecting the decisions they take. Therefore, there is a continuous vertical information flow between the centre and the lower management. Two dimensions of the centre's influence which they identify are planning influence and control influence.

Planning influence concerns the centre's efforts to shape strategies as they emerge and before decisions are taken. Goold and Campbell (1987: 36-40) cite seven elements which the corporate centre uses to influence lower management. These are: organisation structure, review of plans, strategic themes, broad strategic thrusts, specific suggestions, management of linkages between businesses and allocation of resources. These elements are discussed below.

Organisation structure: the way the organisation is structured will affect the degree of planning influence. If a manager has full responsibility for a business unit, without overlaps or functional links with other units, he will feel less pressure to seek top-down advice and guidance. If, on the other hand, there are extensive overlaps and dual responsibilities, the unit manager will feel more external influences on his planning decisions.

Review of plans: business units review their plans with the centre. In some firms this review is carried out only on the annual budget. In others, the centre reviews strategic plans as well as budgets. These formal reviews provide an opportunity for the centre to guide the managers in the unit. Alongside these formal reviews there is continual informal contact between managers at the centre and in the business units. These informal reviews are an important part of the influence process.

Strategic themes: in some firms, the corporate centre has a view of what are the 'distinctive competences' of the firm as a whole. Even though such themes are often stated in the broadest of terms, and allow business units latitude to interpret them in their own way, they can have an important top-down impact on the choice of strategy.

Broad strategic thrusts: a related, but separate, type of influence stems from the specification by the corporate centre of broad objectives or thrusts for particular business units. Such a specification may include broad product range goals, market development objectives and competitive position. It provides a framework for strategic thinking.

Specific suggestions: the corporate centre may influence plans by making specific suggestions. The degree to which central managers refrain from intervening with specific suggestions reflects their commitment to decentralisation. Suggestions may arise from formal or informal budget or plan discussions. Suggestions form part of the topdown influence process. Management of linkages between businesses: central influence in the form of broad thrusts or specific suggestions is exercised where linkages between businesses or divisions need to be managed. Linkages relate to the co-ordination of functional strategies, crosssupply and transfer pricing between units in a vertically integrated chain, sharing or transfer of expertise, and exploitation of a shared resource. Linkages lead to opportunities for the corporate centre's intervention. The degree of influence which the centre exerts is a function of how active central managers want to be in resolving the linkage issues.

Allocation of resources: the most powerful way in which the corporate centre can influence strategy is through the allocation of resources. By supporting one investment project rather than another, the centre can affect the whole composition and performance of the businesses that make up the firm.

Control influence concerns the way in which the corporate centre reacts to results achieved. Whereas planning influence is about the 'inputs' to decisions, control influence is about the results of decisions, *e.g.*, the 'outputs' such as profit or market share. Control influence arises from the targets that the centre agrees with its business units, the way the centre reacts to poor performance, and the frequency with which the centre monitors results. Therefore the vertical information flow derived from control influence is on a discrete basis. The budget process, the capital appropriation system, and the

strategic planning system provide the formal framework for control, which is essentially a linked process of agreeing upon objectives, monitoring results and applying pressure and incentives (Goold and Campbell, 1987: 40-41).

Agreeing upon objectives: the setting of objectives is the first step of the control process. The type of control influence adopted by the corporate centre varies according to the precision and detail of targets, the balance between objective and subjective measures, the time frame for achievement, the influence of the centre in proposing and agreeing upon objectives, the degree of 'stretch' built into objectives, and the emphasis on financial versus non-financial targets.

Monitoring results: the way in which the corporate centre seeks out performance information, the type of information it requests, and such arrangements as the frequency and duration of discussions it has with managers about the results are part of the control process.

Pressure and incentives: bonuses or careers linked to performance targets enhance the pressure of the control process. The reaction of the corporate centre to poor performers is an important influence both on the sorts of strategies that business managers are likely to propose and on the actions they take during the year.

In summary, the centre's planning influence is affected by the organisation structure, the review processes, the type of guidance from the centre, the way overlaps are managed and the way scarce resources are allocated. Control influence relates to the way in which the centre reacts to results achieved. Using these two dimensions of planning influence and control influence, Goold and Campbell (1987) identify different management approaches which they call 'strategic management styles.' (See Figure 3.1)

Based on their research in sixteen British companies Goold and Campbell (1987) conclude that *strategic planning*, *strategic control* and *financial control* are the most common styles in the UK. They are explained below.



Figure 3. 1.: Strategic management styles





Strategic planning: the centre works with the business unit managers to develop strategy. It establishes extensive planning processes, makes contributions of substance to strategic thinking, and may have a corporate strategy or mission guiding and co-ordinating developments across the business units. Less attention is devoted to the control process. Performance targets are set in broader, more strategic terms. Annual financial targets are seen as being less important than the longer term strategic objectives.

Strategic control: the centre prefers to leave the initiative in the development of plans to business unit managers. The centre does review and criticise the plans, but uses reviews as a check on the quality of thinking of business unit managers, rather than as an opportunity to give direction. The control process is an important influence mechanism for the centre. Targets are set for strategic objectives such as market share, as well as for financial performance, and managers are expected to meet the targets. On the other hand, budget can only be missed when important strategic objectives are at stake. Strategic control companies combine moderate planning influence with tight strategic controls.

Financial control: the centre's influence is exercised mainly through the budget process. Corporate management's role in developing strategies is limited, and long-term plans are not formally reviewed by the centre. Instead, the centre focuses on a close review of the annual budget. Profit targets are set when the budget is approved, and careers are at

stake if budgets are missed. Financial control companies combine a low level of planning influence with tight financial controls.

Young (1993) found that the size of the corporate centre is associated with strategic management style, therefore, with the extent to which planning and control influences are exercised by the corporate centre. *Financial control* firms tended to have a smaller corporate centre with fewer functions; and *strategic planning* firms had a larger corporate centre with more functions. Compared to the functions and staff of the corporate centre in *strategic control* firms, *financial control* firms had about four fewer functions, and about half the staff, whilst *strategic planning* firms had approximately three more functions, and about three times the staff (Young, 1993: 22).

Hungenberg (1993) posits an alternative explanation to Goold and Campbell's (1987), contending that the corporate centre may adopt one of the three following management styles with different intensity of coordination (vertical and horizontal): operational holding, management holding, and financial holding.

An operational holding centre is effective if intensive vertical and horizontal co-ordination can add substantially to business values. If so, the corporate centre is intensively engaged in decision-making processes at the business level, for instance, by determining detailed goals for costs, production targets or sales volumes. Additionally, cross-business co-ordination and synergy exploitation are main tasks of the centre. The corporate centre controls strategic management and,

to a large extent, also the operational management of business units. Operational holdings can be effective in industries like chemicals, computers, or telecommunications because of existing synergies and the possible similarities of the business systems of various business units.

The *financial holding* centre can be effective if the potential for value enhancement through the horizontal and vertical co-ordination of businesses is small. This description frequently applies to diversified companies, which control heterogeneous business units. A *financial holding* centre leaves operational management entirely to the business units. Furthermore, the centre is involved in the strategic management of businesses only indirectly, by determining financial goals, *e.g.*, profits, cash flow, and staffing top management positions.

A management holding centre ranges between the two extremes of operational and financial holding in terms of the intensity of coordination. A management holding centre does not interfere with the operational management of business units, but still controls strategic management tasks, thereby involving the business units' managements to some extent. A management holding centre can be appropriate in industries like electronics, automotive, or manufacturing, where business units require similar managerial skills, and advantages of shared business systems can be exploited.

Hungenberg (1993) contends that (a) the intensity of the corporate centre's intervention in decision-making processes at business unit

level (vertical co-ordination) and (b) the extent of cross-business coordination (horizontal co-ordination) depend on the role the corporate centre chooses to fulfil. However, the effectiveness of the role played by the corporate centre should also relate to the nature of the firm's business.

The characteristics of the corporate centre in operational, management and financial holdings correspond to Goold and Campbell's three strategic management styles of strategic planning, financial control, and strategic control types, respectively.

Based on the above contentions of Goold and Campbell, and Hungenberg, the relationships between vertical and horizontal coordinations and strategic management styles may be set out as in Fig. 3.2.





Horizontal co-ordination

Source: Adapted from Hungenberg (1993: 69) and Goold and Campbell (1987: 36)

It is evident from Figure 3. 2. that different strategic management styles adopted by the corporate centre lead to different intensities in the vertical and horizontal co-ordination. This difference in its turn determines their respective information-processing needs in either vertical or horizontal directions, and consequently the necessary size of staff to meet these needs.

3. 4. Sociological Explanations of the Size of the Corporate Centre Discussed in Chapter 2

Subsection 2. 3. 2 presented plausible sociological effects on the organisational size, in particular, overstaffing. This thesis, however, is based on the premise that the size of the corporate centre must be mainly associated with its information-processing needs and capabilities.

The following comparison between the sociological approach and information-processing approach points out that informationprocessing needs and capabilities may be more plausible determinants of the size of the corporate centre than sociological considerations. The two approaches will be further discussed in Section 8. 3. 4 on the basis of the results of the research.

The alternative explanation based on sociological elements argues that: (1) even within the limitations of functional imperatives, there is room for a number of alternative organisational arrangements (Rice, 1958; Trist *et al*, 1963; Miller and Rice, 1967); (2) the organisational arrangements must be accepted by different groups of the organisation (Child, 1984); and (3) the general trend in such organisational arrangements is overstaffing (Parkinson, 1958 and 1980; Child, 1984).

In contrast, this research holds that the centre's size is determined by its information-processing needs and capabilities. If the centre's size

is large, and there appears to be overstaffing, it argues that this is due to the need for larger information-processing capabilities.

The centre's size must not differ too much from other competing peer organisations, because, as Child (1984) points out, there is only marginal scope to maintain a preferred but inefficient form of organisation in a firm operating in competitive markets².

If the centre's size is large, then there must be some value-added justifying its largeness. Otherwise, it has the potential for becoming inefficient and the competitiveness of the firm may be lost. Value is added by the largeness of the centre's size, since the larger the size, the larger the information-processing capability.

The information-processing approach relates to the view that organisation consists in information-processing systems coping with the uncertainty (Tushman, 1978) and equivocality (Daft and Lengel, 1986) surrounding the organisation, as discussed in Subsection 2. 4. 1. Information-processing is handled by organisational design (Weick, 1979). The size of the corporate centre is an essential aspect of

² Based on ILO's (International Labour Organisation) data Gordon (1996) argues that, in 1989, the relative size of the US "bureaucratic burden", namely, private non-farm managerial and administrative employees in non-farm employment, had reached more than three times the level in Japan, with US's 13.0 per cent against Japan's 4.2 per cent. Young (1995) contends, on the contrary, that the Japanese firms have corporate centres 5 to 10 times larger than German or UK firms.

Such divergent data may result from the difference in the definition between "bureaucratic burden" and corporate centre. However, it cannot be denied (1) that the size of indirect staff is largely different from one country to another and (2) that the economic cost resulting from such difference is not easily justified by the foregoing sociological elements. Section 8. 3 will further discuss the issue.
organisational design³. When the uncertainty grows larger in the environment due to competition, the centre's size will increase to cope with an increase in information-processing needs.

Thus, information-processing needs and capability are the main determinants of corporate centre size in Japanese firms, although due recognition should be made, in many situations, to the explanatory power of sociological elements.

3. 5. Previous Researches on the Factors Explaining the Centre's Size

Authors also relate the size of the corporate centre to other factors such as diversification strategy, its type of adaptation to environment, the degree of divisional autonomy, and modes of directing the organisation. Before concluding this chapter, these factors are reviewed.

Table 3. 1. compares the factors considered to influence the centre's size.

³ If the size of the centre is severely affected as a consequence of restructuring and re-engineering, its information-processing capacity seems to be deteriorated in the form of corporate memory. The Economist (1996b) reports that, in Japan, companies have gone to unusual lengths to avoid dismissing middle managers and supervisors, because they put great stress on the importance of "tacit" knowledge — hunches, know-how, ideals and experience. Unpublished research by Monitor, a consultancy, cited by the Economist (1996b) found that nine out of ten firms that had outperformed their industries over a ten-year period had "stable" structures, with no more than one reorganisation and no change in chief executive, which helped them to keep their information-processing capacity intact.

Author(s)	Year	Key factors influencing the size of the corporate centre		
Lorsch & Allen	1973	Organisation structure (Conglomerate or vertically integrated)		
Bower	1974	Diversification strategy		
Miles & Snow	1978	Strategy types		
Miles <i>et al</i>	1983	Configuration of internal structure and process		
Goold & Campbell	1987	Planning & control influences (Discussed in Section 3. 2.)		
Cresap	1988	Number of functions (Hands-on or hands- off management)		
Hungenberg	1993	Vertical & horizontal co-ordination (Discussed in Section 3.2.)		

Table 3. 1.: Key Factors influencing the size of the corporate centre

Lorsch and Allen (1973) address the issue of the size of the corporate centre along with the functions performed at the centre, and the degree of divisional autonomy. Their study is based on empirical research of six divisional firms. Two firms were vertically integrated. The other four were classified as what Chandler (1962) called conglomerates — a small corporate centre with nearly all functions placed within the division.

Lorsch and Allen (1973) measured the amount of integrating effort spent at the corporate centre on corporate-divisional relations and inter-divisional relations. The lower-performing vertically integrated firm was shown to spend too little effort in integrating the interdependent divisions. On the other hand, three of the four conglomerates had a small corporate centre and performed few divisional functions, whilst the one conglomerate that operated as a vertically integrated firm was the poorest performer. In general, the vertically integrated corporate centres were larger and performed more corporate functions than the conglomerates (See Table 3. 2.).



Table 3. 2.: Basic characteristics of the corporate centre units in six firms

Conglomerate Firms			Vertically integrated Firms			
<u>.</u>	1	2	3	4	5	6
A. Size-Total	17	20	25	230	479	250
number of				}		
management						
and protes-						
employees				•		
B. Functions						
performed in						
Reference to						
divisions						
1. Financial-	Χρ	Хр	Хр	Xp .	Хо,р	Хо,р
	Va	¥-	V.		¥	V
nlanning	γ	νh	×φ	λφ.	хоф	χο ^τ ρ
3. Legal	αoX	Xo.p.	Xo.n	Xo.n	Xo.n	Xon
4. Industrial	Хо,р	Хо,р	Хор	Xo,p	Xop	Xold
relations						
5. Operations					Хр	
research		¥_		· ·		
6. Markeung		Χρ ·	XD Via	Xo	Хо	Xo
Manufacturing			λp	₩ .		
/industrial		. •				•
engineering						
8. Planning and					Хо	Хо
scheduling of						
output						
9. Purchasing				Y.	Xo	Xo
ing (other than				λφ		
industrial)	•					
11. Research				Xo.p	Xo	Xo
and develop-		•		* .		•
ment	•					

Source: Lorsch and Allen (1973: 148)

Note: An "X" indicates that certain functions in specified areas are performed by the corporate centre unit for the divisions. A "p" indicates that corporate involvement is of a policy-setting nature (*i.e.*, setting policies, advising, providing basic approach). An "o" indicates an operating responsibility for the corporate centre unit (e.g., actually carrying out some purchasing activities for certain divisions).

Lorsch and Allen's finding (1973) relates the size of the corporate centre to vertical integration. Since more information flows in vertical integration, their argument may be seen as an approach to information flow.

- 112

Bower (1974) posits that the size of the corporate centre varies in accordance with diversification strategy. He holds that the corporate centre in firms which are widely diversified in partially-related businesses has a larger number of executives than that of large financially-based collections of separate operating units. The latter have no executives in research and development, marketing, manufacturing, and purchasing and traffic basic functions, yet have a much higher percentage of executives in general management, finance, planning, and public relations.

It may be conjectured that partially-related businesses exchange a larger amount of information than large financially-based collections of separate operating units. Therefore, Bower's argument could relate to the information flow approach mentioned in the preceding section.

Miles and Snow (1978) and Miles *et al* (1983) identify various types of organisations and, with them, corporate centres. They are *defenders*, *prospectors*, *analysers*, and *reactors*. These types result from differences in the strategy and configuration of internal structure and process of the firms. The size of the corporate centre for these types is not explicitly mentioned in Miles and Snow (1978), whereas Miles *et al* (1983) suggest that the prospector has a large corporate centre dominated by marketing and research and development experts.

Defenders are organisations which have narrow product-market domains. The corporate centre in this type of organisation is highly

expert in its organisation's limited area of operation but does not tend to search outside its domain for new opportunities.

Prospectors are organisations which almost continually search for market opportunities, and they regularly experiment with potential responses to emerging environmental trends. The corporate centres of these organisations are the creators of change and uncertainty to which their competitors must respond.

Analysers are organisations which operate in two different types of product-market domain, one relatively stable, the other changing. In its more turbulent areas, the corporate centre watches its competitors closely for new ideas, and then rapidly adopts those which appear to be most promising.

Reactors are organisations in which the corporate centre frequently perceives change and uncertainty occurring in its organisational environment but is unable to respond effectively.

No one type of corporate centre is proposed as superior to others, as several have been employed successfully at the same time by different firms in the same environmental setting. However, each type must conform with an appropriate configuration of internal structure and process.

In the argument of Miles and Snow (1978) and Miles *et al* (1983), prospectors use 'hands-on' management from the centre than other

types of firms. It follows, then, that vertical information flow in prospectors is more intense than in other 'hands-off' organisation styles. Thus, the argument points to the possibility that the difference in the corporate centre's functions and its size might be influenced by vertical information flow.

On the basis of interview and mail survey with 45 leading British firms, Cresap, a management consultancy firm, found (1988) four different types of corporate centre reflecting modes of directing the organisation - targeting, guiding, directing, and running corporate centres (See Table 3. 3.).

The fundamental management philosophy of the *targeting* corporate centre is 'hands off', underpinned by a belief that this philosophy has motivational benefits for the operating units. A very clear separation between the role of the corporate centre and the activities of the subsidiaries is emphasised. The corporate centre is viewed as 'an overhead'. The *targeting* corporate centre controls the performance of the firm by setting financial targets and monitoring performance against these targets. The mission and fundamental objectives of the firm are decided at the corporate centre. Strategy formulation is the responsibility of the operating units and is not initiated from the corporate centre. The provision of services and support to the operating units is not seen as a role for the *targeting* corporate centre.

The guiding corporate centre espouses a 'hands off' philosophy, but its approach is marked by a greater degree of involvement with the operating units. So while operating decisions are largely left to the individual business units, which are set financial targets and held accountable for their performance relative to these, the philosophy is supported by a desire to 'help' the operating units, and to foster the sense of belonging to an 'active corporate group'. The *guiding* corporate centre sets out to influence decision making at the operating unit level via the provision of 'advice and co-ordination'. Typically, such a corporate centre develops the overall company strategy but is also involved in a dialogue with the operating units in helping them to develop their individual plans within the context of the overall strategy.

In contrast to *targeting* and *guiding* corporate centres, the *directing* corporate centre takes more of a 'hands on' approach in management philosophy. It not only helps the operating units to develop strategic plans, but also provides directional initiatives from the centre, and gets involved in the implementation of business strategies and the integration of major business decisions. The *directing* corporate centre takes the view that the corporate centre is best placed to undertake certain activities on a centralised basis. Therefore, it not only co-ordinates common activities but, in some cases, may undertake entire functions centrally for the whole organisation. Management training, purchasing, and research and development are typical examples of centralised functions. The provision of services and support to the operating units is an important role of the corporate centre.

The *running* corporate centre is the most involved of the four types of corporate centre. It is very much 'hands on' in orientation. There is

often little effective separation between the corporate centre and the business. The role of the corporate centre is integral to the firm's activities. The centre develops and monitors plans, policies, and guidelines, makes major operating decisions and, to all intents and purposes, runs the firm. The traditional distinction between "line" and staff is often neither clear nor relevant. The provision of services to the operating units is a key function for such a corporate centre.

Table 3. 3.: Four types of the corpora	ate centre
--	------------

.

Tasks	TARGETING	GUIDING	DIRECTING	RUNNING
Set	x	x	x	x
fundamental				
objectives				
Monitor	x	x	x	x
tinanciai				
Conduct	~	Y		
external	^	^	×	X
relation-			•	
ships				
Meet	x	x	X	x
statutory				
obligations				
Manage	X	x	x	Χ
senior				
executive				
resources				
Co-ordinate		x	x	x
Dusiness				
Dravido	·····			· · ·
Provide		х	x	x
Integrate			v	· · · · · · · · · · · · · · · · · · ·
husiness		•	×	X
decisions	•			
Control				x
business				, j
decisions		, · · ·		•
Size as a	0.1-0.7%	02-3.4%	1.8-9.1%	3.7-27.5%
percentage				
of total			· · · · · · · · · · · · · · · · · · ·	. ·
company				
staff				

۰.

Source: Cresap (1988: II-5)

Cresap (1985) contends, on the whole, that the 'hands-on' approach in management philosophies reflected in a greater number of tasks (Table 3.2) leads to a larger size of the corporate centre. As the 'hands-on' approach suggests a higher vertical information flow between the centre and lower management, Cresap's contention thus suggests a relationship between the size of the corporate centre and vertical information flow.

The foregoing discussion suggests that diversification strategy, organisation adaptation, the degree of divisional autonomy and modes of directing organisation could all influence the size of the corporate centre. However, a closer examination of these influence factors leads to the conclusion that they affect the relative weight of vertical and horizontal co-ordination in the corporate centre. The resulting pattern of information-processing needs exerts its influence on the functions and size of the corporate centre. Thus, all these factors could be equally explained in the light of the relationship of the corporate centre's size with information flow and information-processing needs⁴.

⁴This interpretation of the multiple influence factors suggests that these factors affecting the centre's size might affect information flow in the organisation, which influences the centre's size.

3. 6. Summary

This chapter has addressed the relationship between the size of the corporate centre and the vertical information flow between the corporate centre and lower management. Two key concepts were analysed: the functions performed by the corporate centre, and the factors influencing its size.

It examined the functions performed by the corporate centre with reference to those relating to information flow and processing. In dealing with these questions, the shaping of the corporate centre was analysed in a historical perspective. The corporate centre has evolved into an agency which discharges the functions of planning and resource allocation, controlling and auditing performance, and central service provision. Planning and control influences are closely related to vertical information flow.

In addition to this "functionalist" relationship between information needs and the size of the corporate centre, the sociological elements influencing them were examined. The overstaffing problem was briefly discussed. Comparison was made between the argument of this research based on information-processing capabilities and the explanation based on sociological elements. It was concluded that information-processing needs and capability are the main determinants of corporate centre size in Japanese firms, although due recognition was made, in many situations, to the explanatory power of sociological elements.

Authors who have empirically analysed the different types and sizes of corporate centres were reviewed. The factors selected by these authors in their analysis of types of corporate centres and sizes were discussed.

To sum up the discussions of this chapter: planning and control influences are important factors affecting the size of the corporate centre, since the size of the corporate function is a function of the information-processing capacity needed to cope with vertical information flow (resulting from the relative focus between planning and control).

The literature review demonstrated that no reported empirical research has explored the association between the size of the corporate centre and vertical information flow related to planning and control influences.

On the other hand, it was pointed out in Chapter 1 that the size of the corporate centre may be influenced not only by vertical, but also by horizontal information flow. It is, therefore, necessary to examine the literature regarding the relationship between the size of the corporate centre and horizontal information flow, before setting out to design our research method and samples, etc.

Accordingly, Chapter 4 deals with the relationship between the size of the corporate centre and horizontal information flow resulting from linkages among divisions or departments.

CHAPTER 4: HORIZONTAL INFORMATION FLOW AND THE CORPORATE CENTRE

4. 1. Introduction

The preceding chapter discussed the relationship between the size of the corporate centre and the vertical information flow resulting from planning and control influences. The next step is a review of the literature and reported research on the relationship between the size of the corporate centre and the horizontal information flow resulting from the linkages among divisions and departments.

This chapter is divided into four sections. Section 4. 2 examines the view advanced by Kanter (1983) about the creation of information flow in firms.

Section 4. 3 presents an analysis and synthesis of the literature on Japanese management in the light of Kanter's view. It is contended that Japanese firms are equipped with elements facilitating the creation of information flow.

Section 4. 4 discusses the horizontal information flow at lower management and especially the shop-floor level, and the mechanism for its co-ordination. Horizontal information flow at lower management levels is then reviewed in its relationship to the centralised administrative functions at the corporate centre. The communications perspective of Japanese management will also be discussed.

The substance of the discussion in this chapter is recapitulated in section 4. 5. It is contended that the size of the corporate centre is associated with the horizontal information flow because of the centralisation of administrative functions needed for its functioning.

Based on the contention described in Section 1. 3, Section 4. 6 discusses the implications of the information-processing and information flow perspective for practising managers. These implications will be further discussed in Chapter 8.

Section 4. 7. summarises the chapter.

4. 2. Network Formation for Information Flow

Kanter (1983) maintains that network formation is important for the exchange of information among divisions or departments. She enumerates several elements as the principal devices that aid network formation in firms: *frequent mobility* (including lateral moves), *employment security, extensive use of formal team mechanisms*, and *complex ties permitting cross-cutting access*.

A brief description of each of these devices outlined by Kanter (1983) is set out below.

124

ent Sana ora mem.

Mobility across jobs

Mobility — circulation of people across jobs — is a prime networking device. Firms fostering information flow frequently move managers from one position to another. Such moves are lateral rather than vertical. In some firms, an average job tenure is about two years. In others, people move, for example, from the corporate centre to district sales office, from finance to manufacturing and from personnel to operations management.

Those who move clearly have an advantage or opportunity in terms of network breadth over those who do not. The organisation's *opportunity* structure — who moves, out of what jobs, and how often — has a direct bearing on its *power*¹ structure because of the impact on networks.

Frequent mobility is an opportunity rather than a threat to people only if it is coupled with overall employment security.

Employment security

The security that comes from the employees' expectation of permanent employment in the organisation creates not only higher flexibility and lower resistance to change, but also a willingness to invest in the future.

¹Kanter (1982: 203) equates power with access to information, resources, and support. Company data and company funds, access to high-level decision makers and company experts are some examples of power.

Employment security is concerned not only with one's own continued employment but also with that of others (Womack and Jones, 1994). In an insecure employment situation, people fear tackling projects that do not guarantee short-term results, but also fear investing their support in the projects of their colleagues for the same reason.

Teams

The frequent use of integrative team mechanisms at middle and upper levels not only encourages the immediate exchange of information as well as support but also creates contacts to be drawn on in the future. The organisational chart with its hierarchy of reporting relationships and accountability reflects only one reality. The "other structure," not shown on this chart, is an overlay of flexible, *ad hoc* problem-solving teams, task forces, joint planning groups, and information-spreading councils.

In these flexible and *ad hoc* teams, members are drawn from a diversity of sources. Many points of view are brought to bear on a problem. It is not reducing risk by spreading responsibility that is sought, but rather the better ideas that come from a clash and the integration of different perspectives.

Collaborative and consultative rather than unilateral decisions are the expressed norm in these teams. It is expected that, thanks to such an integrative mechanism and company philosophies encouraging teamwork, managers will not reach decisions alone, without consulting others.

Complex ties and cross-cutting access

A formal structure acknowledging complex ties forces a great deal of inter-unit contact between managers. Although managers nearly everywhere complain about excessive meetings, such occasions provide them with opportunities to develop formal and informal working relationships with persons from other functions or disciplines. This encourages coalition formation, and helps managers to mobilise support or resources to complete an objective. It discourages over identification with one area and divisive politics.

Legitimacy of cross-cutting access helps managers to cross formal lines and levels in the organisation to find what they need — vertically, horizontally, or diagonally — without feeling that they are violating protocol.

Matrix designs, though not essential for cross-cutting access, can be helpful in legitimising it, for the organisation chart shows a number of links from each position to others. There is no "one boss" to be angered if a subordinate manager goes over his head or around to another area; it is taken for granted that people move across the organisation in many directions. Similarly, formal cross-area and cross-

hierarchy teams may provide the occasion and the legitimacy for reaching across the organisation chart for direct access.

Recapitulating, this section has argued that devices such as frequent mobility, including lateral moves, employment security, extensive use of formal team mechanisms, and complex ties permitting cross-cutting access help network formation, which enhances horizontal information flow.

The review of the literature on Japanese management in the next section will highlight the fact that the devices considered by Kanter as being instrumental for forming network to facilitate horizontal information flow are commonly found in Japanese firms.

4. 3. Network Formation for Information Flow: Japanese Management

This section reviews the literature on Japanese management in order to see how well Japanese firms are equipped with the afore-mentioned devices aiding network formation, and, therefore, horizontal information flow.

Mobility across jobs

Because of the life-time employment system which will be discussed in the following paragraph, Japanese firms tend to look ten to twenty years ahead to determine this year's recruiting profile (Lauenstein, 1985). In addition to this long-term recruitment policy, job rotation is undertaken periodically to produce generalists (Odaka, 1986; Ouchi, 1981; Ouchi and Jaeger, 1978). Levine and Kawada (1980) also found how and why Japanese firms set up internal systems for key workers, to build up a highly versatile, loyal core of permanent employees in a particular firm. Standardised training is given to ensure that everyone is at least of average competence², which is essential to job mobility.

Employment security

The frequently-cited Japanese management practice of life-time employment (Abegglen, 1958; Hanai, 1979; Odaka, 1986; Tsuda, 1987) is a system with a high degree of job security in which the compensation of employees is largely determined by length of service, rather than by their particular job or their output in a given level³. A good part of their compensation is in the form of a bonus related to the overall success of the firm (Abegglen and Stalker, 1985).

Some authors believe that firms use life-time employment to create a sense of communal and familial relationships, but its real function might be to increase managerial power and control over performance. Long-term job security discourages employees from seeking for a job

²This does not mean that Japanese firms do not foster a group of élite employees. Hanada (1993) discovered that élite staff are chosen by the time they have reached the position of section manager (kacho) and groomed for a career in the firm. ³It is argued that Japan and the US are actually quite similar in terms of the amount and distribution of long-term steady employment. In Japan, steady employment occurs in the first half of a working life, whereas in the US it occurs in the second (Johnson, 1980).

elsewhere. Under this circumstance, employees tend to be obedient to such requirements from their companies as excess work, over-time, etc. Thus, rhetoric based on the concept of the firm as a "household" creates an image of culture-bound processes which conceals the reality of power beneath the surface (Fruin, 1980; Van Wolferen, 1989; Sullivan and Peterson, 1991).

Still, others believe that at least part of any commitment by workers found in the Japanese manufacturing economy derives from the wider implementation of certain elements of welfare corporatist control⁴ (Dore, 1973; Lincoln *et al*, 1990; Ariga *et al*, 1992).

On the whole, this internal labour market system has had substantial benefits for the firms, which are generally large firms, that developed it. Levine and Kawada (1980) stress that these management practices in Japanese firms are rational choices by managers interested in financial returns, not the repetition of traditional solutions. Thus Levine and Kawada confirm Fruin (1980) in the sense that the Japanese system of life-time employment did not derive from family paternalism or cultural tradition. Values and traditions were used as rationalisations after the event.

Leibenstein (1984) contends that beliefs about the system may be just as important as the facts themselves. Therefore, the important thing about the life-time employment ideal is not whether people work for a

⁴Novel philosophies and practices will be introduced by managers if they are compatible with the culture which enables welfare corporatist control (Marsland and Beer, 1983).

given firm for their entire life-time, but whether they believe that this is likely to be the case and behave accordingly.

Teams, and complex ties and cross-cutting access

The consensus-building process, one of the characteristics of Japanese management, seems to serve the same purpose as integrative team mechanisms and complex ties do respectively for the encouragement of information flow at middle and upper levels and inter-unit contact between managers.

Consensus-building relates to the fact that Japanese management falls in a category generally described as the "organic type". Such management is performed by consensus (Yang, 1977). Japanese managers have fully accepted the fundamental premise of participative management, namely, that employees are capable of contributing and desire to contribute to the organisational requirements of a supportive supervisory climate (Cole, 1980).

In this climate Japanese try their best to make a decision by unanimous agreement without letting any difference of opinion come out into the open⁵. To this end, they usually resort to *nemawashi*, which literally means preparing for transplanting a large tree by cutting

⁵The strategic management of Japanese firms is characterised by the interaction of different management levels rather than being based solely upon executives' judgement (Kagono *et al*, 1985), which, however, does not mean that chief executives abdicate their leadership in strategic management (Shimizu, 1986). On the contrary, in Japanese firms which have risen to global leadership over the past twenty years, top management plays an indispensable role by creating "strategic intent" (Hamel and Prahalad, 1989).

off all but major roots a year or two before it is to be transplanted (Misumi, 1984). *Nemawashi* refers to the "political" process by which an unofficial understanding is reached before any final decision is made on a particular matter (Howard and Teramoto, 1981). In this decision process, decision makers comprehensively survey possible alternatives and resolve conflict (Takahashi and Takayanagi, 1985).

Combined with *nemawashi*, Japanese managers use the *ringi* procedure⁶. In this procedure a person holding a lower rank in the hierarchical system of an organisation drafts and introduces a proposal and circulates it through the chain of command for final approval. Since the proposal for decision-making is initiated by a lower-ranking person, this decision procedure is a "bottom-up" system (Misumi, 1984).

It has been pointed out, however, that the Japanese managers may prefer less participation (Daley *et al*, 1985), and that, instead of a mere bottom-up type, the decision procedure in Japanese firms is a "U" shape: it starts from top management and the views of lower rank employees are absorbed by top management (Kono, 1984; Shimizu, 1986).

A varying level of overall centralisation of authority in decision-making has been also reported. Spatial dispersion into multiple sites and a higher level of automation of technology increase the centralisation of authority; large company size, the increasing number of levels to which

⁶Takamiya (1980) holds that since the practice of delegating power has become common, the top management's approval through *ringi* has become unnecessary, and *ringi* has come to be used only as a means of reporting to top management.

size gives rise, and greater task variety have the opposite effect, namely, they lower somewhat the levels at which decisions can be made (Marsh, 1992).

In the light of Kanter's (1983) work, this section has determined the extent to which Japanese firms are equipped with the elements aiding network formation, and, therefore, information flow.

The following section discusses horizontal information flow at the lower management level in Japanese firms and the mechanism for its co-ordination. Horizontal information flow at lower management level is then reviewed in relation to the centralised functions at the corporate centre.

4. 4. Information-Processing Structure of Japanese Firms

In line with the argument in the preceding section to the effect that Japanese firms are equipped with those devices necessary for forming network and creating information flow, Aoki (1986; 1988; 1993) suggests that the efficiency of Japanese firms may be largely due to the quality of the information-processing structure within them⁷. This structure is characterised by rapid *intrafirm* horizontal communication, which is necessary for the co-ordinated adjustment of constituent operations in reaction to "global market shocks", *e.g.*, market fluctuations, as well as the decentralised handling of "local shocks", *e.g.*, machine breakdown, to minimise their impact on the system as a whole.

The quality of such an information-processing structure depends to a great extent upon the information-processing capabilities of the employees who operate the system. The way in which such human resources can be accumulated and maintained within a firm is a question of providing employees with the incentive to develop the skills, knowledge, expertise, and co-operative attitude needed to operate the horizontal informational structure⁸ (Aoki, 1988). This view is substantiated by Carney (1986) from the perspective of

⁷In a similar vein, based on a survey of 1000 firms in Japan and the US, Kagono *et al* (1985) concluded that Japanese firms employ numerous mechanisms which encourage frequent interactions horizontally and vertically, and place tremendous emphasis upon personal interactions within and between groups.

⁸This is also an example of "invisible resources," as termed by Itami (1983). Examples of invisible resources include: know-how of marketing, brand name, management skills, corporate culture, and so on. These resources are important mainly because their availability in the market is much more limited than other "visible resources" such as physical or monetary resources.

communications process in companies. He claims that Japanese management is characterised by advantages contributed by "little brains" that are autonomous work groups as well as by the empowerment of "self" (*i.e.*, individual worker).

Accordingly, based on Aoki's foregoing argument, the first part of the present section (4. 4. 1) discusses the information-processing structure of Japanese firms. Whilst the discussion mainly refers to shop-floor level operations, the information systems that are considered exhibit a generic element of a co-ordination pattern operating throughout different levels in Japanese firms. The second part (4. 4. 2) discusses the generalisability of shop-floor level information flow. This is followed, in the third part, by a discussion of the incentive schemes used by Japanese firms to develop skills and co-operative attitudes among employees.

4. 4. 1. Information-processing structure at the shop-floor level

The way Japanese firms organise work at the shop-floor level, and coordinate operations among shops, influences their informationprocessing structure.

Flexibility in job design

Job classification in Japanese firms is simple and broad: demarcation is fluid and ambiguous and assignments are flexible. The degree of specialisation is low. For example, Japanese firms do not have a specialised reliefman to cope with absenteeism. Instead, they rely on mass relief or the *ad hoc* reassignment of jobs at the discretion of the foreman. Inspection positions are often rotated among operating workers on the shop-floor. As a result of broad job classification and flexible job assignments, workers tend to undertake a wide range of jobs and are given an opportunity to develop different skills.

Some factories have taken the fluid and flexible approach a step further by introducing regular job rotation. In most instances where the job rotation scheme is used, even inexperienced workers may be assigned to a very difficult job, in which case the most experienced workers may assist in a side-by-side position. Job rotations allow every worker to become familiar with the whole work process at the shop. The job rotation system therefore facilitates "knowledge sharing" among workers in that the knowledge possessed by a single worker is extended beyond the sphere of a particular job. Thus, there is considerable overlap in the knowledge of individual workers of different status on the shop-floor.

The wider range of job experience in general and the regular rotation system in particular sacrifice some economies of specialisation. The loss in economies of specialisation is compensated for by the following benefits, however. In the rotation system, the diffusion of skills from senior to junior workers may be continuously facilitated on an informal basis through the assistance and advice of senior and imitation by junior workers. Those nurtured in a wide range of skills may be able to understand why, for example, defective products have increased, and may be able to devise and implement measures to cope with the problem and thus prevent it from recurring. This can be done without much "outside" help from specially designated craft workers, reliefmen, repairmen, and other specialists. Thus, a wide range of job experiences helps to develop workers' intellectual skills to cope with irregular events and beyond those necessary for routine operating tasks. Therefore, from the perspective of the information system, benefits of collective learning consist in enhancing workers' capacity for processing information relevant to shop-floor efficiency (Womack and Jones, 1994).

Inter-shop co-ordination

Inter-shop co-ordination is related to workshop organisation. Whereas job specialisation on the shop-floor in typical Western firms is coupled with hierarchical co-ordination of shops by managers as specialists, autonomous problem solving on the shop-floor in Japanese firms is coupled with horizontal and direct dealing among shops in inter-shop co-ordination. In other words, in Japanese firms, the co-ordinating and operating tasks at the shop-floor and the inter-shop levels tend to be integrated.

The horizontal and direct dealing among shops in inter-shop coordination can be most effective when the inevitable unexpected event occurs. An unexpected event could be a change in the final demand (demand shock), or an in-process shock in a particular shop or between particular shops. Its repercussions could be far-reaching, e.g., quality defects in part production, malfunction of machines, breakdown in transport of semi-products. It is normally impossible to specify exante inter-shop transaction plans free of unexpected events, as each transaction plan is contingent upon the occurrence of a probable particular emergency. Thus such events often have to be dealt with ex post. In Japanese firms the centralised scheduling of production provides a general framework only for a certain period of time. As a supplement to the centralised plan, horizontal and direct co-ordination among interconnected shops governs the actual transfer of materials, parts, and in-process products. This co-ordination fine-tunes actual final output for changing market conditions and in-process

emergencies. The lack of a central office to control and expedite the flow of materials among shops is a conspicuous feature of the assembly factory in Japan.

The kanban system

The kanban system⁹ provides an example of this horizontal and direct co-ordination. The term kanban refers to a card placed in a vinyl envelope. In implementing the daily production schedule, the final assembly line places a production-ordering kanban containing the data about the needed quantity of each type of part or semi-product on a post adjacent to the relevant inventory store whenever it withdraws its inventory. Upstream shops are supposed to adapt their production according to demands by their downstream shops, as indicated by the kanban.

In the kanban system, a tentative production schedule may be worked out by the central planning office on the basis of market forecasts at a regular interval — for example, once every two weeks — and every shop is informed of its production plan. But this centralised schedule provides each shop with only a general production guideline for that period. Information regarding actual customer demands for varieties of products is used for fine-tuning the actual production schedule in a

⁹Differences in manufacturing practices in the US and Japan discussed were: Quality management (Garvin, 1983; Hayes, 1981; Juran, 1978), quality circles (Cole, 1980; Takeuchi, 1981), inventory control (Abernathy *et al*, 1981; Schonberger, 1981), and manufacturing innovation (Cusumano, 1988) have been studied.

However, it is the *kanban* system that is considered to be instrumental in the success of Japanese manufacturing industry (Monden, 1983; Womack *et al*, 1990).

shorter time frame. The important point is that this up-to-date production schedule, based upon a two to three day short notice from sales department, is fed only into the final assembly line; thus, because of the use of *kanban*., communications necessary for finetuning production at upstream shops, to match the daily schedule of final outputs, need not be channelled through the administrative office.

Although the *kanban* system may appear to be a rather crude information system, it is an effective mechanism that facilitates an intricate system of production which may require multiple component units to respond to unpredictable consumer demand for a variety of final outputs, while keeping the amount of in-process inventory as low as possible.

An essential feature of the kanban system is that the fine-tuning of production is performed in response to evolving market conditions without the intervention of the administrative offices above the operating units. Thus, the system exhibits a generic element of a co-ordination pattern operating in Japanese firms — a direct, horizontal co-ordination among operating units, which is often informal and bypasses hierarchically layered administrative offices.

e multi altreament a se 🗳

Nevertheless, two sets of technological and market conditions of the firm ought to be considered under which manufacturing operations can be co-ordinated in the *kanban* system for productive and informational efficiency.

First, the function of the *kanban* system is to integrate the flows of production and information for the purpose of responding to market fluctuations with minimal inventories. If a product variety is limited or market demand is very stable, the saving of inventory cost by fine-tuning final outputs in response to market signals through horizontal co-ordination may be trivial. On the other hand, horizontal co-ordination without the centralisation of information may not be effective for large changes in demand. According to Monden (1983), demand variations of about 10 percent can be handled by changing only the frequency of *kanban* transfers. However, larger seasonal changes in demand, or a larger change in actual monthly demand over the predetermined plan, requires that all of the production lines be re-arranged administratively.

Second, the capacity of the worker to cope with changes in product composition and local emergencies may be specific to the global framework established through strategic business decisions of top management such as product mix and factory design. Such capacities may be effective to cope with continual and incremental changes within that framework, but not with drastic changes in market and other conditions. It may be very costly to develop versatility in workers to enable them to cope with a very wide range of changes, instead of recruiting specialists from markets as the need arises. On the other hand, if the market and other environments are very stable, workers' information-processing capacities may not be used effectively and may become redundant resources. Therefore the development of wide-

ranging skills may be most effectively used for continual environmental changes.

Efficiency of horizontal co-ordination

The efficiency of horizontal co-ordination like the *kanban* system involves two aspects. First, it concerns administrative efficiency. In response to market fluctuations, machine breakdown, etc., shops can function as communication nodal points and require no administrative apparatus on top of the production structure for operational coordination. Thus, manufacturing operation and co-ordination are integrated.

Second, it refers to "localised horizontal co-ordination." It is an important characteristic of Japanese firms that no manager specialises in inter-shop co-ordination; the task of inter-shop adjustments that cannot be automated by regular *kanbancirculation* is relegated to direct negotiation between foremen of the relevant shops.

A possible advantage of localised horizontal co-ordination vis-à-vis hierarchical co-ordination would be the saving of time needed for communication and computation via the administrative office and the quick use of "on-the-spot" information concerning emergent events. For example, defects in in-process products, breakdowns in transport, and so on, are most likely to be recognised first by the directly relevant shops and they can react or propose to react quickly. To the extent that

shops understand the nature of their own work processes as well as those of neighbouring shops, they may be jointly able to devise a good solution to emergent problems by making use of first-hand information about what is actually happening at the shop level.

However, there are some prerequisites for localised horizontal coordination to function well. To achieve a local optimal solution among the shops concerned it is necessary that individual shops internalise goals appropriately "decomposed" from the overall organisational goal, such as local cost minimisation, if the overall objective is the minimisation of total costs in the factory. In addition, if shops in localised horizontal co-ordination want to reach the best solution, it is necessary for them to obtain information on problems and to be equipped with problem-solving capacities at the shop level. If shops do not fulfil these requirements, then, the efficiency of localised horizontal co-ordination in response to market fluctuations, machine breakdown, etc., is a relative one.

In brief, the internal efficiency of Japanese firms may be largely due to the quality of the information-processing structure institutionalised within it. That structure is characterised by rapid *intrafirm* communication. The quality of such an information-processing structure depends on the information-processing capability of the workers. Accumulating and maitaining such human resources within the firm is a matter of providing employees with the incentive to develop the skills, knowledge, expertise and co-operative attitude.

Centralised personnel administration

The centralisation of company-wide personnel administration in Japanese firms may explain why Japanese firms are able to delegate more decision-making authority to lower management in operating and co-ordinating spheres. Although information-processing and decisionmaking are dispersed and decentralised in Japanese firms, employees are monitored for the development of contextual skills over their entire career in the firm, and their records are put into the organisational memory formally and informally. Those employees who have established a reputation over time for developing contextual skills, which they still appear capable of developing, may be given better opportunities for promotion. Employee behaviour is thus moulded to be compliant with the organisational orientation in the long run.

In Japanese firms, personnel administration needs to be centralised not only for the purposes of control, but also for the efficient utilisation of the contextual skills of the worker nurtured in the nonhierarchical information-processing structure. The centralised approach to personnel administration is necessary, for instance, for a systematic in-house programme of career development or for the redeployment of workers released from an abandoned work process. This is related to the *internal* labour market which refers to a system of inter-related and institutionalised practices by which a sector of the job market and its related prices or earnings has become closed to potential employees who do not possess certain attributes (Loveridge,
1983). Their forms and rationales display a cross-national diversity which indicates a difference in employer strategies and employee response (Loveridge, 1983). In the case of Japan, the centralisation of personnel management creates a group of core employees, who are protected by a company-wide work union.

Since personnel assessment is a standardised procedure and files are kept and utilised centrally, the personnel department occupies a strategic position in the management of Japanese firms.

Thus it appears that the decentralised approach to information processing in the operational and co-ordinating sphere needs to be associated with, and complemented by, a centralised approach to personnel administration for the purposes of organisational effectiveness.

4. 4. 2. Generalisation of the information-processing structure

Two more points need to be added to the above argument. First, the discussion of the centralised approach has been limited to personnel management. Further scrutiny will determine whether this central approach is applied to other operational administrative functions. Second, as the argument has so far revolved around the shop-floor level co-ordination, it is necessary to examine whether the horizontal co-ordination prevailing in the shop-floor level can be extrapolated to middle management.

Centralised administration

Kono (1984) observes that centralisation is not limited to personnel management in Japanese firms. In a competitive environment, strong marketing and production functions are necessary to ensure that the products are of high quality and low cost in order to strengthen the sales channel. These functions may be more vigorously performed if there are strong centralised staff departments in the corporate centre which collect wider global market information, and co-ordinate and integrate different divisions or departments. As a consequence, the corporate centre in Japanese firms is strong, irrespective of either the level of diversification or the functional or divisional organisational structure¹⁰.

Horizontal co-ordination at middle management

Concerning the general applicability of the co-ordination mechanism and information flow, Aoki (1988: 24) noted that:

The kanban system can be only used to control production processes with special technical characteristics. However, the system exhibits a generic element of a co-ordination pattern operating in Japanese firms, the direct, horizontal co-ordination among operating units, which is often informal and bypasses hierarchically layered administrative offices.

Kanai's (1991, 1993) survey of middle management in Japanese firms points to a similar horizontal co-ordination pattern. Kanai examined two "task contingencies" that middle managers of Japanese firms face: task uncertainty and task dependence. The former has to do with the gap between the amount of information one has in completing the task and what one already has from one's organisational position (Galbraith, 1973). The latter is defined as the gap between information, resources and other supports that are required to complete the task and those which an organisational position automatically brings to a manager. Since power is instrumental in mobilising information,

¹⁰The Japanese approach to strategic issues is soft, qualitative and sometimes almost spiritual. The responsibilities of a Japanese CEO are to articulate a vision of the nature of his firm, the direction it will pursue, and its long-term goals, expressed in terms of markets, growth, technologies, products and industry position (Lauenstein, 1985; Hammel and Prahalad, 1989). These statements of vision, direction, and goals represent the official policy of the firm, which all employees are expected to support (Lauenstein, 1985).

resources, and other supports, task dependence is, in essence, the indicator of the degree of power gap (Kanai, 1993).

Kanai's (1991, 1993) finding is that as a contingency confronting middle managers, task dependence is much more important than task uncertainty. This means that a manager who, for example, initiates and implements a brand new management control system on his own initiative is quite likely to suffer from power gap. In implementing something new to the organisation, albeit compatible with the existing organisational goals, it is necessary to obtain information which is not automatically available to his organisational position. The manager has to mobilise resources beyond the discretionary budget, and gain informal support from peers in other departments as well as from the bosses. Thus, horizontal co-ordination is keenly needed at middle management level.

Nonaka (1990) found this same information co-ordination pattern in the innovation process of Japanese firms. Nonaka postulates that information redundancy is fundamental to the innovation process. Information redundancy refers to a condition where some types of excess information are shared in addition to the minimal amount of requisite information held by every individual, department, or organisation in performing a specific function. While this excess information could be considered superfluous from the standpoint of large-scale information processing efficiency, from a qualitative standpoint such an excess enriches the "meaningful" functions of the organisation. When excess information is shared within the

organisation, it clarifies the meaning of the specific information required within the purview of distinct individuals and groups. Information redundancy stimulates the creative powers of information and is linked to the generation of information with new meanings.

Nonaka (1990) emphasises that in the innovation generation process in Japanese firms, the concept of the division of labour, in the sense that areas of activities are clearly specialised, is not completely adopted. Rather, every phase of innovation generation is loosely connected and overlaps, expanding and contracting with the unrestricted elasticity of diversity. As each phase is autonomous yet loosely linked, the innovation generation process induces interaction between phases and promotes an abundant sharing of information.

Structurally, Japanese firms seem to attempt to create overlaps or linkages between divisions or departments. Less than half of 874 firms surveyed by the Japan Development Bank (1989) had adopted a divisional form with complete business autonomy. Even those that had divisionalised, more often than not, had created incomplete divisions, namely, divisions focused on the marketing function but without the production function, or divisions focused on production to the exclusion of the marketing function, etc.

The centralised approach to administration and decentralised one to horizontal co-ordination in Japanese firms are better depicted "in the round" if contrasted with the different approaches in American firms (See Figure 4. 1.). Aoki (1988) holds that American firms are embedded in a societal framework that emphasises egalitarianism, and consequently workers are more mobile, moving between firms in search of better individual opportunities. In this environment, management's authority is intrinsically insecure and may have to be reasserted through the institutionalisation of a hierarchical informationprocessing structure within the firm, with the threat of discharge as an important means of discipline for supervisors. Conversely, the development of a functional hierarchy based on fine job specialisation and a well-defined classification of jobs has helped create standard job markets, external and internal to the firm, and has encouraged workers to become more mobile. Such a development may be called the decentralisation of personnel administration.



		Information		
		Centralised	Decentralised	
Incentives for personnel .	Centralised		Japanese firms	
	Decentralised	American firms	•	

Source: Adapted from Aoki et al (1989: 56)

4. 4. 3. Incentive scheme: the "ranking hierarchy"

To be effective in the horizontal co-ordination of operations, workers must not only be able to handle a specialised task, but must also be capable of bargaining smoothly with their peers in inter-shop issues and skilfully mediating between the conflicting demands of subordinates. Therefore, such workers need to develop an understanding of and insight into the full range of organisational activities. Informal personal networking with colleagues across shop, unit or departmental boundaries serves as a valuable asset since it allows the individual to collect information, identify problems, and reach agreements in the firm. Such a broad understanding and networking can be developed by experiencing various operational and managerial tasks within the context of a single firm.

It may therefore be concluded that Japanese firms assess the value of an employee, both blue-collar and white-collar, by his or her *contextual skills* — a relatively wide range of integrative skills that have been developed and are useful in a non-hierarchical information-processing structure in which information flows horizontally.

Japanese firms have developed a system of incentive schemes with which to evaluate and reward employees on a long-term basis for competing in the development of such contextual skills, while using them co-operatively. Under such schemes, the firm can identify slow learning, low productivity, low motivation, and uncooperative workers by actual observation and differentiate them in pay and status over the

long run, while attempting to lock in fast-learning, highly productive, highly motivated, co-operative workers by discouraging them from quitting in mid-career. This system of incentive schemes has three important elements: (1) the wage system, which combines seniority and merit rating, (2) internal promotion discriminately applied to employees on the basis of merit rating¹¹, and (3) a lump sum payment at the time of (mandatory) separation.

The incentive scheme systems in Japanese firms are characterised by a notion of *ranking hierarchy*. This is a system of ranks wherein the remuneration paid to an employee depends upon his individual rank and in which his internal promotion from lower to higher rank, is based on a certain formalised standard.

Ranking hierarchy

In the *ranking hierarchy* structure, ranks are not directly associated with particular jobs. Employees in the same rank may perform different tasks, and employees in different ranks may perform the same task. Thus rank ordering is not congruent with task assignments. Remuneration to employees is associated primarily with their ranks as such, and not with job or individual output.

¹¹Hanada's (1993) survey on the promotion system in Japanese firms showed that a severe merit rating system had been traditionally applied for promotions, although many firms had tried to disguise such "reality" so as not to demotivate "dropouts" at early stages of their career.

Newcomers are placed at the bottom of the *ranking hierarchy* and remunerated uniformly regardless of their tasks and potential productivity at the time of entry. However, the implicit understanding at the time of recruitment is that employees are selected for higher ranks according to their observed performance within the firm over long periods, subject to the threat of costly "separation" in the case of misconduct.

The ranking hierarchy system operates distinctly in the short and long runs. In the short run, for example, one year, the employees' basic remunerations are fixed *ex ante* according to their rank, but are not directly related to their job or output performance. However, they are promoted over certain periods on a rank-by-rank basis as their performances meet certain standards. In any one period, they move up one rank, at most, but a differential arises among employees in the long run because of discriminatory treatment with respect to the speed of promotion. This results in a substantial dispersion of individual earnings and separation payments. An employee whose performance does not meet the required standard of his rank is separated in midcareer and must find a job elsewhere.

The notion of a hierarchy as a incentive system described above can obviously be distinguished from that of hierarchy in Western firms in which hierarchy functions as the vertical information-processing structure. It is based on the efficiency attained through job specialisation and rational hierarchical control. In a vertical information-processing structure, lower ranks (subordinates) communicate with others only via higher ranks (superiors), and the superiors have authority over their subordinates in the sense that they give subordinates commands.

Efficiency of the ranking hierarchy

It now remains to see how the *ranking hierarchy* copes with five incentive problems facing any large business organisation. First, how can the firm provide incentives that will discourage individual employees from "shirking"?

Alongside the labour market in which employment contracts for entry into the *ranking hierarchy* are transacted, there exist markets for fairly standardised jobs that do not require as much firm-specific skill. In these markets for standardised jobs, contracts provide less generous regular pay increases and separation payments, albeit possibly a higher starting pay. Such jobs are found more frequently in the service, retail, and wholesale industries, where the mobility of workers is higher. Also, with the growing presence of Western multinational firms in Japan, there are more alternative employment contracts for managerial and highly skilled white-collar employees that offer higher initial pay, but less job security.

Applicants hoping to enter the *ranking hierarchy* of Japanese firms that are characterised by relatively low initial pay, and relatively high *intrafirm* upward mobility, are the types who are willing to endure the long training needed to accumulate contextual skills and also to wait for the financial returns, which are realisable only after long periods. Once revealed, that information can be used both to provide appropriate rewards and to assign employees to appropriate tasks, including costly training. Thus, the coexistence of employment contracts for relatively standardised jobs (alongside those for entry to the *ranking hierarchy*) functions as a "self-selection" mechanism which copes with the adverse selection problem.

Second, how can the firm select employees who are highly productive and highly motivated from among many candidates?

As indicated above, substantial costs are imposed on employees if they cannot meet the standard required by their ranks, since they will be separated in mid-career and must find a job elsewhere. The separation does not have to be an outright firing. Although appearing to be voluntary, it may actually be imposed through pressure from the supervisor and peer group, or both. In any case, mid-career separation involves a financial penalty because of the disadvantages attaching to the separation payment. Furthermore, because of adverse selection, potential employers may judge the attributes of mid-career job changers by their personal job history. The fact that they were separated from their internal promotional mechanism in mid-career may suggest that they did not perform satisfactorily in their previous ranks. Because of this "reputation" effect, they may only get jobs of lower rank and their life-time perspectives may be lower than if they had advanced within the promotional hierarchy of their original firm. The cost of mid-career separation therefore functions as a device for enforcing discipline on employees after entry.

Third, how can the firm promote employees' integrative learning and discourage employees who have acquired contextual skills useful to the firm from quitting early?

In order that the *ranking hierarchy* conforms with labour market competition, it must be incentive-compatible in two senses. First, it must be in the interest of the firm not to terminate the contract, provided the employee performs at the required level or higher, that is, the required level of performance at each rank must yield output greater than the level of pay associated with the rank. Second, given the potential for dismissal involving costs from the loss of reputation and other benefits, it must be in the interest of employees at each level to perform at the required level, provided that the firm continues to offer the associated pay.

With the above-mentioned incentive scheme built in, the ranking hierarchy may be interpreted as a device for promoting learning and preserving costly skills within the firm. Suppose that the current productivities of the new entrants to the ranking hierarchy are indistinguishable and that all are placed at the bottom, but that they differ in potential ability, which may be realised only through learning. As learning accumulates, the potential is gradually realised, and those performing better get promoted. At the same time, those who fail to develop their potential incur costs in that they sacrifice the higher life-

time earnings associated with higher ranks or they risk being discharged in mid-career. Employees may be motivated to develop their potential, striking a balance between the cost of "shirking" learning and the benefits of promotion.

Suppose that there are situations in which it is efficient for employees to engage in two types of effort: work that increases one's own output and work that increases someone else's output. That is, shifting some effort to helping others would increase others' output more than it decrease one's own. If the *ranking hierarchy* is of the competitive type, there is no incentive for individuals to help someone else, because it is costly in terms of one's own output to contribute to the other's output, and such help thereby hurts one's own chances of promotion. If the principle of reciprocity develops under the *ranking hierarchy* so that a person's help is reciprocated by another's help, an efficient collaborative mix will occur. And, in a relatively wide range of situations, it is expected that everyone will be better off under the *ranking hierarchy* in Japanese firms may act as an incentive which promotes efficient co-operation.

Summing up, two issues have been discussed in Section 4. 4: the information-processing structure and the incentive system in Japanese firms.

The information-processing structure of Japanese firms is horizontal. Decisions involving a high degree of uncertainty, such as investment and research and development, as well as those responding to a high degree of irregularity are placed under hierarchical control. However, once an overall framework for production is laid down by such strategic decisions, "decentralised" horizontal informational exchanges and the semi-autonomous co-ordination of operations by relevant subordinates are emphasised.

In the "decentralised" horizontal structure, production decisions are co-ordinated among semi-autonomous shops (interrelated operational sub-units) that have only incomplete knowledge of technologies at the outset, but gradually become capable of responding to emerging events more quickly by better use of "on-the-spot knowledge." In order to prevent the localised use of the "on-the-spot knowledge" from causing inefficient haggling between sub-units, the sharing of knowledge among neighbouring sub-units (among workers and/or shops, etc.) is emphasised. To help foster the sharing of knowledge, workers rotate among various jobs with some frequency within, as well as beyond, workshops. Through this practice, workers are gradually made familiar with the whole work process and become capable of coping with unexpected emergencies.

The co-ordination of production between workshops to facilitate a smooth production flow along the production stream is often done by

horizontal communication without the intervention of a supervisor, as the *kanban* system at the Toyota factory exemplifies.

The ability of sub-units to co-ordinate their decisions between themselves in a way actually consistent with the organisational purpose would be limited, if they had only partial understanding of the whole mechanism operating within the firm. This understanding can be enhanced by learning by doing, but is costly in terms of time. The accumulation of such understanding as well as sharing of knowledge can be fostered only over time.

The incentives Japanese firms use to foster information-processing capabilities and sharing of knowledge of employees as well as to enhance employees' interest in identifying themselves with their firms are the *ranking hierarchy*, namely, a hierarchy as a system of ranks, with the remuneration paid to an employee depending upon the individual's rank and with internal promotion from lower to higher ranks, based on a certain formalised standard.

4. 5. The size of the corporate centre and horizontal information flow

Recapitulating the discussions in this chapter, Japanese firms are equipped with the devices for forming networks, necessary for horizontal information flow and co-ordination. As the quality of such an information-processing structure depends on the informationprocessing capabilities of employees, they are provided with the *ranking* *hierarchy* incentive to develop skills, knowledge, expertise, and a cooperative attitude towards a horizontal information-processing structure.

Likewise, from a viewpoint of organisational structure, Japanese firms seem to foster the creation of the need for horizontal co-ordination. Fewer Japanese firms have divisional forms. Those which do so are more functional and less self-contained. Even those that had divisionalised had created incomplete divisions. Such incompleteness results in linkages, which are sharing of common clients, of common production technology, etc., among divisions or departments. Linkages require co-ordination.

Thus, co-ordination in Japanese firms is horizontally made through information exchange among divisions or functional departments. It means that information exchange and co-ordination among them are decentralised, but that the decentralised approach is found to be associated with a centralised approach to administrative management, which leads to a larger size of the corporate centre in terms of its staff, a prerequisite to cope with the management need.

From this contention, it may be assumed that the size of the corporate centre in Japanese firms is likely to be associated with horizontal information.

4. 6. Implications of the Information-Processing and Information Flow Perspective for Practising Managers

In Chapters 3 and 4, information-processing was reviewed as a contingency factor influencing the size of the corporate centre. The relationships between centre's size, and vertical and horizontal information flows were addressed. What implications for practising managers could the information-processing and information flow perspective have?

As mentioned in Section 1. 3, there could be two implications: (1) for the internationalisation process of Japanese firms, and (2) for structural changes that were taking place these past years.

In anticipation of the discussions in Chapter 8 (Section 8. 4.) in which these implications will be addressed in detail, key points in the discussions are outlined below.

Internationalisation process

As discussed in this chapter, "contextual skills" fostered by years of service in a firm are an essential part of horizontal information flow in Japanese firms. Contextual skills signify a relatively wide range of integrative skills that have been developed and are useful in a nonhierarchical information-processing structure in which information flows horizontally.

This has two implications related to the internationalisation process. Japanese firms must learn how to cope with the information flow between overseas operations and the corporate centre. Because of the contextual skills requirement, Japanese firms would have to second some of their Japanese staff in key positions for communications. Section 8. 2. 2 describes how a Japanese automobile manufacturer coped with this issue, not only basing a whole department near a market outside Japan, but also creating a new unit in Japan to process information flow.

Questions raised as a consequence are as follows: Can Japanese firms internationalise their management structure of overseas operations? Can they take advantage of the human resources of the countries where they establish their operations?

Another issue related to the internationalisation process concerns the transfer of production technology, typically the Just-in-Time System. Again, the question is how important or essential contextual skills are in the transfer and implementation of Japanese production technology. As the appreciation of the yen against the dollar was steep these last years, driving manufacturing bases outside Japan, this is an important issue.

Process re-engineering and corporate down-sizing

Management scientists are increasingly aware of the importance of knowledge-creation and of the essential role played by information flow

in this knowledge-creation process if firms are to survive an intense market competition. Nonaka (1996) emphasises that firms must become "agile" and gain competitive advantages in agile delivery of goods and services through knowledge-creation. Nonaka and Takeuchi (1995) argue that the creation of new knowledge is the product of a dynamic information flow among front-line employees, middle managers, and senior managers.

On the other hand, many firms have undergone re-engineering and down-sizing in recent years. Process re-engineering does not necessarily mean down-sizing, but in the majority of re-engineered firms down-sizing took place, as a consequence of process reengineering, especially of the indirect workforce including the corporate centre (Hammer and Champy, 1993).

Process re-engineering and down-sizing, therefore, affect the size of the corporate centre and, on account of it, entail the loss of the corporate memory including contextual skills. Then, how does the dismissing of the staff who have contextual skills affect the capacity to process information flow?

A survey by the American Management Association (AMA) quoted by the Economist (1996b) reports that fewer than half of those companies that had down-sized since 1990 went on to report higher operating profits in the years following the move; even fewer saw improved productivity. Unpublished research by Monitor cited also by the Economist (1996b) found that nine out of ten firms that had outperformed their industries

over a ten-year period had "stable" structures, namely, without downsizing and, therefore, without the loss of corporate memory embodied in their employees.

The Economist (1996c) suggests that Japanese firms may have found a way to avoid the loss of the corporate memory as a consequence of down-sizing. They make use of three devices to transfer contextual skills: (1) *sempai-kohai* arrangement, *i.e.*, "older-younger" bonding, (2) quality-control circles, and (3) the suggestion box.

First, "older-younger" bonding between male employees is encouraged by Japanese firms. Though often no more than a few years older, the senior member (*sempal*) becomes responsible for guiding the younger person (*kohal*), passing along expertise for doing the job from one generation to the next. The bonding with the immediate boss—done mostly in bars after work or on the golf course at weekend—is as strong on the shop floor as in the office.

Second, Japanese companies keep their institutional memory intact through quality-control circles. The meetings held as an activity of quality-control circles are far more than a mere "talking-shop" for resolving production problems. It is part of much broader *jishu kanri* (voluntary management) approach that Japanese companies use for nurturing satisfaction and creativity among their staff.

Despite being voluntary, after-hours meetings in Japan are characterised by formal goals and strict agendas. Much use is made of statistical material, and of charts showing causes and effects. The person voted group leader "invariably" receives special training from the company on how to keep records, write reports, assign tasks and chair discussions. One thing is of a paramount importance: the findings are carefully documented so newcomers can learn what has gone before.

Third, the suggestion box is a device widely used in Japan. In the West, companies tend to make much of providing heavy financial rewards for the individual who comes up with a money-saving idea. In Japan, the focus is on capturing all the undocumented procedures going on without foreman's being aware of them.

4. 7. Summary

This chapter started by examining the view advanced by Kanter (1983) about the creation of horizontal information flow in firms. Kanter maintains that network formation is important for the exchange of information among divisions or departments, and enumerates several elements as the principal devices that aid network formation in firms: frequent mobility, including lateral moves; employment security; extensive use of formal team mechanisms; and complex ties permitting cross-cutting access.

In the light of Kanter's framework, an analysis and synthesis of the literature on Japanese management was made to demonstrate that Japanese firms are equipped with elements facilitating the creation of horizontal information flow. Horizontal information flow at lower management levels was reviewed from the standpoint of its relationship to the centralised administrative functions at the corporate centre. The horizontal information flow relates to the information-processing structure and incentive system in Japanese firms.

The co-ordination of production between workshops in order to facilitate a smooth flow along the production stream is often effected by horizontal communication without the intervention of a supervisor, as the *kanban* system at the Toyota factory exemplifies.

Japanese firms use incentives to foster information-processing capabilities and the sharing of the knowledge possessed by employees, as well as to enhance employees' interest in identifying themselves with their firms. These incentives are the *ranking hierarchy*, namely, a hierarchy as a system of ranks, with the remuneration paid to an employee depending upon the individual's rank and with internal promotion from lower to higher ranks based on a formalised standard.

It was argued that Japanese firms can "decentralise" decision-making and co-ordinating authority to lower management in operating and coordinating spheres, thanks to the centralisation of personnel administration. Although information processing and decision-making are dispersed and decentralised in Japanese firms, employees are monitored for the development of contextual skills over their entire

career in the firm, and their records are put into the organisational memory formally and informally.

The foregoing arguments concerning the centralisation of personnel administration can be extrapolated to the centralisation of other administrative functions and the shop-floor level horizontal coordination to middle management.

Finally, it was contended that, as (a) the horizontal information flow is linked to the centralisation of functions at the corporate centre, and (b) the centralisation supposes the influence in the centre's size, such size is associated with the horizontal information flow.

CHAPTER 5: RESEARCH ISSUES

5. 1. Introduction

In this chapter the concepts presented in the literature review of Chapters 2, 3, and 4 are used. They are summarised and synthesised to draw up general propositions, which are then formulated as specific hypotheses.

The chapter starts with a definition of the scope of the research. Thereafter, the relationships in Japanese firms between the size of the corporate centre and various variables relating to information flow, both vertical and horizontal, are established by way of the hypotheses for the present research. These hypotheses constitute the issues for the present research.

5. 2. Scope of the Present Research

Changes in the environment lead to changes in the strategy and organisational structure of firms (Figure 5. 1.) The changes in the environment do not directly bear upon firms. They go through different intermediate stages. As Porter (1986) points out, these changes alter the type of competition in every industry. Firms have to adapt their strategies to the new competitive pattern if they are to survive and grow. On the other hand, changes in strategies require modifications in the organisational design of the firm in such aspects as structures, systems and processes. Finally, the "goodness" of the strategy and the quality of its implementation are translated into results in terms of growth and profitability as measures of the performance of the firm in the market.

Figure 5. 1.: General framework	and	scope	of	the	research
---------------------------------	-----	-------	----	-----	----------

Level of nfluence	Environment	Industry	Strategy	Structures, Systems, and Processes	Results
	Changes in business environment ⇒	Type of competition	Strategic answer to competition	Implementa tion	Growth and profitabil- ity measure- ment
pe of riables	Structure	Variables	Behaviour	Variables	Perfor- mance Variables

Note: The scope of the present research lies in the shaded area.

In terms of the structure, behaviour and performance variables, the general framework of research contains three types of variables (Caves, 1977; Martínez, 1988). The elements configuring the environment (regulations and requirements of governments, cultural patterns, characteristics of national markets, etc.) and those shaping the type of competition in each industry (number of competitors, market growth rate, pace of technology changes, etc.) are structure variables exerting influence on individual firms, which receive these variables as given "inputs" out of control. Based on them firms design their strategies, structures, systems and processes.

The elements of strategy and internal organisation are typical behaviour variables, since they relate to the actions which firms take based upon their objectives and policies and management values. Finally, the results of the firm's performance in the market (measured in sales growth, profitability, etc.) correspond to performance variables.

The present research is centred only on the behaviour variables elements of strategy and internal organisation — because these are the variables on which the firm has direct influence.

5. 3. Research Issues

In Chapters 2, 3, and 4 the factors influencing the size of the corporate centre were examined, and vertical and horizontal information flows were identified. From that examination, some general propositions are derived below, with a view to advancing the hypotheses for the present research.

5. 3. 1. First proposition

The first proposition is that there is a relationship between the size of the corporate centre and the vertical information exchange between the centre, and divisions and functional departments. This proposition relates to the way the centre influences managers lower down and affects the decisions taken by them. Two dimensions of the centre's influence were identified: planning influence and control influence. (Goold and Campbell, 1987)

Since planning influence reflects the involvement of the centre in major decisions, strong planning influence requires that there should be intense exchange of information between the centre and the rest of the organisation in the formulation and implementation of strategies (Goold and Campbell, 1987).

Control influence concerns the way in which the centre reacts to results achieved more than the way in which it helps managers lower down to shape strategies as they emerge. (Goold and Campbell, 1987). Therefore, the centre is less involved and the information exchange between the centre and the remaining part of its organisation is less intense.

Moreover, it has been argued that the size of the corporate centre is positively associated with information-processing capabilities (Haleblian and Finkelstein, 1993; James, 1951; Hill, 1982; Harrison, 1975; Cummings *et al*, 1974). Thus, the more the exchange of information, the larger the size of the corporate centre.

Accordingly, the higher the planning influence, the greater the vertical information flow, and therefore, the larger the size of the corporate

centre to cope with this need for information-processing. This relationship is set out below as Hypothesis 2.

On the other hand, the higher the control influence, the less the vertical information flow, and therefore, the less need for a large corporate centre to cope with information-processing needs. This possible inverse relationship between control influence and the size of the corporate centre is advanced below as Hypothesis 3.

In their interview survey of fourteen Japanese firms Campbell *et al* (1990) observed that in Japanese firms there is a continuous and large amount of flow of information between the divisions and the centre.

Based on this observation by Campbell *et al* and the relationship between information flow and the strategic management styles described in Chapter 3, it can be hypothesised that Japanese firms are high on planning influence (on account of the large amount of vertical information flow), and low on control influence (which seems more probable, since high control influence may reduce the flow of vertical information.) This, as a key argument in the thesis, is given the position of Hypothesis 1.

The following hypotheses are therefore advanced.

HYPOTHESIS 1: The influence the Japanese corporate centre exerts is high on planning and low on control.

HYPOTHESIS 2: The size of the corporate centre measured in terms of the number of regular full-time personnel is positively associated with the level of planning influence the corporate centre exerts.

HYPOTHESIS 3: The size of the corporate centre measured in terms of the number of regular full-time personnel is inversely associated with the level of controlling influence the corporate centre exerts.

5. 3. 2. Second proposition

The second proposition of the present research is that there is a relationship between the size of the corporate centre and horizontal information exchange in Japanese firms. It relates to the way Japanese firms are organised.

Few Japanese firms have divisional structures. Those which do so are more functional and less self-contained. Even in divisionalised firms, the divisions are "incomplete" more often than not (Japan Development Bank, 1989; Campbell *et al*, 1990) and linkages among divisions or functional departments exist (Kagono 1992). Linkages can result from sharing common clients, common production technology, etc. (Goold and Campbell, 1987).

The existence of linkages requires co-ordination. In Japanese firms coordination is horizontally made through information exchange among divisions or functional departments (Aoki, 1988; Nonaka, 1990; Kagono, 1992). Thus, information exchange is decentralised, but the decentralised approach to information exchange of Japanese firms is found to be associated with a centralised approach to administrative management (Kono, 1984; Aoki, 1988).

From this proposition it may be hypothesised that:

HYPOTHESIS 4: The size of the corporate centre in Japanese firms measured in terms of the number of regular full-time personnel is positively associated with the level of linkages among divisions or functional departments.

5. 4. Summary

This chapter defined the research issues of the thesis. It started by defining the scope in which the research is developed.

Thereafter, two propositions were formulated concerning the relationships, in the management of Japanese firms, between the size of the corporate centre and various variables relating to information flow, both vertical and horizontal (See Figure 5. 2.).

Based on these propositions four hypotheses were developed for testing of which the present research is carried out.







CHAPTER 6 : RESEARCH DESIGN

6. 1. Introduction

In the preceding chapter, the conceptual scheme and the hypotheses of the research were formulated. This chapter presents the design of the research, namely, the specification of the methods and procedures chosen to collect the evidence for the validation of these hypotheses. In the first place, an analysis is made of the type of the research which has been conducted. Second, explanation is given of the procedures for the selection of the sampling frame representing a wide range of production technology, followed by analysis of the sample. Third, the instruments used to select the data are described. Fourth, the sample or respondent firms are described. Fifth, it is explained how the variables are "operationalised" and what scales are used for the measurement. Sixth and last, the method data analysis is described to show how the results of the present study were obtained. Possible limitations of the methodology used are also mentioned.

6. 2. Type of Research

There are several criteria for the classification of empirical research. In this section, mention will only be made of two of the commonest: 1) the source from which data are gathered, and 2) the main purpose of the research. On the basis of the source from which data are gathered, researches can be classified into five types: 1) those using secondary data, 2) those gathering primary or first-hand data by means of communication with respondents, 3) natural experiments, 4) controlled experiments, and 5) simulation. The present research corresponds to the second type of research, namely, to research based on a questionnaire survey which contains primary data gathered specifically for this purpose. The following sections will describe the survey including the population, the sampling frame and the sample.

According to a second classification of the research purpose, there are three types of research: exploratory, descriptive, and causal. Although the present research has certain elements of a descriptive study, its main purpose is to demonstrate relationships of "causal type" between the size of the corporate centre and the role it played, as described in the preceding chapter. In Chapter 5, it was hypothesised that Japanese firms have relatively large corporate centres, independent of production technology, industry environment, and organisational structure (whether functional or "incomplete" divisional structure), and that this relatively large corporate centre is necessary for good sharing of information in the organisation. It is, then, obvious that this research aspires not only to demonstrate the existence of a relationship between the size and the role played by the corporate centre, but also to explain the "probabilistic cause" of this relationship, which is one type of causal study¹. A probabilistic cause is necessary but not sufficient for the subsequent occurrence of an event².

The research is of the "causal type", but it does not explain complete causal relationships, since, on account of the nature of its design, it is not possible to demonstrate the total causal relationship. Green and Tull (1975: 74) argue that there are three conditions or three types of evidence for drawing inferences about complete causal relationships: (1) associative variation, which is a measure of the extent to which occurrences of two variables are associated; (2) sequence of events, namely, the requirement that the causal factor occurs first; and (3) absence of other possible causal factors. The questionnaire survey method adopted in this research allows us to meet only the first of the three conditions, namely, the existence of certain relationship, in addition to a coherent argument which may explain and support the existence of such a relationship. It is impossible to comply with the remaining two conditions through a questionnaire survey. The remaining two conditions could only be complied with through an experimental design with a control group, in other words, a controlled

 $\mathbf{Y} = \mathbf{F}(\mathbf{X}_{1}, \mathbf{X}_{2})$

where X_1 and X_2 are independent. Assuming that it is possible to specify F, the variables X_1 and X_2 become jointly necessary and sufficient to determine Y. The effect on Y of a known change in X_1 may be reinforced, counteracted, or left the same, depending upon what happens to X_2 . X_1 is not sufficient to determine Y.

¹ Another type of causal study is about deterministic causation. A deterministic cause is an event that is necessary and sufficient for the subsequent occurrence of another event.

 $^{^2}$ A probabilistic causation can be typically seen in a multivariate relationship of the form:

experiment, which, obviously, does not fit in with the characteristics of this research.

This research limits itself to a focus on the "functionalist" relationships between information flows and the size of the corporate centre. It does not exclude, nevertheless, the possibility of alternative influences on the size of the corporate centre, such as political process and dysfunctional effect of bureaucracy as well as other institutional influences on it. Further research based on the "non-functionalist" approach is suggested in Chapter 8, while the empirical findings of this research are substantiated with some case study material.

In summary, the chosen research design may not demonstrate the complete causal relationship, but the research will provide evidence of causal explanation of the "probabilistic" type, which means that the explanation will give an account of a cause that is necessary but not sufficient for the subsequent occurrence of the event.

6. 3. Population and Sampling Frame

6. 3. 1. Population

The unit of analysis, therefore the population of the present research, is Japanese firms. Such a population is, however, so vast that selection is needed on the basis of three criteria: 1. the firms in the population should represent industries in different environments and with different production technologies.

2. the firms selected should be relatively homogeneous. This includes that they have a certain size in either their turnover or number of employees.

3. the population should be sufficiently large to extract from it a sample of reasonable size.

Using all Japanese firms as the population would mean including firms of very different sizes and from the primary, the secondary, and the tertiary sectors. This would comply with the first criterion but would be against the second because of their heterogeneity. For example, the production process and activity is very different in an agribusiness, in an automobile, and in an insurance firm.

Accordingly, it was decided to consider as the population the firms listed on the First and Second Sections of the Tokyo, Osaka and Nagoya Stock Exchanges, operating in manufacturing industries in Japan.

To build up the population of the present research "JAPAN COMPANY HANDBOOK — FIRST SECTION Summer 1993", "JAPAN COMPANY HANDBOOK — SECOND SECTION Summer 1993", both of which are published by Toyo Keizai, and "Nikkei Kaisha Joho — 93-IV" published
by Nippon Keizai Shimbun, were used. The first two provide the latest financial information on all Japanese firms listed in the First and Second Sections of the Tokyo, Osaka and Nagoya Stock Exchanges and cover a total of 2,060 firms as of March 30, 1993. The third provides the financial information on 2,138 firms listed in the First and Second Section as of September 1st, 1993.

Based upon these publications and SICC codes applied to each company, it is possible to identify the companies in the manufacturing industry. Based on this identification it is found that the number of firms complying with the sampling criteria — namely, the population of this research — is around 1300.

6. 3. 2. Sampling Frame

Industries are classified according to the securities code number which is a four-digit number assigned by the Securities Identification Code Conference (SICC). The industries chosen for the present research have code numbers from 2001 to 7999, which correspond to manufacturing industries.

Non-probability sampling procedures were used for the sampling procedures. The procedures included two principal varieties: convenience sampling and judgmental sampling. Two criteria were used to select the sample from the sampling frame. (a) A criterion was used to attempt to make the sample representative of the population of the firms in the selected industries, and of the reality of the business community in Japan. These conditions would be fulfilled if it was ensured that principal firms in each industry were included in the sampling frame. This criterion could correspond to sampling on a judgmental basis. Firms with over 51 per cent foreign ownership were excluded, with the aim of limiting the samples to firms of Japanese ownership and, therefore, of "Japanese management".

To meet the first criterion, the full-time employee number of over 1,500 was used as the criterion for selection.

(b) The second criterion is the accessibility to the firms. Contacts of the researcher or the Manchester Business School were used. Approximately fifteen firms in the sampling frame were approached in this way. The rest were contacted by means of a letter addressed to the office of the president or managing director. This criterion corresponds to convenience sampling.

Table 6.1. presents the final sample of firms classified according to industries. In total, 375 firms were included, and they were divided to eight industry categories, with a maximum of 111 firms in the Chemicals and Petroleum Products industry category, and a minimum of ten in Shipbuilding and Rolling-stock.

375 firms listed in the First Sections of the Tokyo, Osaka and Nagoya Stock Exchanges, operating in manufacturing fields in Japan, belonging to eight industries of Table 6.1., constituted what could be called the SAMPLING FRAME (Bailey, 1982). From this sampling frame the 55 firms which answered the questionnaire were used for the analysis of the research.

Table 6. 1.: Sampling frame

SICC code	Industry	N°	of	Firms	Per	cent	(%)
2001-2899	Foods & Beverages		28	3		7	
3001-5393	Chemicals & Petroleum Products		11	1		30	
5401-5992	Metal Products		43	3		11	
6005-6652	Machinery & Parts		48	3		13	
6701-6999	Communications & Electronics		48	3		13	
7003-7142	Shipbuilding & Rollingstock		1()		3	
7201-7309	Motor Vehicles		36	5		10	
7407-7999	Miscellaneous Manufacturing		51	1		14	
	TOTAL		37	5		100%	

Source: Japan Company Handbook — First Section 1993 Summer (1993)

Table 6. 2. summarises some economic and financial information in terms of total assets, turnover and full-time employees in 1993 on the firms included in the sampling frame. Total assets include all assets possessed by the firm and represent the total of current assets, fixed assets, and deferred assets.

183

Table 6. 2.: Information on the firms in the sampling frame

	Maximum	Average	Std Dev	Minimum
Total Assets (a)	9,582,707	554,540	1,119,449	30,453
Turnover (a)	10,163,376	503,113	1,021,684	19,667
Employees (b)	83,821	6,606	10,320	1,505

Notes

(a) in ¥ millions.

(b) in number of people.

Source: Japan Company Handbook — First Section 1993 Summer (1993)

6. 4. Data Gathering Procedure

It was explained in the preceding section that one of the criteria for the selection of the sample was accessibility to the firms. A list was prepared for inclusion of the contacts that the author of the dissertation has through his past business experience. Then, other contacts available from the Manchester Business School were added. The accessibility criterion was balanced by other criteria for the selection of the sample, as described in the preceding section.

The contacts on the list were approached in personal interviews, by telephone, or by letter to explain the purpose of the study, as well as to request their help. In exchange, a copy of the survey result both for the entire sample and for their industry category was offered. About 15 firms were contacted in this way. All telephone calls were made by the author of this dissertation. The majority of the executives interviewed or contacted by phone immediately accepted the request, though a few of them asked for an official letter to file the request. The firms in the sampling frame were contacted by a letter dated 23rd November, 1993 addressed to the Public Relations department. A copy is attached as Annexes 1a (English Original) and 1b (Japanese translation). The addresses were found in "JAPAN COMPANY HANDBOOK — FIRST SECTION Summer 1993" published by Toyo Keizai. One and a half month after the despatch of the first letter, a reminder was sent to confirm receipt of the letter and/or to remind the firms of the request.

Out of the total 375 firms contacted, seventeen apologised for not being able to collaborate with the research, 283 never responded.

Fifty-eight firms sent back the questionnaire before the despatch of the reminder. Fifteen responses out of these fifty-eight did not contain the data on the size of the corporate centre.

After the reminder fourteen firms sent in their responses, but five of these lacked the data on the size of the corporate centre.

Twenty-four telephone calls were personally made by the researcher of the present survey to the people who had filled in the questionnaire in order to get the data on the size of the corporate centre. As a consequence, three firms sent back a fax with the requested data.

As shown in Table 6. 3., fifty-five firms form part of the sample (to be referred to hereafter as Respondent Firms).

Table 6. 3.: Respondent firms

Sampling frame	375
Before the reminder	
Respondent firms	58
Lacking the size data	(15)
Usable responses	43
After the reminder	
Respondent firms	14
Lacking the size data	(5)
Firms responded with	3
the data on the size	1.35
Usable responses	12
Total usable responses	55

The motives given by the firms for declining to participate in the research were diverse. They were as follows:

- "The data requested are confidential information and our firm's policy is not to provide it"

- "Information on the corporate structure is regarded as confidential at our firm"

- "Some of the requested information is connected with confidential matters at our firm"

- "After discussing the issue among us, we decided that our firm's organisation does not merit your study"

- "We are sorry for not being able to respond your request, but we have to decline it"

186

- "Basically, our firm does not publish the information related to the overall corporate structure"

The collaboration of each firm in the research consisted in the filling-in of a questionnaire. The questionnaire is described in the following section and a copy is included as Annex 2a (English Original) and Annex 2b (Japanese translation).

As the questionnaire contained questions on strategic aspects of the firm, it was very important that the respondents should be in a position to have sufficient data or information, or to be able to obtain it from relevant departments or divisions. Table 6. 4. presents a summary of their posts and departments of the respondents to the questionnaire. It is observed that all respondents are either from departments which hold overall information about their firms (General Affairs, Public Relations and Personnel Department) or from departments directly related to strategic issues (President's Office and Corporate Planning), which guarantees that their responses reflect the reality of their firms. Exceptions are four respondents who did not mention their positions in their firms. Table 6. 4.: Departments and posts of respondents (All responses)

Department	Pas	ation	Frequency	Percent
Corporate Planning	0.075.2	and the second second		
	General	Manager	4	A March
	Section	Manager	3	2611
	Deputy	Manager		2180
	Staff		2	Set State
	Sub	ototal	9	16%
General Affairs				12455
	General	Manager	1	
	Section	Manager	3	1000
	Deputy	Manager	1	
	Staff		• 3	
	Sut	ototal	8	15%
Personnel				
	General	Manager		
	Section	Manager	1	
	Deputy Staff	Manager	Same S.	
	Sub	total	1	2%
President's Office				
	General	Manager		
	Section	Manager	2	19 M. A.S.
	Deputy Staff	Manager		
	Sub	total	2	4%
Public Relations				
	General	Manager	5	
	Section	Manager	17	20 8 2
	Deputy	Manager	3	
	Staff		6	
	Sub	total	31	56%
Unknown			4	
	Sub	total	4	7%

When the questionnaires were received — which covered the period between mid-October, 1993 and end-January, 1994 — a short thankyou letter was sent to each respondent, to inform them that the datagathering phase had ended successfully and to remind them that a summary of the survey results would be sent to them in June or July.

6. 5. Respondent Firms (Sample)

In this section four explanations will be given of the fifty-five firms whose responses were used for the empirical analysis, the results of which will be described in depth in Chapter 7. First, a list of these firms is provided within the constraint of a strictly confidential handling of their responses.

Second, these respondent firms were classified by themselves according to their own assessment of the industry categories mentioned in the questionnaire, which follow those of JAPAN COMPANY HANDBOOK. All manufacturing industry categories of JAPAN COMPANY HANDBOOK are represented in the responses.

Third, the respondent firms are classified according to their production technology in terms of the rountineness/non routineness criterion. As Woodward (1965) defined it, the most routine technology corresponds to continuous process, whereas the most non-routine, to custom technology.

Production technology is an important contingency factor, since it is essential in determining organisational structure and in setting the tone of human relationships inside the firms (Woodward, 1958).Likewise, it has become a significant vehicle for changes in employment and organisation (Child 1984).

189

The even representation and wide spread of production technology in the sampling frame shows that the respondent cases are unbiased in terms of the technology used. However, due to the small number of sample firms in each industry category, a valid comparison on the basis of industry category and production technology was not possible in our analysis.

Fourth, information is provided on the comparison of industry category and production technology.

The respondent firms

As all the responses and their data are to be treated strictly confidential, it is not possible to give any information or clue which might lead to the identification of each reply with a specific firm. Only the names of the respondent firms are disclosed below:

Table 6. 5.: Respondent firms

FIRM	TOTAL ASSETS	TURNOVER	EMPLOYEES
AIWA	113,314	161,650	1,703
Ajinomoto	582,657	595,000	6,074
AMADA	320,294	146,756	1,847
Asahi Glass	1,515,626	1,011,815	9,808
ASICS	118,596	115,424	1,998
BRIDGESTONE	1,846,290	725,441	16,391
Cleanup	69,457	88,259	2,679
DAIHEN	101,755	74,000	1,839
DAIKIN INDUSTRIES	438,963	342,000	8,579
DAINIPPON SCREEN MFG	240,607	119,625	3,232
EZAKI GLICO	183,562	93,500	1,818
Fuji Heavy Industries	825,130	873,042	14,813
Fujikura	439,452	251,700	4,460
HONDA MOTOR	3,156,181	2,694,836	31,089
Ishikawajima Harima Heavy Inds	1,421,839	803,389	16,273
Japan Aircraft Mfg	53,264	33,218	1,710
KAKEN PHARMACEUTICAL CO	73,165	68,495	1,857
Kawasaki Heavy Industries	1,162,201	953,000	17,495
KOKUYO	265,942	303,737	3,335
KOMATSU	1,452,818	525,704	9,238
Komatsu Forklift	101,709	70,364	1,524
KURABO INDUSTRIES	173,179	130,543	3,403
Kureha Chemical Industry	142,208	93,376	2,067
KYOCERA	666,897	300,550	13,914
KYOWA HAKKO KOGYO	388,239	296,000	5,500
Meiji Milk Products	226,483	424,297	6,053
Mitsubishi Steel Mfg	99,309	78,760	2,211
MIURA	52,130	38,921	. 1,572
Morinaga & Co	192,166	155,636	2,776
NIKON	366,618	204,897	7,351
NIPPON PAINT	277,525	161,163	2,905
Nippon Steel Chemical	389,214	281,509	2,937
Nippon Yakin Kogyo	147,899	87,000	1,778
Nissin Electric	109,460	107,728	3,609
NTN	471,904	272,000	7,738
OMRON	609,855	3,640,000	7,900
SANKYO ALUMINIUM INDUSTRY	356,744	298,024	6,435
Sasebo Heavy Industries	72,409	56,400	2,100
SEGA ENTERPRISES	241,645	346,900	3,500
Sharp	214,690	1,152,800	28,300
Snionogi Sn Diladia Criterica I	367,619	238,839	7,037
SS PHARMACEUTICAL	2 672 470	35,000	2,117
Sumitomo Metal Mausines	2,073,470	1,100,007	22,009
	070,319	774,223	4,195
TAISHO PHARMACEUTICAL	519,550	204,035	4,551
	11,414	03,320	2,614
TANADE SETTARU	203,100	211,030	4,909
	027,341	367,340	8,739
TOA STEEL CO., LID.	210,830	100,499	1,834
	5 724 420	2 1 50 571	14,404
Tomma Chemical	3,124,439 07 170	3,130,371	74,883
Toyama Chemical	216 012	40,000	2,311
	171 649	122 500	J,400 4 EAA
I AWAIARE-NUNEIWELL	171,343	122,300	4,500

191

Note: In millions of Japanese yen.

Classification by total assets, turnover and full-time employees

Table 6. 6. summarises the information on the respondent firms in terms of total assets, turnover and full-time employees in 1993.

Table 6. 6.: Information on the respondent firms

	Maximum	Average	Std Dev	Minimum
Total Assets (a)	5,724,439	592,787	943,235	52,130
Tumover (a)	3,640,000	465,147	722,157	33,218
Employees (b)	74,883	7,893	11,314	1,524

Notes (a) In ¥ millions

(b) Number of people

Source: Japan Company Handbook — First Section 1993 Summer (1993) and the data received from the respondent firms.

Industry classification

Respondent firms are distributed among eight industries, which cover all the manufacturing industry categories defined in JAPAN COMPANY HANDBOOK. The number of firms by industry ranges from fifteen for miscellaneous manufacturing to two for shipbuilding and rolling stock industry (See Table 6. 7).

Industry category N	umber of Firm	s Per cent (%
Foods & Beverages	5	9
Chemical & Petroleum Products	7	13
Metal Products	7	13
Machinery & Parts	8	15
Communications & Electronics	6	11
Shipbuilding & Rollingstock	2	4
Motor Vehicles	5	9
Miscellaneous Manufacturing	15	27
TOTA		

Table 6. 7.: Respondent firms by industry

Source: Japan Company Handbook — First Section 1993 Summer (1993) and the information received from the respondent firms.

Classification by production technology

Production technology will be analysed in Chapter 7 as one of three environmental factors, along with the stability of business environment and diversification. As shown in Table 7. 4 in the next chapter, the production technology of the respondent firms ranges from 1 (=extremely high routineness) to 7 (=extremely low routineness). Scores 4, 6 and 2 and 3 have large values with 30.9, 20, and 16.4 and 16.4 per cent, respectively. Figure 6.1 illustrates the data from the responses. It is observed from the figure that the production technology in terms of rountineness and non-routineness is normally distributed, except for score 5.



% of responses

Comparison of industry classification and production technology

In terms of industry classification, it was found that Miscellaneous industry has the most non-routine production technology with 5.55, and Communications and Electronics, the highest routine production technology with 2.83. The details of the routineness of production technology is as shown in Table 6.8.

Figure 6. 1.: Production technology

Industry	Average	Standard	Number of					
category	routineness	deviation	firms					
Foods &	3.00	1.41	5					
beverages								
Chemical &	2.86	1.68	7					
petroleum								
products								
Metal products	3.29	1.25	7					
Machinery &	4.14	. 1.77	8					
parts								
Communications	2.83	1.60	. 6					
& electronics								
Shipbuilding &	4.00	1.22	2					
rolling stock			·					
Motor vehicles	5.25	1.50	5					
Miscellaneous	4.57	1.74	15					
manufacturing								
Total	3.76	1.71	55					

 Table 6. 8.: Routineness of production technology

6. 6. Representativeness of the respondent firms

Since the number of the respondent firms are 14 per cent of the sampling frame, a test of the representativeness of these firms was carried out.

Table 6. 9. presents the results of the two-sample t-test to evaluate the representativeness of the respondent firms within the sampling frame.

Table 6. 9.: t-test for the sampling frame and the respondent firms

(In ¥ millions: 1993)

	Number of Cases	Mean	Std Dev	td Error
Sampling Frame	375	554,540	1,119,449	57,808
Respondent Firms	55	592,787	943,235	127,185
			a distant of the	
	F Value	1.41		
	2-tail Probability	0.124	1946 1 - 64 - 3	
Pooled	t value	-0.24		
Variance	Degrees of Freedom	428	Sector 1	
Estimate	2-tail Probability	0.81		
Separate	t Value	-0.27		
Variatice	Degrees of Freedom	78.13	Let State	
Estimate	Z-tail Probability	0.785		
	Number of Cases	Mean	Std Dev	td Error
Sampling Frame	375	503,113	1,021,684	52,759
Respondent Firms	55	465,147	722,157	97,375
	FValue	2		
	2-tail Probability	0.002		
Popled	t value	0.27		
V:BFIBFICE	Degrees of Freedom	428		
Conordo	2-tall Probability	0.79		
OEDECARE	L value	0.34		
vaciacise Setimate	2-tail Brobability	0 7 2 2		
Catiniare	Z-Lan Probability	0.733		
	Number of Cases	Mean	Std Dev	td Error
Sampling Frame	375	6,606	10,320	532
Respondent Firms	55	7,893	11,314	1,525
	E Value	12		
	2-tail Probability	0 336	- '	
Pashad	t value	1.85		
Variance	Degrees of Freedom	428		
Estimate	2-tail Probability	0.394		
Separate	t Value	1.8		
Verience	Degrees of Freedom	67.84		
	2 sail Drahabiliar	0.420		
	Sampling Frame Respondent Firms Pooled Variance Estimate Separate Variance Estimate Sampling Frame Respondent Firms Pooled Variance Estimate Separate Variance Estimate Separate Variance Estimate Separate Variance Estimate Separate Variance Estimate	Number of CasesSampling Frame375Respondent Firms55Pooledt valueVarianceDegrees of FreedomEstimate2-tail ProbabilitySeparatet ValueVarianceDegrees of FreedomEstimate2-tail ProbabilitySeparatet ValueVarianceDegrees of FreedomEstimate2-tail ProbabilitySeparatet ValueVarianceDegrees of FreedomEstimate2-tail ProbabilityPooledt valueVarianceDegrees of FreedomEstimate2-tail ProbabilityPooledt valueVarianceDegrees of FreedomEstimate2-tail ProbabilitySeparatet ValueVarianceDegrees of FreedomEstimate2-tail ProbabilitySeparatet ValueVarianceDegrees of FreedomEstimate375Sampling Frame375Sampling Frame375Sampling Frame375Separatet ValueVariance55F Value375SeparateFirmsSampling Frame375Sampling Frame375Separate1 ProbabilitySeparate2-tail ProbabilitySeparate2-tail ProbabilitySeparate2-tail ProbabilitySeparate2-tail ProbabilitySeparate2-tail ProbabilitySeparate2-tail ProbabilitySep	Number: of Cases MeanSampling Frame375 554,540Respondent Firms55 592,787F Value1.412-tail Probability0.124Pooledt value-0.24VarianceDegrees of Freedom428Estimate2-tail Probability0.81Separatet Value-0.27VarianceDegrees of Freedom78.13Estimate2-tail Probability0.785Sampling Frame375 503,113Bespondent Firms55 465,147F Value22-tail Probability0.002Pooledt value0.27VarianceDegrees of Freedom428Estimate2-tail Probability0.002Pooledt value0.27VarianceDegrees of Freedom428Estimate2-tail Probability0.79Separatet Value0.34VarianceDegrees of Freedom89.25Estimate2-tail Probability0.733Sampling Frame3756,606Respondent Firms557,893F Value1.22-tail Probability0.336Pooledt value1.85VarianceDegrees of Freedom428Estimate2-tail Probability0.336Pooledt value1.85VarianceDegrees of Freedom428Estimate2-tail Probability0.336Pooledt value1.85VarianceDegrees of Freedom428	Number of Cases Mean Std Dev Sampling Frame 375 554,540 1,119,449 Breapondent Firms 55 592,787 943,235 F Value 1.41 2-tail Probability 0.124 Probability 0.124 24 Variance Degrees of Freedom 428 Estimate 2-tail Probability 0.81 Separate t Value -0.27 Variance Degrees of Freedom 78.13 Separate t Value -0.785 Number: of Cases Mean Std Dev Sampling Frame 375 503,113 1,021,684 Beapondent Films 55 465,147 722,157 Image: Std Dev Sampling Frame 375 503,113 1,021,684 Beapondent Films 55 465,147 722,157 Image: Std Dev Sampling Frame 2-tail Probability 0.002 Variance Degrees of Freedom 89.25 2-tail Probability 0.73 Separate t Value 0.34 2-tail Probability 0.73 Sampling Frame 375

۰.

As t = -0.24, df = 428, and p = 0.810 for Total Assets, there is no significant difference between the means of the Sampling Frame and the Respondent Firms. Similarly, for Turnover and Employees, t = 0.27, df = 428, and p = 0.790; and t = -0.85, df = 428, and p = 0.394, respectively. There is no significant difference between the Sampling Firms and the Respondent Firms in terms of turnover and employees, either. Thus it may be concluded that both the Sampling Firms and the Respondent Firms come from the same population.

6. 7. Instrument for Data Gathering

The data were gathered by means of a structured questionnaire based mainly upon closed-ended questions. Some of the data were gathered by open-ended questions, which could be easily and briefly answered. The questionnaire was composed of five parts, which are briefly described below. (See Questionnaire in Annex 2a):

1) General Information: the purpose of this section was to obtain general information on:

- activity of the firm
- name of the firm
- respondent's name/position
- turnover -
- number of employees
- number of corporate staff

197

expenditure on corporate functions

2) Environmental Factors: information on the environment of the respondent firm and on its production technologies was gathered in this section.

3) Organisational Structure and Systems: information was collected on the divisional form or functional form, the number of divisions, and the functions performed by divisions.

4) Corporate Centre's role: planning influence and control influence exerted by the corporate centre were measured in this section.

5) Linkages: information on the sharing of customers, reciprocal purchase of products, sharing of any production technology, sharing of marketing skill and product development was gathered in this section.

6) Scope and Manning of the Corporate Centre: information on the number of the corporate personnel and of central functions was gathered in this section.

As can be observed, the questionnaire goes from general questions to more specific ones, which are the real objective of the research. This method of "narrowing down" helps respondents to view specific questions from a wide angle, which enables them to understand better the "meaning" of these questions.

The majority of the authors on the design of research hold that a good questionnaire is one that has been submitted to in-depth scrutiny, and that there is no substitute for this stage (for example, de Vaus, 1990). With this view in mind, the questionnaire of the present research was reviewed by the doctoral supervisor of the author and by a doctoral student at Manchester Business School. It was then translated into Japanese by the author of this dissertation. Once translated into Japanese the questionnaire was reviewed by a Japanese Certified Public Accountant (CPA) in comparison with the original in English. The questionnaire in Japanese was then test-answered by three Japanese second-year MBA students at Manchester Business School, all of whom are trainees from Japanese firms. At the same time, it was test-answered by three expatriate managers at three Japanese multinational enterprises, one in UK (a leading electric and electronic goods manufacturer) and two in Spain (a motor vehicles manufacturer and a pharmaceutical firm). This test stage conducted in October, 1993 turned out to be useful, since, although the questionnaire was in general well understood, the test allowed the introduction of some interesting modifications (e.g., an explanation on the corporate centre).

6. 8. Scales for the Measurement of Variables

The research issues described in Chapter 5 were "operationalised" in twenty-two variables measured by means of twenty-two interval-scaled questions. Four questions were measured by nominal scales, ordinal scales and ratio scales. Given the nature of the elements of strategic management which are not amenable to quantification, it was necessary to "operationalise" these variables by means of perception measurements. All the questions involving perceptions and opinions of the respondents were measured by ordinal scales, which are, however, considered by several authors to be a good approximation to interval scaling (Lehman, 1979; Aaker and Day, 1983). In a research like this, it is desirable that the scales used should be interval scales or at least a good approximation to them, since, with these scales, it makes sense to talk about average and standard deviation, and multivariate statistic techniques can be applied.

The scales used, which are no more than a variation of the scales of the semantic differential (SD) method developed by Osgood, Suci, and Tannenbaum (1957), are known as bipolar scales, because they assign a set of pairs of antonyms, the extreme of each pair being separated by odd-number intervals, perceived by respondents to be equidistant intervals. An example of the scale is as follows:

To what extent does the corporate centre provide a strategy and direction as guidance for the divisions? Please circle the number which best represents the situation at your firm.

a very			midpoint			little or no
great deal					i	strategy
of strategy						and
and						direction
direction		-				
1	2	3	4	5	6	7

It can reasonably be supposed that the respondent considered the alternative 2 to be as distant from 1 as the alternative 6 from 7.

(This example refers to planning influence. It relates the way the corporate centre influences managers lower down and affects the decisions taken by them, and concerns the centre's effort to shape strategies as they emerge and before decisions are taken. In this research, it is measured by the average of aggregated values in interval scale questions answered by the sample firms.)

Twenty questions about the corporate centre's role³, business environment and linkages are based on this scale to measure

How far does the headquarters provide a strategy and direction as guidance for the divisions? Tick one of the following:

1

(b) Some themes, objectives and policies from headquarters, but strategy mainly bottom-up.

³ For Part 4 Corporate Centre's Role of the questionnaire, Ashridge Strategic Management's questionnaire entitled "Strategic Management Style: Understanding your company's style" was taken into account.

Ashridge's questionnaire is based on fixed-alternative questions. An example of these questions reads:

⁽a) Clear headquarters strategy articulated. Division expected to follow the centre's lead.

responses. The "operationalisation" of these questions can be seen in the questionnaire used for gathering the data of this research (Annex 2a and 2b). Table 6. 10 summarises how the key variables of this research are operationalised.

Variables	Operationalised definition	Question numbering in Annex 2a
Planning influence	······································	· · ·
	•	
	Involvement of the corporate centre in	4 (1)
	the decisions and strategies of lower	
	management	
	Provision of strategy and direction	4 (2)
	Importance of formal strategic planning system	4 (3)
	Centre's investment initiatives	4 (4)
	Strategy criteria in the approval of capital requests	4 (5)
	Importance of non-financial objectives set for lower management	4 (7)
Control influence		
	Importance of financial return for large capital requests	4 (6)
	Financial targets versus strategic objectives in performance review	4 (8)
	Agreement on financial targets	4 (9)
	Financial targets as a promise	4 (10)
	Monitoring of financial performance	4 (11)
	Division's performance and manager's career	4 (12)
Linkages		
_	Common customers	5 (1)
	Purchase of products among divisions	5 (2)
	Common production technology	5 (3)
	Shared marketing skills	5 (4)
	Shared product development	5 (5)
The centre's size		
•	Number of staff	6

Table 6.	10.:	Measurement	and	operationalisation	of	the	key
variables	3			-			

Like all other interval-scaled questions in the questionnaire, there are two extremes of the scale, which assumes two points:

(c) Headquarters detached; all initiatives, except for major acquisitions, come from below.

1. That the respondent correctly interpreted the term "strategy" and "direction"; and

2. That the adjectives "a very great deal of strategy and direction" and "no strategy and direction at all" bear the same meaning to each respondent.

Therefore, it is possible to compare the involvement level of the corporate centre in strategy and direction setting among different firms in the respondent sample.

Some of the remaining questions on the corporate centre's role used interval-scales (which are, moreover, interval-ratio scales), insofar as they refer to objective quantifiable data and facts. An example of this scale is as follows:

Are major investment ideas that affect the division initiated at corporate centre level? Please circle the number which best represents the situation at your firm.

100% of cases	over 75% but	over 50% but	over 25% but	over 0% but
	less than	less than 75%	less than 50%	less than 25%
	100% of cases	of cases	of cases	of cases
1	2	3	4	5

Regarding the number of response categories or response alternatives, Cox (1980) argues that the optimum number of categories is between 4 and 9, and that an odd number is preferable, since it enables respondents to adopt a neutral position. Accordingly, the measurement scales of the questionnaire of the present research followed this argument, and gave seven categories.

6. 9. Data Analysis

The data gathered by means of the questionnaire were statistically analysed using various multivariate techniques, especially multiple regression analysis, at two stages. The reason why multiple regression was mainly used and the explanation of the two stages for the data analysis are set out below. Further details of the data analysis will be described in the next chapter.

First, the reason why multiple regression was mainly used in the data analysis. Methods for the analysis of associative data may be classified into two sets: dependence structure and interdependence structures.

If the data matrix is partitioned into criterion (dependent) and predictor (independent) variables, the problem becomes one of analysing *dependence structures*. This, in turn, can be broken down into two subcategories — 1) single criterion/multiple predictor association and 2) multiple criterion/multiple predictor association. Multiple regression, analysis of variance and covariance (ANOVA and ANCOVA), and twogroup discriminant analysis are used for single criterion/multiple predictor analysis, whereas multiple discriminant analysis, canonical correlation and multivariate analysis of variance and covariance (MANOVA and MANCOVA) are used for multiple criterion/multiple predictor variables.

If the data matrix is not partitioned into criterion and predictor subsets, the problem is one of analysing *interdependence* structures. Depending on whether the emphasis is on variables or objects, factor analysis and cluster analysis are used for interdependence structures.

The present research deals with the analysis of dependence structures in which the single criterion or dependent variable is the size of the corporate centre and predictor variables are multiple. Multiple regression was used as one of the main data analysis tools, since other analytical instruments including ANOVA, ANCOVA and discriminant analyses are either more suitable for providing the tools to judge the reliability of any observed effects (*i.e.*, whether an observed difference is due to a treatment effect or to random sampling variability) or for research problems in which the criterion variable is categorical, and the predictor variables, metric.

The explanation of the two stages of the data analysis is set out below.

At the first stage, descriptive analysis was made of the responses. This included the size of the corporate centre, business environment, and organisational structure.

Various descriptive statistics were calculated, including mean, standard deviation, minimum and maximum scores of the variables.

The second stage of the analysis of the data consisted in testing the hypotheses formulated in Chapter 5. For Hypothesis 1, the paired ttest for planning and control variables was conducted with the aim of finding out whether the scores of planning influence variables were significantly different from those of control influence variables. Regression analysis was carried out to determine the relationship between planning and control influences.

Concerning Hypotheses 2 and 3, the scattergrams for the size of the corporate centre, and for planning and control influences were plotted, which was followed by the multiple regression for the size of the corporate centre and planning and control influences. Pearson's product moment coefficients were computed for the size of the corporate centre, planning and control influences.

For Hypothesis 4, multiple regression was made for all respondent firms. Dummy variables were used to test the effect of the employee size on the relationship between the size of the corporate centre and linkages. Multiple regression was further made for the firms with over 5,000 employees.

6. 10. Limitations of the Methodology Used

In any research, the design chosen presents its advantages and limitations, and the present one is no exception. The advantages have already been referred to in the preceding section. It is necessary, then, to call attention to the limitations.

One limitation relates to the sampling procedure adopted. A nonprobability sampling procedure makes it impossible to estimate the probability that a given firm of the sampling frame be included in the sample. This means that the size of the "sampling error" is unknown, and there is no objective method to evaluate it. However, it does not mean that the sample is less representative than if a probability procedure had been adopted.

A second limitation is that the measurement of many variables is based upon respondent perceptions. Despite the precautions taken to ensure the reliability of the SD-based interval scales, it cannot be guaranteed that the respondents' opinions were the same at two different points of time — even if they are not far away from each other. Nonetheless, as mentioned in the preceding section, it was necessary to use perception measurement, due to the nature of several elements of strategic management which are not amenable to being quantified.

An alternative method to measure each "construct" would have been not to use a single perception variable, but to use a series of objective but indirect variables, which might measure, when taken all together, each "construct". Examples of this measurement are Pugh *et al.* (1968) and Child (1972) who measured the formalisation and centralisation of the organisation as a whole by means of a large number of nonperception, indirect variables.

Despite the fact that this method may ensure the achievement of highlevel objectivity, the method presents at least three types of inconvenience. First, it takes for granted that selected indirect variables are good estimators of the direct variable. Second, to take all the indirect variables as a whole implies integrating them into a sole and general measurement. As the weight or relative importance of each indirect variable in the general index is not known, the weights are deemed to be equal, which may be a crass error. Therefore, the average obtained from these objective measurements can be even more biased than the simple perception of a respondent. Third, this indirect method can be too laborious when many "constructs" are to be measured at the same time (which is the case with the present research). The use of indirect variables for each of twenty-one perceptual variables contained in this research would have required an extensive questionnaire exacting great efforts from respondents, which would have made the questionnaire survey impracticable.

6. 11. Summary

In this chapter, all methodological aspects of the present research have been reviewed. In the first place, it was mentioned that the present research consists of a questionnaire survey with a large number of firms, for the purpose of obtaining empirical evidence to test the relationships of causal type between the size of the corporate centre and various variables, expressed as hypotheses in the preceding chapter. It was also mentioned that the sampling frame of this research is composed of 375 Japanese firms listed on the First Sections of the Tokyo, Osaka and Nagoya Stock Exchanges, operating in manufacturing field in Japan, and with over 1,500 regular full-time employees. The sampling frame represents a wide range of production technology. The instrument used to gather data was a structured questionnaire. Fiftyfive firms returned the completed questionnaire.

The research variables were "operationalised" by means of perceptual and non-perceptual type questions, for the measurement of which semi-interval and interval scales were respectively employed. For the analysis of the data, three stages were followed using several multivariate techniques.

Finally, two limitations of the methodology used were discussed. One of these relates to the non-probability procedures employed to select the sample, and the other to the use of perceptual measurements to "operationalise" some variables of the conceptual scheme. CHAPTER 7: PRESENTATION AND ANALYSIS OF THE DATA

7. 1. Introduction

The preceding chapter described the methods used for the gathering of the data to test the hypotheses of this research. This chapter presents, analyses and comments on the results obtained from the statistical treatment¹ of the data thus gathered. Case studies of Japanese firms are described to illustrate the results. Empirical studies in Western firms are provided to show them "in the round".

As briefly mentioned in Section 6. 8 in Chapter 6, the data were analysed at two stages — namely, descriptive analysis and hypothesis testing.

Section 7. 2 deals with the first stage of the analysis. First, a descriptive analysis of the responses was conducted. This included the size of the corporate centre, the business environment (stability of business environment, diversification, and production technologies) and the organisational structure (divisional form and functions performed by divisions and by the corporate centre).

The descriptive analysis at the first stage also included the twosample t-test in order to examine if there was a statistically significant difference between the respondent firms and the sampling frame. Various descriptive statistics were also calculated, including frequency, mean, standard deviation, minimum and

¹ All the statistical data processing was conducted by the use of the SPSS statistical package (Release 4).

maximum scores of the variables. Correlation and multiple regression analyses were also conducted to see if there was significant relationship between the centre's size and environmental variables.

Section 7. 3 describes the second stage of the data analysis. It consisted in testing the hypotheses formulated in Chapter 4.

To test Hypothesis 1 about planning and control influences, the paired t-test for planning and control variables of each respondent firm was conducted in order to find out if the scores of planning influence were significantly different from those of control influence. A scattergram was plotted to visualise the distribution of planning and control variables. Regression analysis was carried out to determine the relationship existing between planning and control influences.

For Hypotheses 2 and 3, which posit the relationship between the size of the corporate centre and planning and control influences, the scattergrams for the size of the corporate centre and planning and control influences were plotted. Then, a multiple regression analysis was conducted, separately, for planning and control influences, with the aim of finding out association among variables related to planning influence, control influence, and the size of the corporate centre. Pearson's product moment correlation coefficients were calculated for the size of the corporate centre, planning and control influences.

211

Lastly, for Hypothesis 4 concerning the size of the corporate centre and linkages, a multiple regression analysis was made for all respondent firms. Dummy variables were then used to test the effect of the employee number on the relationship between the size of the corporate centre and linkages. As a consequence, the respondent firms with over 5,000 employees were tested with the multiple regression analysis.

Section 7. 4 summarises and evaluates the results. Findings are mentioned concerning the hypotheses formulated. Findings not mentioned by way of hypotheses but relating to the size of the corporate centre are also summarised.

Section 7. 5 adds data from case study materials in the sample which illustrate and give more meaning to the results obtained from the statistical analysis.

Section 7. 6 compares the results with those from other empirical studies on Western firms.

Section 7. 7 summarises the chapter.

7. 2. Descriptive Analysis of the Respondent Firms

Before proceeding to the hypothesis testing, it would be useful to have a general view of the respondent firms, through descriptive analysis, in such aspects as the size of the corporate centre, environmental elements, production technologies, and organisational structure. Such a general view may lead us to a better interpretation and understanding of Japanese strategic management in transformation today.

Five steps have been taken with the aim of preparing a general view of respondent firms.

First, the size of the corporate centre is analysed. Analysis of environmental elements comprising business environment stability, diversification, and production technologies follows. Third, the organisational structure of the respondent firms is examined. Fourth, the functions performed by divisions and the corporate centre are analysed. The size of the corporate centre is regressed in relation to environmental elements, organisational structure, and corporate and divisional functions.

7. 2. 1. The size of the corporate centre

Measured by the number of full-time regular employees, the size of the corporate centres of the respondent firms ranges from a minimum of 86 to a maximum of $5,049^2$. The average size is 638 (See Table 7. 1.).

Mean	637.58	
Standard deviation	794.75	
Minimum	86.00	
Maximum	5,049.00	

Table 7. 1.: Data on the size of the corporate cer
--

Note n = 55

The size of the corporate centre is likely to be associated with the number of regular full-time employees. Accordingly, a regression analysis for size and employee number was carried out. The result showed that, as expected, the size of the corporate centre was positively associated with the total number of employees with the multiple R = .4871 and p = .0002. The regression equation was expressed as the following equation 7. 1.

Size	=	370.4746092	+	.033837b	(Number of
employee	es).				
Notes		a = Significant at	.0024	level	
		b = Significant at	.0002	level	
		n = 55			

This finding agrees with Young's survey result (1993) that companies with more employees tended to have larger headquarters.

² The average of the ratio = (Size of the corporate centre)/(Total regular full-time employees) was 9.1 per cent. The maximum case (a giant heavy industry machinery manufacturer) representing 30 per cent contrasted with the average. The company was contacted for a clarification on this point by a letter, first, and by a phone call. The person who filled in the questionnaire verbally confirmed it.

7. 2. 2. Environmental elements

The questions about environmental elements included in the questionnaire concern the environmental stability of the business, its diversification, and the routine nature of production technology.

Business Environment

Respondents were asked to select from a seven-interval scale ranging from extremely stable (1) to extremely unstable (7). As shown in Table 7. 2., the mean of scores was 4.182. Of the fifty-five respondent firms, 27.2 per cent answered that their business environment was stable.

As also shown in Table 7. 2., it was found that 72.8 per cent of the respondent firms believed that their business environment was either at the middle point (4 in the questionnaire scale) or unstable. Scores concentrated at points 3, 4, 5, and 6. Extreme points (1, 2 and 7) represented only 7.2 per cent.

Table 7. 2.: Stability of business environment

cores of the environment(*)Perc	entage (%)Can	BUIST	ve percentage (%)	
1	3.6		3.6	
2	1.8		5.4	
3	21.8		27.2	
4	32.7		59.9	
5	27.3		87.2	
6	10.9		98.1	
7	1.8		100	
Mean	4.182			
Standard deviation	1,219			

The higher the score, the less stable is the business environment (1 = extremely stable; 4 = mid-point; 7 = extremely unstable)n = 55

Diversification

*:

Respondents were asked to assess how diversified³ their firms were on the basis of a 7-interval scale with 1 being extremely diversified and 7, not diversified at all.

As shown in Table 7. 3, of the fifty-five respondent firms 21.8 per cent answered that they were diversified, 56.3 per cent (corresponding to scales 5, 6 and 7) considered themselves not diversified.

³ Based on Rumelt (1982), it was defined in the questionnaire that a firm is not diversified, if related products or product lines sharing the same common market or technology represent over 95 per cent of the total turnover; and a firm is diversified, if unrelated products or product lines which do not share the same common market or technology represent over 30 per cent of the total turnover.
Table 7. 3.: Diversification

1	9.1	9.1
2	7.3	16.4
3	5.5	21.9
4	21.8	43.7
5	21.8	65.5
6	27.3	92.8
7	7.3	100
Mean	4.509	
Standard deviation	1,709	

*: The higher the score, the less diversified is the firm (1 = extremely diversified; 4 = mid-point; 7 = not diversified at all).
 n = 55

Production Technologies

Production technologies were ranked in terms of seven levels of routineness, based on a scale developed by Woodward (1965) and adapted by Kagono *et al* (1985). The categories range from *continuous process* (the most routine = 1) to *custom technology* (the most non-routine = 7).

As shown in Table 7. 4. the average score of the fifty-five respondent firms was 3.833. Out of fifty-five firms 40.1 per cent considered their production technologies to be routine, while 29 per cent regarded them as non-routine.

Table 7. 4.: Production technologies

1	7.3	7.3
2	16.4	23.7
3	16.4	40.1
4	30.9	71
5	3.6	74.6
6	20	94.6
7	3.6	100
Mean	4	
Standard deviation	1.645	

*: The higher the score, the less routine are the production technologies (1 = extremely high routineness; 4 = mid-point; 7 = extremely low routineness).

n = 55

The size of the corporate centre and environmental elements

Pearson's product moment correlation coefficients were computed for the size of the centre and the environmental elements (stability of business environment, diversification, and production technologies) and two-tailed significance was tested.

The result showed that the size of the corporate centre in terms of the number of full-time employees was significantly correlated with the firm's diversification at the p < 0.05 level (See Table 7. 5.). A negative correlation coefficient means that the less diversified the firm, the smaller the size of the corporate centre. Two other environment variables, namely, the stability of the business environment and the routineness of production technologies, did not have significant correlation with the size. Table 7. 5.: Correlation between stability, diversification, and the size of the corporate centre

	Stability	Diversi- fication	Production tech- nologies	Size of the centre
Stability	1.00		and the second	
Diver- sification	.0525	1.00		Section 1
Production tech-	-0.1144	1419	1.00	
Size of the centre	-0.0833	-0.3140*	.1725	1.00

Notes: The scores of the questionnaire go from 1 (extremely stable; extremely diversified; and extremely high routineness) to 7 (extremely unstable; not diversified at all; and extremely low routineness)⁴. * significant at .05 level. n = 55

Based on the foregoing result that the diversification was significantly correlated with the size of the corporate centre, a stepwise multiple regression analysis was conducted to find out whether there was any significant associative effect among environment variables and the size of the corporate centre.

Due to the entry and elimination criteria for stepwise regression being F < 2.71 and p < .10, only the diversification variable was selected for regression, as expected. As shown in Table 7. 6., the multiple R was .31396. Equation 7. 2. shows the regression which was significant at .05 level.

⁴ A negative correlation coefficient means that the less diversified the firm, the smaller the size of the centre.

Size = 1295.864915^a - 145.990203^b(Diversification)......(7. 2.)

Notes

a = Significant at p < .01b = Significant at p < .05n = 55

Considering the fact that the diversification scores range from 1 (extremely diversified) through 4 (midpoint) to 7 (not diversified at all), the negative slope of the regression equation is interpreted to indicate that the higher the score of diversification (that is, the lower the diversification), the smaller is the size of the corporate centre. In other words, the actual relationship is "positive".

Table 7. 6.: Multiple regression on the size of the corporate centre and diversification

Multiple R R Square Adjusted R S F Signif F	Square	.31396 .09857 .08156 5.79548 .0196			
Variables	Ba	SE Bb	Betac	Т	Sig T
Diversi- fication	-145.990203	60.642766	313959	-2.407	.0196
Constant	1295.864915	292.094239		4.436	.0000
Notes ,	a) Coefficient b) The Stand regression eq c) The standa n = 55	s of the regres ard errors for 1 Juation. ardised regress	ssion equation. the coefficient ion coefficient	s of the ts	

7. 2. 3. Organisational structure

Issues concerning divisional organisation form, functions performed by divisions, and functions performed by the corporate centre are addressed below.

Divisional form

80 per cent⁵ of the respondent firms reported that they had divisional structure. This contrasts with Kagono *et als* finding (1985) that 59.8 per cent of their sample firms had adopted divisional form. Japan Development Bank (1989) stated that 62.3 per cent of the listed firms with over 1,000 full-time employees had adopted divisional form.

Functions performed by divisions

As shown in Table 7. 7., the percentage of the functions performed by divisions is similar to that reported by Kagono *et al* (1985).

Divisi Funct	onal ions	Kagono et al (1985)a	This research ^b
Production		59.8%C	73.3%
Sales		91.5	86.7
Marketing p	lanning	82.6	82.2
Personnel		35.5	37.8
Control		40.1	35.6
Finance		12.2	17.8
Basic resea	rch	28.5	20.0
Applied rese development	earch and t	75.6	71.1
Purchasing		52.4	35.6
Notes.	n = 2 $n = 5$ $Perce$	91 5 ptage of firms in which t	the function is performed by

Table 7. 7.: Comparison of divisional functions

divisions.

⁵ This percentage is high due to some bias which could not be determined. This point will need further empirical research.

Table 7. 8. shows the number of functions performed by the divisions of the respondent firms. The median number of functions performed by divisions was four. On average, the divisions of the respondent firms had 4.6 functions. The maximum number of functions devolved onto divisions was nine, and the minimum number, one.

Number functions performed divisions	of Percentage (%) by	Cumulative percentage (%)	
1	6.7%	6.7%	
2	2.2	8.9	
3	20.0	28.9	
4	24.4	53.3	
5	24.4	77.8	
6	6.7	84.4	
7	2.2	86.7	
8	6.7	93.3	
9	6.7	100.0	
Mean		4.600	
Standard dev	riation	2.016	
Minimum		1.00	
Maximum		9.00	
Note n	= 55		1.5

	Table	7.	8.:	Functions	performed	by	divisions
--	-------	----	-----	-----------	-----------	----	-----------

As shown in Table 7. 8, nearly 80 per cent of the fifty-five respondent firms had less than five divisional functions. In 26.7 per cent of all divisionalised firms, the divisions did not have a production function and in 14.3 per cent they did not have a sales function. These figures imply that divisions in Japanese firms are often not self-contained and that functional autonomy is emphasised over the integration of functions at the division level. This fact reflects the organisational structure which Kagono (1992) named a "functionalised divisional" structure.

In the functionalised divisional structure, divisions are not selfcontained and complete; namely, they do not contain a full range of functions. Therefore these divisions are not closed, or autonomous. They are grouped into functional divisions performing production and development, and into functional divisions performing sales and marketing units. They need to conduct transactions with each other. Either through strategic and financial plans or through budgeting process, these divisions have to effect mutual coordination. In general, mutual co-ordination is carried out laterally among the functional divisions through horizontal information exchange.

Functions performed by the corporate centre

All respondent firms reported that finance and accounting, personnel and human resources management, and strategic and business planning functions are performed by the corporate centre⁶. Other functions performed by over 80 per cent of the corporate centres of the respondent firms are secretarial and legal services, government and public relations, marketing services, information systems, and management services (See Table 7. 9.).

⁶ Kagono *et al* (1985) reported that in their sample Japanese firms centralised their personnel and finance functions.

Table 7. 9.: Corporate functions

Function	Percentage of firms		
	(a)	(b)	
Finance	100	0	
Secretarial and legal	98	2	
Personnel	100	0	
Public relations	65	35	
Strategic planning	100	0	
Marketing	92	8	
Internal audit	44	56	
Information system	89	11	
Management service	87	14	
R+D	23	77	
Engineering	46	54	
Physical Distribution	23	77	
Purchasing	23	77	
Property service	64	37	
Administration	54	46	
Others	56	44	

Notes: (a) = Percentage of firms whose corporate centres perform the function. (b) = Percentage of firms whose corporate centres do not perform the function. n = 55

As shown in Table 7. 10, the corporate centre in over 70 per cent of the fifty-five respondent firms performed between 8 and 12 functions. The median of the number of functions performed by the corporate centre was 10, whereas the mean was 10.635, with the standard deviation being 2.701. The minimum and maximum numbers of corporate function were 6 and 16, respectively.

Number of corporate functions	Percentagea	Cumulative percentage
6	3.8%	3.6%
7	5.8	9.6
8	17.3	26.9
9	11.5	38.5
10	13.5	51.9
11	5.8	57.7
12	25.0	82.7
13	3.8	86.5
14	0	86.5
15	5.8	92.3
16	7.7	100.0

Table 7. 10.: Number of corporate functions and their frequency

Mean	10.365	
Standard deviation	2.701	
Minimum	6.00	
Maximum	16.00	

Note: a) Percentage of the respondent firms whose corporate centres perform the number of the functions. n = 55

The size of the corporate centre, and divisional and corporate functions

As shown in Table 7. 11, both the number of functions performed by the corporate centre and the number of functions performed by divisions⁷ are positively associated with the size of the corporate centre (measured by the number of regular full-time employees) at p < 0.05. The multiple R was .43448. This association can be expressed as the following regression equation (7. 3.):

['] The number of the functions performed by divisions and that of the functions performed by the corporate centre were negatively correlated (-.0872), but it was not statistically significant (p=.578).

Size = $-1055.254471 + 101.409814^{a}$ (N° of divisional functions) + 137.020956^b (N° of corporate functions)......(7.3)

Note:

a = Significant at .05 level. b =Significant at .05 level. n = 55.

Table 7. 11.: Multiple regression on the size of the corporate centre and the numbers of corporate and divisional functions

Multiple R			.43448			
R Square			.18877			
Adjusted R	Sq	uare	.14821			
F			4.65400			
Significant	F		.0152			
Variables		Ba	SE B ^b	Beta ^C	т	Sig T
N ° corporate	of	137.020956	60.864260	.321828	2.251	.0299
functions N [°] divisional	of	101.409814	45.119019	.321307	2.248	.0302
functions Constant		-1055.25447	599.114202		-1.761	.0858
Notes		a = Coefficien b = The Stand regression eq	nts of the regre dard errors for uation	ession equation the coefficient	on. hts of the	

c = The standardised regression coefficients

n = 55.

7. 2. 4 Summary of the descriptive analysis

The descriptive analysis presented in this section is summarised in

Table 7. 12.

Table 7. 12.: Association of the size of the corporate centre with employees number, business environment, and organisational structure

Variables	Statistics
Number of employees*	
Number of employees Multiple regression	Significant at $p = .0002$ Size = 370.474609 + .033837 (Number of employees) Multiple R: .4871
Business environment*	
Stability Diversification Production technologies Multiple regression	Not significant Significant at p < .05 Not significant Size = 1295.864915 - 145.990203 (Diversification) Multiple R: .31396
Organisational structure*	
N° of divisional functions N° of corporate functions Multiple regression	Significant at $p < .05$ Significant at $p < .05$ Size = -1055.254471 + 101.409814 (N° of divisional functions) + 137.020956 (N° of corporate functions) Multiple R: .43448

Note * n = 55

7. 3. Tests of the Hypotheses

So far data on the general aspects of the fifty-five respondent firms have been presented including the size of the corporate centre; environmental elements comprising the stability of business environment; diversification and production technologies; and the organisational structure, including divisional form and the number of divisional and corporate functions.

This section analyses empirical evidence to test the hypotheses formulated in Chapter 5 on the relationship between the size of the corporate centre, and vertical and horizontal information flows.

7. 3. 1. First proposition

The first proposition was about the relationship between the size of the corporate centre and the vertical information exchange between the centre and divisions and functional departments.

This proposition relates to the way the centre influences managers lower down and affects the decisions taken by them. Two dimensions of the centre's influence were identified: planning influence and control influence.

Hypothesis 1

Hypothesis 1 postulates that the influence the Japanese corporate centre exerts is higher on planning and lower on control. In order to test this hypothesis, the paired t-test was made for planning and control variables of each respondent firm.

As shown in Table 7. 13., the means of planning and control variables were 2.4091 and 2.7030, respectively. Since the higher the score, the less are the planning and control influences (Scores were typically -1 = extremely important; 4 = midpoint; and 7 = not important at all), this means that average planning influence was higher than average control influence.

The paired t-test was carried out for planning and control influences in order to see if the difference between these two dimensions of influences was significant or not. The t-score resulting from the paired t-test was -2.72 with the degree of freedom being 54. Since the 2-tail probability was .009, the difference between planning and control influences could be considered significant. Therefore, it can be concluded that planning influence is higher than control influence.

Table 7. 13.: Paired t-test for planning and control influences

	Number of cases	Mean	Standard deviation	Standard error	Max- imum	Mini- mum
a = Planning vari- ables	55	2.4091	.788	.106	4.00	1.03
b = Control vari- ables_	55	2.7030	.610	.082	4.17	1.33
Differ- ence mean = (a-b)	Standard deviation	Standard error	t Value	đ. f.	P	
2939	.801	.108	-2.72	54	.009	3.00

d.f. = degrees of freedom p = probability of 2-tail test

As shown in Figures 7. 1 and 7. 2, it was also found out that both the planning and control variable scores were located in the area delimited between 1 and the mid-points 4. As the scores range from 1 (extremely important) through 4 (midpoint) to 7 (not important at all), this means that both planning and control influences are located in high-high area. This conclusion is also confirmed by the fact that averages of planning and control influences are located in Quadrant (IV) of Figure 7. 2.





Figure 7. 2.: Planning and control influences



Note

* Average control and planning influences of the respondent firms

The strategic management style corresponding to the above, according to Goold and Campbell's classification (1987: 42), might be "strategic programming". Goold and Campbell describe the characteristics of strategic programming as follows:

"...the centre is involved in developing strategies at the business unit level, but in addition it attempts to set clear performance targets and to insist that they are met. The style embraces detailed planning, central sponsorship of strategy and tight control against financial objectives and strategic milestones."

However, as shown in Table 7. 14., there was a significant difference between the incentives variable (Variable 6) and the average of other control variables (Variables 1 - 5) at p < .001. This points to the possibility that in Japanese firms the control influence is strong, but the type of pressure exerted on managers is different compared with Western firms. Goold and Campbell (1987: 41) report:

"Where bonuses are linked to performance targets, or where careers are at risk, the pressure of the control process is enhanced. Companies such as Hanson Trust and Vickers believe in providing bonuses that can amount to 30 per cent or more of salary. A number of companies link promotion prospects to the achievement of planned targets."

Both the data analysis of the present research and the survey of relevant literature (e.g., Kono, 1984) on Japanese management provide evidence that the situation at Japanese firms is quite different from Goold and Campbell's description. Further discussion will be made on this point in the next chapter.

Table 7. 14.: t-tests for paired control influence variables

	Number of cases	Mean	Standard deviation	Standard error	
a = Mean of Var.1 to Var.5	55	2.3455	.627	.085	
b = Var. <u>6</u>	55	4.4909	1.399	.189	
c = Differ- ence mean (a -b)	Standard deviation	Standard error	t Value	d. f.	p
-2.1455	1.425	.192	-11.17	54	.000
Notes .	Var.1 Var.2 Var.3 Var.4 Var.5 Var.6 d. f. =	(Financial (Financial (Financial (Financial (Monitorir (Division's degrees of	returns) targets ver targets) targets as ng) performar freedom	rsus strate a promise) nce and ma	gic objectives) anager's career)

Table 7. 15. shows the regression equation of planning influence with control influence (See also Figure 7. 1.). The slope was .47077 > 0. With the scores ranging from 1 (extremely important) through 4 (midpoint) to 7 (not important at all), the positive slope can be interpreted to indicate that the stronger the control influence, the weaker is the planning influence. In other words, the actual relationship is "negative" and there is a kind of "trade-off" relationship between these two influences within Quadrant (IV).

Table 7. 15.: Regression equation of planning influence with control influence

Corre-	.36466	R ^{2*}	.13297	S.E. of Est. ^b	.74018	Sig. C .0062
Inter- cept ^d (S. E.)	1.1366	(.4574)	Slope ^e (S. E.)	.47077	(.1651)	

Notes.:

b The standard error of estimate

a Pearson's r

c Statistical significance. In this case, p < .01

d The intercept of the regression line.

e The slope of the regression line.

* The coefficient of determination.

Hypotheses 2 and 3

Hypotheses 2 and 3 are about the relationship between the size of the corporate centre and planning and control influences.

Hypothesis 2 postulated that the size of the corporate centre measured in terms of the number of regular full-time personnel is positively associated with the degree of planning influence the corporate centre exerts.

In an attempt at scanning graphically this relationship, a scattergram was plotted for the size of the corporate centre and planning influence. For clearer representation⁸, the size of the corporate centre was converted into log scale. Fig: 7. 3. seems to point out that there is a positive, albeit not strong, association between the size and planning influence.

⁸ As absolute size of the corporate centre ranges from 86 to 5,049 (Cf. Table 6. 1.), it is easier to detect some association if the Y-co-ordinate scale is "shortened" by means of the conversion of Y-values into log values before plotting them.

In order to test the relationship between the size of the corporate centre and planning influence, a multiple regression analysis was conducted, using planning variables as independent variables.

The results are summarised in Table 7. 16. F-value (p = .6713) shows that the multiple regression equation did not have statistical significance. Similarly, none of the planning variables showed any significant t-values regarding their regression coefficients. These results imply that there was no statistically significant association between the size of the corporate centre and planning influence, as conjectured. Hypothesis 2 should then be rejected.





234

Table :	7. 16.	.: Multiple	regression	on	the	size	of	the	corporate
centre	and	planning	influence						

Multiple R R Square Adjusted R Square F		.27871			
		.07768			
		03761			
		.67380			
Signif F		.6713			
Variables	Ba	SE B ^b	Betac	т	Sia T
Var. 6	-157.995019	140.416391	200464	-1.125	.2661
Var. 4	72.191261	73.902520	.158509	.977	.3335
Var. 3	203.833492	115.898992	.286495	1.759	.0850
Var. 1	42.889771	136.343774	.073732	.315	.7545
Var. 5	71.852886	113.723235	.111779	.632	.5305
Var. 2	-87.808843	147.484350	146419	595	.5544
Constant	211.548800	411.564958		.514	.6096

Notes a = Coefficients of the regression equation.

b = The Standard errors for the coefficients of the regression equation.

c = The standardised regression coefficients n = 55

Now that Hypotheses 1 and 2 have been discussed, the only remaining hypothesis forming part of the first proposition is Hypothesis 3. It posited that the size of the corporate centre is negatively associated with the control influence the corporate centre exerts.

Following the example of the testing of Hypothesis 2, a scattergram was plotted for the visual examination of the relationship between the size of the corporate centre and control influence. As shown in Figure 7. 4., visually it was impossible to determine if there was any relationship between these two sets of variables.





As in the case of Hypothesis 2, a multiple regression analysis was then conducted. The multiple regression equation does not have statistical significance at p = .1492 (See Table 7. 17). Accordingly, Hypothesis 3 should be rejected⁹.

[°]The correlation coefficients between planning influence, control influence and the size of the corporate centre were computed. With p = .846 (size and planning) and p = .648 (size and control), none of these coefficients was significant. This confirms the results obtained in the multiple regression.

Table 7. 17.: Multiple	regression	on the	size	of the	corporate
centre and control in	fluence				

						_
Multiple R		.41929				-
R Square		.17581				3
Adjusted R S	quare	07059				
F		1.67089				
Signif F		.1498				
Variables	Þð	ee ob	PotoC	Ŧ	EI- T	
Var C	40.127100	JE D*	Dela-	0.40	51g 1	
var. 6	-48.127199	51.18/524	130781	940	.3519	
Var. 1	-98.063139	70.136206	209582	-1.398	.1686	
Var. 4	-153.715673	127.263430	237451	-1.208	.2331	1.5
Var. 5	-10.111928	68.799072	021005	147	.8838	
Var. 2	-61.150472	65.316581	139950	936	.3539	
Var. 3	335.129109	121.955614	.557233	2.748	.0085	
Constant	884.746246	322.574553		2.743	.0086	
Notes	a = Coefficier	ts of the rear	ession equation	n	A NUMBER OF STREET	
	h - The Stan	dard arrow for	the coefficier	ate of the		

b = The Standard errors for the coefficients of the regression equation.
 c = The standardised regression coefficients
 n = 55

In summary, on the one hand, it was found that Japanese firms have large corporate centre (9.1 per cent over the total employee number) and, on the other, they have more "hands-on" management style and a high planning influence is exerted from the centre.

Therefore, though Hypotheses 2 and 3 of this research were rejected, it might be concluded that "hands-on" approach increases the size of the corporate centre, because it reflects more information-processing. Further research will be needed on how differently planning and control influences are combined in "handson" and "hands-off" approaches.

7. 3. 2. Second proposition

The second proposition concerned the relationship between the size of the corporate centre and horizontal information linkages. This relates to the way that Japanese firms are organised and managed.

Linkages can result from the sharing of common clients, of common production technologies, etc.

The existence of linkages requires co-ordination. In Japanese firms co-ordination is made through horizontal information exchange among divisions or functional departments. Thus, information exchange is decentralised, but the decentralised approach to information exchange is claimed, *e.g.*, by Aoki (1988), to be associated with a centralised approach to administrative management, which leads to increase in the size of the corporate centre.

Therefore, Hypothesis 4 postulated that the size of the corporate centre in Japanese firms measured by the number of regular fulltime personnel is positively associated with the linkages among divisions or departments.

Multiple regression was used to probe into the possible association existing among linkage variables and the size of the corporate centre. For n = 55, the multiple R was .41666. R square was .17361. Since p = .0867 > .05, the result was not statistically significant. As shown in Table 7. 18., Var. 5, Var. 3 had significant terms in t-tests. These variables correspond to product development and production technologies, respectively. Overall, the hypothesis did not seem to hold.

Table 7. 18.: Multiple regression on the size of the corporate centre and linkages

Multiple R		.41666		•	
R Square	R Square				
Adjusted R S	Square	.08928			
F		2.05875			
Signif F		.0867			
Variables	Ba	SE Bb	Betac	т	Sia T
Var. 5	-301.402542	101.389991	691433	-2.973	.0046
Var. 1	-137.138266	75.436376	301271	-1.818	.0752
Var. 2	22.665618	76.709291	.047429	.295	.7689
Var. 3	191.976825	95.381243	.466727	2.013	.0497
Var. 4	139.605422	121.088720	.296874	1.153	.2545
Constant	917.159945	350.541621	1. S.	2.616	.0118
Notes	a = Coefficien b = The Stand regression eq c = The stand Var. 1 = Cus Var. 2 = Puro Var. 3 = Prod Var. 4 = Ma	its of the regre dard errors for juation. dardised regres tomers chases betwee duction techno rketing skills	ession equatio the coefficien sion coefficien en divisions logies	n. nts of the nts	

n = 55

However, based on a scattergram to visualise the relationship between the size and linkages, it was suspected that there was some intermediate effect upon the relationship between the corporate centre size and linkages by the stratification of the total employees. All of the fifty-five respondent firms were stratified by the use of critical points relating to 5,000 and 10,000 employees, and the sample was divided into three groups of similar sub-sample size¹⁰.

Dummy variables were then used to test the effect of employee number on the relationship between linkages and the size of the corporate centre. As Table 7. 19. shows, the multiple R increased to .56895, and there was a significant change in the value of Sig F (= .0084). Var. 6 and Var. 7, which represent the dummy variables, had p = .02 and .0165, respectively.

Therefore, a sub-sample was taken of the respondent firms with over 5,000 employees (n = 23) to further test the relationship between the linkages and the size of the corporate centre¹¹.

¹⁰ The cutting-off was rather arbitrary; however, the sample could be divided into two (over 5,000 or less than 5,000 employees):

Classification by employee numbers	Employee number
< 5,000 employees	32
10,000 > employees ≥ 5,000	13
10,000 ≤ employees	10

¹¹ The multiple regression analysis on the other two sub-samples was also run, but neither of them was significant.

Table 7. 19.: Multiple regression on the size of the corporate centre and linkages with employee size-related dummy variables

		SALAS I.	Sale aller		
Multiple R		.56895			
R Square		.32371			
Adjusted R S	quare	.22080			
F		3.14544			
Signif F		.0084			
Variables	Ba	SE B ^b	Betac	Ŧ	Sig T
Var. 7	149.277079	63.612268	.305994	2.347	.0233
Var. 5	6.186777	26.417574	.051463	.234	.8159
Var. 1	23.797367	18.852140	.188951	1.262	.2132
Var. 6	-182.930246	73.513566	340697	-2.488	.0165
Var. 2	.013495	19.968474	.000098	.001	.9995
Var. 3	-23.688762	25.430380	210969	932	.3565
Var. 4	16.464608	30.052206	.125047	.548	.5864
Constant	181.372835	103.909414		1.745	.0876

Notes

a = Coefficients of the regression equation.

b = The Standard errors for the coefficients of the regression equation.

c = The standardised regression coefficients

Var. 1 = Customers

Var. 2 = Purchases between divisions

Var. 3 = Production technologies

Var. 4 = Marketing skills

Var. 5 = Product development

Var. 6 = Dummy variable¹² Var. 7 = Dummy variable

n = 55.

¹² Dummy variables used were:

	Var. 6	Var. 7
Employees < 5,000	0	. 0
5,000 ≤ Employees ≤ 10,000	0	1
Employees > 10,000	1	0

The result was that the multiple regression equation was statistically significant at p < .05. The multiple R was .70723. The linkage variables with significant t-scores were Var. 5 and Var. 1, which represented shared product development skill and shared customers. The regression coefficients were, for each of these two variables, -514.103714 and -449.636439, respectively (See Table 7. 20.).

Chapter 8 will discuss the question why these two areas, namely, shared product development skill and shared customers, are associated with the size of the corporate centre.

As the variable scores range from 1 (extremely important; or definitely true) through 4 (midpoint) to 7 (not important at all; definitely incorrect), the negative coefficients actually indicate a 'positive' relationship, which means that when there is a stronger linkage among divisions or functional departments, as the size of the corporate centre increases¹³.

¹³ While Var. 4 (Marketing skills) showed a positive relationship with the size of the corporate centre, Var. 2 (Purchase from each other) and Var. 3 (Production technologies) were negatively associated with the size of the corporate centre. However, the regression coefficients of these three variables did not have statistical significance (p = .6875, .2677, and .0835, respectively).

Table 7. 20.: Multiple regression on the size of the corporate centre and linkages for the respondent firms with over 5,000 employees.

Multiple R		.70723						
R Square Adjusted R Square F		.50017						
		.34398	.34398 3.20223					
		3.20223						
Signif F		.0342						
Variables	Ba	SE B ^b	Beta ^C	T	Sia T			
Var. 5	-514.103714	178.465293	744639	-2.881	.0109			
Var. 1	-449.636439	164.161475	623743	-2.739	.0146			
Var. 2	242.735606	211.357056	.275820	1.148	.2677			
Var. 4	-110.184582	268.957186	.720836	1.846	.0835			
Var. 3	452.087935	244.899883	.720836	1.846	.0835			
Constant	2369.886653	990.113501		2.394	.0293			
Notes	a) Coefficient b) The Stand	s of the regres	ssion equation	n. ts of the				

regression equation.

Var. 1 = Customers

n = 23

Var. 4 = Marketing skills Var. 5 = Product development

Therefore, it may be concluded that Hypothesis 4 holds with a modification relating to the size of the firm measured by the number of employees and that, with this modification, linkages in terms of shared customers and shared product development are positively associated with the size of the corporate centre.

c) The standardised regression coefficients

Var. 2 = Purchases between divisions Var. 3 = Production technologies

7. 4. Recapitulation and Evaluation of the Analysis Results

This section contains a recapitulation and evaluation of the results to reassess the propositions and hypotheses of this research. These results will be illustrated with some case materials in the next section.

The results of the data analysis indicate that the propositions and hypotheses need to be modified. There are four sets of findings.

In the first place, although planning influence was higher than control influence, it was found that the average scores of both these dimensions were located in the high-high area, namely, in the area of high planning and high control influence. This means that, although the planning influence and control influence were in a negative association, it would still be difficult to separate the sizeincreasing effects of planning influence (Hypothesis 2) and the sizediminishing effects of control influence (Hypothesis 3). Therefore, the separate effects of planning and control influences on the size of the corporate centre, as evidenced by some researchers (*e.g.*, Young, 1992), should be modified in the case of Japanese firms.

In the second place, no statistically significant association exists between the size of the corporate centre and the planning influence or between the size of the corporate centre and the control influence. On the other hand, it was found in this research that (1) the sample Japanese firms have a large corporate centre representing an average 9.1 per cent of the total employees and (2) the sample firms exert from their centre high planning and high control influences.

In the third place, in the sub-sample of the firms with over 5,000 employees, the size of the corporate centre and the linkage variables were positively associated. The linkage variables which were statistically significant in this association were shared product development skills and shared customers.

In the fourth place, though not in our main hypothesis testing, it was found that diversification and the size of the corporate centre were positively associated. The same was true of the number of corporate and divisional functions, both of which had a significant positive relationship with the size of the corporate centre. Likewise, employee number was positively associated with the size of the centre.

7. 5. Illustration of the Findings with Case Materials

For the illustration of the findings mentioned in the foregoing 7.3 and recapitulated in 7. 4., this section provides some case materials:

(1) on the central role the corporate centre plays in the promotion of new business, creating a linkage in the form of shared customers — Kawasaki (with over 5,000 employees);

(2) on the high planning and high control influences exerted by *large* Japanese corporate centres — Matsushita; and

(3) on how one of the linkages (shared product development) is coped with by the corporate centres of the firms with over 5,000 employees — Sharp and Kao.

In this research, of the four hypotheses, Hypothesis 1 and Hypothesis 4 were found to prove significant, albeit with some qualifications. The Matsushita case relates to Hypothesis 1, whereas the Kawasaki, Sharp and Kao cases correspond to Hypothesis 4.

The case materials are a mixture of the primary and secondary sources. The Kawasaki case is drawn from the professional experience of the dissertation's author. Takahashi (1981) and Bartlett and Ghoshal (1988) provide case descriptions for Matsushita, and both Nonaka and Takeuchi (1995: 160-196) and Bartlett and Ghoshal (1989), for Sharp and Kao. Sharp and Kawasaki are sample firms of this research.

7. 5. 1. Relationship between the corporate centre and shared customers

The following Kawasaki case illustrates the role the corporate centre plays in the company-wide promotion of new business, creating a linkage in the form of shared customers. It also gives an insight into how the size of the centre increases to cope with a larger horizontal information flow as a consequence of the creation of linkages.

Kawasaki Heavy Industries

(Employees: 17,495; the corporate centre: 1,500)

Kawasaki Heavy Industries (KHI) is the nucleus of the Kawasaki group formed by KHI, Kawasaki Steel and K-Line. It is one of a few cases among the sample firms which have the complete divisional form in which divisions are given the majority of functions including personnel management (although the recruitment is made by the corporate centre).

After the Petroleum Crisis of 1970s, KHI became knowledgeable in JIT (the Just-in-Time System) by despatching a group of engineers to Toyota, and adapted it to different manufacturing methods from mass production products (such as motorcycles) to made-to-order products (such as huge structures). In 1985, Price Waterhouse Spanish Firm (PW) decided to seek for its Management Consultancy Division support from a Japanese manufacturing firm concerning JIT. PW instructed the author of this dissertation to establish contact with KHI.

The author sent a private letter to Mr K Andoh, a personal friend of his, and, at that time, the manager of Battle Ship Department at KHI. (The letter was a private one instead of a formal and official one from PW, since an official letter would have been taken up and studied through the official channel which might have hindered Mr Andoh from making use of his personal network within the firm.)

The letter was followed by several telephone calls from the author. Mr Andoh contacted the Production Technology Department functioning as the internal technology adviser to KHI line divisions. The reaction of the Production Technology Department was negative.

Mr Andoh then transferred the issue to the Corporate Marketing Group (CMG)¹⁴. The head of CMG considered it very interesting and initiated contacts with several divisions that might help him. An invitation was then sent to PW so that its managers might visit KHI factories and explain PW's interest in signing a technical agreement with KHI.

PW's visit to Japan was followed by a trip of the CMG's head to Spain to reconfirm PW's intention. Although the head of CMG was not sure if JIT could technically be implemented in Spain, he

¹⁴ The Corporate Marketing Group (CMG) is a staff department. It is therefore a cost centre, its function being the co-ordination or sometimes the conducting of businesses which are not taken care of by line divisions. A typical example is a feasibility study of a shipyard in Pakistan carried out by CMG. The study was made on behalf of Japan's Plant Export Association. KHI's name did not appear anywhere in it (but if it had been done by line division it would have charged the cost to the Association and KHI's name would have been mentioned in the report). Such a feasibility study was a necessary step for KHI to bid for the actual construction work of the shipyard. It enabled KHI to incorporate some information that could direct the entity inviting tenders towards the specifications of KHI.

appreciated the keen interest of PW in the issue and determined to do his best to help. Moreover, KHI was interested in experimenting in "pure" technology transfer as business, because, although it had a number of overseas licensees, the technology transfer had always been linked to the sale of equipment components.

In view of the Production Technology Department's reluctance towards this business, the head of CMG recruited from line divisions an engineer specialising in JIT and a co-ordinator who could speak Spanish (*i.e.*, the size of the corporate centre increased). At the same time, the head of CMG initiated his efforts to build up a network of sympathisers to this business in relevant divisions. Above all, the Consumer Product Division (CP) and Structures Divisions showed interest in it. The KHI's president was also supportive, because he had travelled many times to Spain for marine turbine business and liked the country.

The licence agreement between PW and KHI was signed in 1986. The CMG signed on the part of KHI, a situation observed by KHI people to be quite unusual, since it was not customary that a staff department should appear as the active part of a contract. Such an arrangement was necessary, however, given the nature of the business, which comprised several divisions.

The head of CMG was in a privileged position to promote a business that might need the support of different divisions. First, he had a wide personal network cutting across the organisation. He firmly believed that a person's capability is a function of the level of network he or she may have. In a firm like KHI, famous for its individualistic approach to business, this is very important. This atmosphere is summed up in the expression: " I will do it, because the person in charge is Mr so-and so."

The head of CMG had been assigned to several departments. He first worked for Corporate Planning, which allowed him to get acquainted with people from different line divisions. After working for Corporate Planning for ten years, he was transferred to the Prime Mover Division, where he worked as a marketing specialist. Eight years afterwards he was appointed the head of KHI's Houston Office, before finally landing in CMG. KHI's personnel department rotates KHI's staff through different posts which could offer them a chance to widen their personal network. This seems particularly true in the case of the head of CMG, since he is a graduate of the Tokyo University. KHI, as a typical "old-establishment" firm used to set great store by the graduates from state universities, above all, those of Tokyo and Kyoto.

Second, CMG was assigned the role of distributing the Business Development Fund, used for the promotion of new businesses that could be thwarted by the lack of budget from line divisions. Although the amount of the Fund is as small as 200 to 300 million yen (and a normal business project could be 500 to 600 million yen), the allocation process of the Fund enabled CMG to foster good relationships with line engineers.

This operational mode has some rather peculiar characteristics. For example, even if the agreement became quite well known among KHI's staff and everybody including the chairman and the president

knew that the head of CMG was instrumental for it, he could not expect any promotion nor special remuneration. The promotion of a person is not dependent on each separate and individual event, but is based on a global review of his or her performance. This dissuades individuals from seeking their personal advancement by short-winded projects to the detriment of their firm.

Furthermore, for KHI staff, there is no such thing as Kawasaki Heavy Industries, Ltd., but many "Kawasakis" based on business divisions. It is therefore very important that the corporate centre plays an active role in the issues related to the firm-wide and across-the-board businesses.

7. 5. 2. High planning and high control influences by large corporate centres

The Matsushita case illustrates how a large corporate centre exerts high planning and control influences. Avoidance of initiatives from the centre, the budget as a contract between the centre and the businesses, frequent monitoring of performance, and strong pressure on businesses with performance problems are the means of high control influence used by the centre. A high planning influence is seen in centralised functions, values, people development, control of prices, and financial growth.

Matsushita Electric Industrial

(Employees: 49,762; the corporate centre: 3,500)

At a cursory glance, Matsushita is a Financial Control company, namely, a company with the high control and low planning influences from the centre (Goold and Campbell, 1987). The characteristics of a Financial Control company stand out in Takahashi's (1981) description of Matsushita. Thus, at Matsushita, the centre's relationship with the divisions and departments is characterised by (1) the delegation of responsibility for strategy development to business unit and even profit centre level, (2) a transparent flow of information between the businesses and the centre, (3) a focus on the budget process as the most important planning process, (4) avoidance of initiatives from the centre in the form of themes or thrusts, yet liberal use of suggestions from the centre where appropriate, (5) insistence that the budget is a contract between the centre and the businesses and that annual financial performance is the critical measure of achievement, (6) frequent monitoring and review of performance against budget, and (7) strong pressure immediately applied to businesses with performance problems and management changes readily made in the face of continued problems.

High control and high planning influences

Despite the characteristics of a Financial Control company described above, Matsushita has some added ingredients that are not observed in UK's Financial Control companies. The difference lies in the centralised functions performed by the large centre, combined
with strong shared values, people development, control of prices, and financing growth (of which later).

Therefore, Matsushita may be classified as a high control and high planning influence firm on the ground that, in addition to the high control influence which is explained below, the centre exerts in a subtle manner planning influences through:

- strategic themes revolving around the centre's vision on the firm's 'distinctive competence';
- broad strategic thrust (although the businesses are free to devise plans to achieve their targets, the guidelines are provided by the centre);
- specific suggestions.

High control influence: divisionalisation with business autonomy

The company is decentralised. Each product and some functions, such as distribution, have been set up as business units ('departments' in Matsushita terminology). The general managers in charge of the departments are autonomous. They are expected to develop strategy and advance their businesses without assistance from the centre. All inter-company trading is carried out at arm's length — at the market rate. There are no centrally imposed transfer prices.

Departments are grouped into divisions of like products. The division managers have an important role in monitoring, encouraging and developing each department and the product range of the

division. But the division layer is 'transparent' — the department general manager is responsible for performance directly to the head office.

High control influence: Budget system

The central control process is focused on a 6-month operating budget. Whereas the Financial Control companies in the UK all use an annual budget cycle (Goold and Campbell, 1987), Matsushita has a 6-monthly cycle. The budget is reviewed by the finance function and the division managers, and then goes to the president for approval. It is then handed back to the department as the 'Basic Management Programme' and becomes — in the company's own words — a 'promise between the president and the manager'.

The programme is monitored closely, and variances frequently result in the formation of a task force, including controllers from the finance function, to get the department back on track. Failure to deliver on the 'promise' is viewed with great concern by both the department and the centre, and may result in the manager being assigned a different position 'where his talents are more suited to the task'.

In a sense, Matsushita's 6-month cycle gives a stronger short-term focus than the normal British cycle of 12 months. Directors in Britain complain about the short-term focus of capital markets that respond to 6-month earnings statements (Goold and Campbell, 1987). But the response within Matsushita to variances against the 6-month programme of a department is much more acute and focused than the reaction of the capital markets. The director of finance explained the system: 'If a department's performance is good, we will give them a word of praise. If performance is bad, we offer understanding and advice. But we will not give them any special financial help. We expect their accounting to be merciless. We know there is a risk involved in leaving performance to the managers; but we view that risk as tuition for learning management.' The tightness of the control is evident in the phrase 'we will not give them any special financial help'. By this he means that the department is expect to solve the problem without help from the centre. Failure to solve the problem is likely to lead to a management change.

A further illustration of the 'merciless' accounting is given by Takahashi (1981):

"When new departments are established, the company deliberately underestimates the department's need for working capital. Facing a shortage of operating funds, the new department must work hard to retain and accumulate its earnings."

Almost all the basic elements of Matsushita's management system fit the Financial Control model — the company is divided into small autonomous units; the centre is careful not to interfere or overrule department managers; controls are based on a 6-monthly promise between the centre and the units; and failure to deliver the promise is treated severely.

Financial results have been excellent and, with a strong focus on profitability and a 'merciless' approach to poor performers,

Matsushita is one of the most profitable manufacturing firms in Japan, with a normal return on sales of more than 7 per cent (Takahashi, 1981). Matsushita also builds strong loyalty and high levels of satisfaction among managers. There are critics of the management systems, but they believe that overall the firm is well managed.

To this point, Matsushita looks like a Financial Control company. But there are important differences. First, the centre does occasionally become involved in directing the company as a whole. This is not normal for Financial Control companies. Second, Matsushita has achieved remarkable organic growth, and departments within the firm have attacked and won in many fierce, fast-growing and technically advanced markets, such as video recorders. Both the rapid organic growth and the bold strategies of the departments are unusual features. As Goold and Campbell (1987) point out, the Financial Control style is not likely to be suitable where business unit managers are trying to build positions in fast-moving, R&D led, highly competitive markets. Matsushita appears to have found a combination of values, systems and policies that go a long way to avoiding the disadvantages of the Financial Control style, without losing the benefits¹⁵.

¹⁵ Hamel and Prahalad (1989) have developed an explanation for why Japanese firms appear to have tight short-term controls and yet can follow long-term ambitions. Traditionally this is achieved by the corporate level strategy which lays down medium-term ambitions. The Japanese do not seem to have corporate strategies. Rather they are committed to a 'strategic intent' a target that defines 'winning' for them. Hamel and Prahalad (1989) argue that it is the combination of a 10-, 15- or 20-year strategic intent and a 1- or 2-year budget that enables Japanese firms to grow so successfully.

High planning influence: Five features

Matsushita has five features of its management system that do not occur in Financial Control companies:

- 1. centralised functions
- 2. values
- 3. people development
- 4. control of prices
- 5. financial growth.

These five features seem to be channels through which planning influences, epitomised in strategic themes, broad strategic thrust, and specific suggestions are exerted. These are the features of the Matsushita system that help to reduce risk aversion and avoid the low levels of organic growth that are typically associated with tight financial controls.

1. Centralised Functions

Finance, personnel and training are all fully centralised. As a result, the centre contains 3,500 people, which compares with the 100 or so people at the centre of the high control influence firms like Tarmac, the Union Pacific, ABB, GE, IDG and CNN (Goold and Campbell, 1987; Peters, 1992; Young, 1993).

In the finance function all controllers and accountants report directly to the centre. Controllers and accountants are seconded from the centre in divisions. (There are as many as 1,500 in the whole organisation.) As such, those in the divisions and departments have two bosses, reporting both to the centre and to their general manager. But their first loyalty is to the discipline of Matsushita; they have a vital role in the control system. It is likely that these controllers and accountants may be a conducting channel of the centre's 'specific suggestions'.

The other two central functions, personnel and training, exist to create 'harmony' within the company. Takahashi (1981) quoted the personnel manager as saying : "The only word that best describes our personnel philosophy is "harmony". We are often told by our friends that Matsushita employees have no appeal outside the 'firm. They say that we have no variety, that we are all alike. But I don't think so. To me we are quite varied.' In other words, the central role of these two functions is to help build and maintain the Matsushita culture.

People (what they believe and the way they think) are seen as the critical resource. As a result, no employee with qualifications beyond those of basic schooling is hired without central personnel screening. Every management promotion is also carefully reviewed by the centre. Moreover, all graduates go through introductory training which involves 3 weeks of general education at the centre, 3 months working with retailers and 3 months working in a factory. The 3 weeks at the centre is devoted almost entirely to philosophy and to explaining Matsushita values. It is to be emphasised, however, that this is not an unusual practice among Japanese firms. What sets Matsushita apart from others is its thoroughness.

Another outstanding aspect of the central personnel activity is job rotation. Each year 5 per cent of the employees rotate from one business unit to another. Equal numbers of managers, supervisors and shop floor employees are involved.

By controlling promotions, indoctrinating and vetting new recruits and moving people about, the personnel and training functions can keep the central values of Matsushita pure and can maintain its strong culture which are instilled as explained below.

2 Values

In 1932 Konosuke Matsushita became involved in a religious movement in Japan and came to recognise that employees would benefit from spiritual leadership. It comes clear to me', he said, 'that people need a way of linking their productive lives to society' (Takahashi, 1981). He set about developing a philosophy that articulated his own beliefs. 'A business should quickly stand on its own', said Matsushita, 'based on the service it provides society. Profits should not be a reflection of corporate greed, but a vote of confidence from society that what is offered by the firm is valued. When a business fails to make profits it should die — it is a waste of resources to society.'

Takahashi (1981) relates that:

Chairman Matsushita taught his executives that the autonomous management method was still effective today; detriments could be complemented by managerial devices. Interpreting his ideology, all executives held firm belief that what was essential for the divisional organisation system was everyone's way of thinking towards it, not a mere shape or pattern; unless top management, more than anyone else, displayed 'go-ahead' attitude, the system could not work; its success was up to top management whether it had a clear-cut policy or not.

Ever since the foundation of the corporation, Mr Matsushita summoning all corporate members early every January, delivered a speech on his business philosophy and policy. In addition to it, whenever the corporation encountered newly developed situations, he expressed his basic attitude and direction to be taken by the members to cope with them.

The late Chairman Matsushita's business philosophy supports the divisional organisation system. Takahashi (1981) cites some examples of the Chairman's philosophy:

Continue business expansion within credit limit! The scale of our business activities is to be determined by society which evaluates our contribution to it. Business expansion should be made on a profitable basis within the limit beyond which reasonable profit could not be secured.

Human unity and harmony are indispensable for job achievement.

Always show clear objectives and perform business, unifying all members as a united whole! The foundation for unity and harmony is sincerity.

A division manager explained the importance of this philosophy: 'Many Westerners tend to smirk at the higher purposes to which Japanese organisations avowedly dedicate themselves, and assume that these calls to higher values are just thinly disguised manipulation. But it becomes a belief system for thousands of people who work for that company — a human value beyond profit to which their productive lives are dedicated.' Values such as 'continue business expansion within credit limit' and 'human unity and harmony are indispensable for job achievement' go a long way to easing some of the problems created by decentralisation and by tight controls. By a combination of the value system and the behaviour of central managers, the divisions and departments keep their eyes on the long term, while being pushed to deliver short-term performance.

The firm has some systems that help to reinforce the need for cooperation, harmony and the greater good. For example, once a month department and division managers meet the chairman for an informal coffee morning. This provides managers with a chance to co-ordinate with others in the firm. It is also an opportunity for the chairman to reinforce the values and ideals. Each morning before work, every employee recites the company values. It seems silly to Westerners', said one executive, 'but it's like we are all a community' (Takahashi, 1981). Also three or four times per year each employee is expected to give a 10 minute talk to his work group on the firm's values and its relationship to society.

These systems make sure the values are imprinted on the firm. But there are also systems to reinforce the importance of short-term performance. Division managers attend quarterly peer reviews, at which their performance is shared with the group. The managers are grouped into A, B, C and D classes. The A-rated managers give their presentations first and the Ds come last. No one is openly criticised, but each leaves determined not to be in the C or D class at the next presentation.

The observers of Matsushita agree that its value system is an essential ingredient to the success of the management approach. It makes it possible for managers to interpret the activities of the centre in the most favourable light. Criticism from the centre can then be viewed as 'training'. The transfer of poor performers to more suitable jobs is seen as character-building for the individual and "does not cause others to view them negatively" (Takahashi, 1981).

It is the value system that makes managers feel comfortable with tight financial controls. And it is their belief in the values that causes them to maintain a long-term view. Despite the centre's insistence on meeting the budget, one manager is quoted by Takahashi (1981) as describing the performance evaluation system as taking the long-term view:

[They judge us] I'm sure, by our total abilities as men rather than our fragmented capabilities as managers. Even if our department's performance is very poor, we will not be fired. Rather we will be judged by the reliability of our policy and programmes for the upcoming periods. Speaking for myself, I don't feel like I am being evaluated constantly, nor do I feel guilty for one year's bad performance. I am evaluated, I believe, on the long-term perceptions rather than short-term performance.

To an outsider, these words seem loaded with contradictions. But the contradictions are judged against a value system. By managing these values, Matsushita appears to have squared the circle; to have achieved a focus on short-term performance without sacrificing a long-term commitment.

3. People Development

One of the central values is the importance of people. In the relationship between the centre and the departments, whether at a formal control meeting or an informal gathering, attention is always paid to the issue of how best to develop the managers involved. A pure task orientation is virtually non-existent. Every situation involves a task which is seen also as training opportunity. Even when the centre is probing and challenging managers about some trouble spot, it is not seen as threatening. 'The curious twist to these interventions', says one Matsushita division manager, 'is that they never seem like inquisitions. You feel senior management is doing it to train you, to build your competence for the day when they will no longer be around.' This fundamental tenet provides a human dimension even for a trouble-shooting session that might otherwise seem harsh and cold-blooded.

Matsushita's corporate centre believes that it is more important to get managers to understand what needs doing than to fix the problem immediately. Once they understand, they will be able to avoid the problem next time. A product group manager talking about conflicts of opinion said 'we try to get the facts out on the table and let reason speak for itself. We build "acceptance time" into these discussions. By that I mean thinking about things. We press — but we always try to allow people to come around to a point of view in their own way.'

4 Control of Prices

Up until the 1970s founding chairman Matsushita himself controlled price level. Product prices could not be changed without his approval. Since the 1970s this responsibility has been devolved to the division levels.

The importance of central decision-making on prices is unclear. It obviously must have a major influence. The most important influence is presumably to signal to the general manager of the department whether he should be going for growth or not. It would be fascinating to know on what criteria the centre choose one price from another and how the negotiation between the centre and the department is carried out. It seems that prices are frequently set below the market level. The centre's contribution is, therefore, a critical strategic input.

5 Financial Growth

The departments must submit 60 per cent of their profit to the centre, but they can retain 40 per cent for expansion and growth. In addition, departments can borrow from the centre. The 60-40 split is not sacrosanct — some departments may be asked for more or less than 60 per cent — but it is a normal split.

The interesting aspect of this system is that it acknowledges that some of the profit and all depreciation belongs to the department. If the department think they can get a better return than the money market rates offered by the centre, they are free to invest as they

please. This system is not operated in divisionalised firms in UK or USA.

Retained earnings can be spent freely. Managers believe that this creates an added incentive to strive for high profitability, as well as helping to focus attention on growth. It also means that capital approval requests are limited to major proposals, where a business wants to reach beyond its own resources to expand. The centre can then deal with these requests in much the same way that the capital markets would.

7. 5. 3. Relationship between the corporate centre and shared product development

The Sharp and Kao cases depict the way the corporate centres of Japanese firms (with over 5,000 people) cope with one of the linkages, namely, shared product development. This relates to the finding that in firms with over 5,000 employees, the size of the centre and the linkage variables are positively associated.

The Kao case describes the way the corporate centre creates information flows and makes them fluid through the use of free access to information, open floor allocation, open meetings, and fluid personnel change. All these information flow devices are designed to equip its divisional system with the mechanisms for active information sharing and direct employee interaction.

Kao Corporation

(Employees: 7,021; the corporate centre: N/A.)

Kao is at present Japan's leading household and chemical products manufacturer. It has 18 divisions, including the Home Product Division, Sanitary Product Division, and Chemical Division. Kao's corporate centre believes that direct communication among the employees of different divisions becomes limited in the conventional division system, and is thus striving to achieve active interaction among its employees. It also believes that direct interaction among employees generates creative ideas. But organisational members cannot interact equally when holding different amounts of information. To correct this imbalance, "information sharing" is regarded by Kao's centre as the principal tenet that defines Kao's organisation. The centre has introduced various mechanisms and support systems that assure the sharing of information. They include "free access to information," "open floor allocation," "open meetings," and "fluid personnel change."

To assure "free access to information," computer systems have been introduced throughout the Kao organisation, with all information being filed in a database (Bartlett and Ghoshal, 1989). Through this system, anyone at Kao can tap into databases included in the sales system, the marketing information system, the production information system, the distribution information system, and the total information network covering all of its offices in Japan. The unique feature of this system is that any company member, no matter what his or her position or to what section she or he

belongs, has full access to the database (except for a limited amount of personnel information).

In the "open floor allocation" system, the divisions and functional groups within Kao are all configured around a large open space. Half of the executive floor space, for example, is occupied by an open space called the "decision-making room." In fact, executives rarely stay in their own offices. Divisional heads hold meetings at the round table located in one of the large open spaces. In the laboratories, researchers do not have their own desks, but share big tables.

The corporate centre further accelerates information sharing and employee interaction through "open meetings." Any meeting at Kao is open to any employee, and top management meetings are no exception. Any employee can attend the relevant portion of the meeting and make his or her opinion known. Through this practice, the centre can acquire insights from those most familiar with the issues at hand, while employees can gain a better understanding of the general corporate policy.

What is known within Kao as the R&D conference is typical of these "open meetings." Through this conference, which is held every quarter, the corporate centre learns about research projects directly from the researchers, while research members gain an opportunity to voice their opinions directly to the corporate centre (including top management). This conference, open to anybody outside R&D, is regularly attended by some 1,800 people (out of a total 7,000 employees).

Kao's corporate centre has designed the "fluid personnel change" system to enhance interaction among members with different experiences (Nonaka and Takeuchi, 1995). For instance, researchers in one division are often transferred to other divisions or to other functional areas, such as sales or finance, on a "whoever is needed, wherever he or she is needed" basis. This kind of active job-rotation system, especially among R&D people, enhances the accumulation and sharing of knowledge and promotes interdisciplinary product development within the firm. For example, Kao entered the cosmetics market in the mid-1980s with the introduction of a skincare product called "Sofina" that resulted from the co-operative effort of people working in surface-active science and those in biological skin care.

As mentioned above, Kao's organisational structure can be explained as a divisional system equipped with various mechanisms for active information sharing and direct employee interaction. The structure can be considered to be flat, as all members of the organisation are placed on an equal footing and create new knowledge through the direct interaction of their respective functions.

Although Kao's organisational structure is basically a traditional divisional structure, its corporate centre has designed a mechanism by which, when it comes to new-product development, marketing innovation, and human resource management issues, the divisions co-operate in a horizontal manner (Nonaka and Takeuchi, 1995). Besides the vertical product divisions, the corporate centre

organises three "horizontal" committees to deal with crossdivisional strategic issues. They are the Division Strategy Committee, Marketing Innovation Committee, and Human Resource Management Committee.

The Division Strategy Committee, which meets twice a year and is attended by the vice presidents and division heads, determines which new products need to be developed by cross-divisional team. Examples of cross-divisional projects include a hair treatment project for controlling hair hardness, a new cosmetics project for men's use, and an ultrathin paper products project for such products as nappies and sanitary napkins. Members of these teams come from the various divisions as well as from the R&D and production departments.

Kao's project team activities are not limited to new product development; they are applied widely throughout the entire organisation, as in the case of the Marketing Innovation Committee. This committee examines common marketing issues across divisions, including effective market research techniques, the appropriate advertising media mix, and environmentally conscious packaging. The committee forms Marketing Innovation Projects, which tackle these issues and develop appropriate recommendations.

The Human Resource Management Committee is another horizontal cross-divisional committee, which meets once a month and is attended by division heads. This committee reviews the overall status of human resource development across divisions, and is also

responsible for selecting the appropriate members from each division for new product and marketing innovation projects.

Kao's corporate centre applies the idea of horizontal, crossdivisional team activities even to its own staff operation. Each "centre" specialising in public relations, legal affairs, accounting/finance, or human resources carries out normal staff functions, but cross-centre project teams are formed in order to deal with corporate-wide issues, such as the reduction of fixed cost, risk management, working-hour reduction, and the simplification of corporate staff operation.

Sharp Corporation

(Employees: 22,700; the corporate centre: 1,420)

Since its foundation in 1912, Sharp has had a reputation for creating new products — from a self-adjusting belt buckle and Sharp pencil in the early years to liquid crystal projection TVs and electronic organisers today. Our case study will focus on how the corporate centre created an organisation for the pursuit of innovation.

The centre has organised Sharp's day-to-day R&D activities in a typically traditional and hierarchical manner (Nonaka and Takeuchi, 1995). The actual structure consists of the Corporate R&D Group, Business Group labs, and Business Division labs. These three structures are separated on the basis of the time frame required for technological/product development. The Corporate R&D Group deals with long-term (three or more years) R&D themes; and Business Group and Division labs with short-term (one year and half or shorter) themes.

These three structures are aligned as a traditional hierarchy, with research findings passed down the structures in a top-down fashion. Research findings at the Corporate R&D Group are transferred to the research laboratories of the nine Business Groups, then to the labs of each Business Division. During the product development process, rough prototypes are prepared in advance. Researchers at both the Business Group and the Business Division labs, who receive the prototypes, sometimes relocate to the Corporate R&D Group for a few months to improve their understanding of the research findings from the Corporate R&D Group. When research findings have to be utilised quickly for product commercialisation, the Corporate R&D Group's researchers, in turn, sometimes move down to either the Business Group labs or the Business Division labs. Knowledge concerning R&D is transferred efficiently and combined effectively under this kind of hierarchical structure.

Various meetings or conferences are used to co-ordinate the activities of the laboratories at the three levels (Nonaka and Takeuchi, 1995). They allow R&D members at Sharp to share knowledge not only within each level but also across the different levels. The first is the General Technology Conference, which is held once a month and is attended by the president, vice presidents, executive directors, and managers of the nine Business Group laboratories. They discuss what sort of R&D activities should be conducted at each laboratory for the upcoming one-year period. These discussions, which deal with the grand design of corporate

R&D, often become heated and last as long as six hours, with a break for lunch. The second is the Laboratory Director's Conference, which is held once a month and is attended by managers of both the Business Group and the Business Division labs, the director of the Corporate R&D Planning Office, and the director of the Intellectual Property Office. The Laboratory Directors' Conference makes specific and detailed decisions, including when and how to transfer certain technology to the business groups and which collaborations are needed with an outside party. The third is the Technology Development Strategy Conference, which is held once a month in each Business Group. Participation in this conference is not necessarily limited to lab members of the Business Group, but can include members of the Product Planning Department and the Engineering Department, as well as selected individuals from the Corporate R&D Group. The Technology Development Strategy Conference is not merely a place for receiving technology from the Corporate R&D Group, but a place for deliberating actively on what kind of key technologies should be developed by each Business Group.

Sharp's R&D operations have a traditional hierarchical structure, but when it comes to new product development Sharp's corporate centre utilises the task-force organisation, which is a completely independent, parallel organisational structure. While normal product development activities are carried out within each Business Division, the strategically important product development projects are conducted under the "Urgent Project System." The members of the Urgent Project System are relocated from their original departments and work exclusively for the project team.

The Urgent Project System gives its members, who could be recruited from any section or department within the firm, the same "golden-badge" authority as corporate directors during the project period. Wearing the gold badge carries special significance not only for project members but for other employees at Sharp as well. Urgent Project members develop a priority product or technology within a year or two. But since it is managed directly under the president, the project budget is unlimited. People with the gold badge and their project are given top priority in using company facilities or equipment and in procuring materials.

In addition, members of the Urgent Project can be taken from anywhere in the firm at any time. A department may be deprived of its best people for over a year. The corporate centre has to make every effort during the initial stages to ensure that the system is enforced as originally intended. Each Business Division proposes projects that require company-wide development efforts and completion in a brief period of time. These proposals are either adopted or rejected, or "justified" at the above-mentioned General Technology Conference, the highest decision-making meeting at Sharp.

The success of the Urgent Project System led to changes in Sharp's organisational structure. Sharp's corporate centre recently initiated two strategy meetings — New Life Strategy Meeting and NEWING Product Strategy Meeting — in order to diffuse the Urgent Project idea widely within the entire organisation. In the New Life Strategy Meeting, held once a month and attended by the president, vice

presidents, and managers of the Business Group and the Business Division, the division managers explain new product development plans.

The NEWING Product Strategy Meeting is also held monthly and is attended by 20 people, including the president, vice presidents, and managers of the Business Group and the Business Division. The candidates for new product concepts are proposed by each Business Group or Business Division manager and reviewed for their originality and marketability.

Once a product development plan is recognised by New Life Strategy Meeting or NEWING Product Strategy Meeting, development work begins within the division. The authority given to the development team is similar to that given the Urgent Project, since development team members receive direct support from the president and have the right to ask for whatever co-operation they need from within the firm. It differs from the Urgent Project in that the members basically stay in their original division or department and conduct other work during the development process.

7. 6. Comparison with other empirical studies

This section compares the findings from Japanese firms with other empirical studies on the Western corporate centre. Child (1973c) conducted a study on the proportion of administrative or supportive employees. Though the concept of supportive employees includes those outside the centre as well, Child's study bears out the relationship between the size of supportive employees and information flows embodied in the organisation's complexity. Therefore, it points to the possibility of such relationship between the centre's size and information flows.

Cresap (1988) posits that information flows resulting from a more hands-on management style increase the size of the centre.

Young (1995) contends that Japanese centres are larger than their counterparts in Germany and UK, and that the number of functions performed by Japanese centres is larger than those of German and UK counterparts.

Child (1973c)

In his search for readily accessible data on the structure of organisations, Child (1973c) brings in the concept of the proportion of employees in administrative or supportive roles. He points out the progressive establishment of specialised functions by which supportive activities are removed from line managers into the care of staff employees.

He hypothesised several determinants of the supportive component, namely, staff employment in organisations and examined them in the light of data from 54 British manufacturing companies. These determinants include the size of the organisation, organisational complexity, spatial dispersion, technology, ownership and control, and membership of a larger group.

Child's definition of the supportive employment differs from that of the employment in the corporate centre. It refers to the total number of employees minus the combination of all direct employees and all managers of direct employment or production areas of work. Therefore it includes the supportive components outside the centre. Still, even taking into account this difference, Child's (1973c) study bears out some of our findings and hypotheses: namely, (1) the size of the centre is in direct proportion to the size of the organisation; and (2) the size of supportive component at the centre relates to information flows — the number of different relationships between personnel in an organisation and the range of specialised tasks they perform — which require larger investment in supportive component to co-ordinate and control.

In the first place, Child (1973c) found out that the relationship of supportive component employees to total employees is strongly linear, with a very slight tendency for supportive employment to curve upwards as size increases. Size of organisation accounts for 91 per cent of the variance ($\mathbb{R}^2 = .92$) in supportive components.

Child's (1973c) findings suggest that the relative size of the supportive component in organisations will rise (1) the more these are spatially dispersed, (2) the more complex are their workflow operations, (3) the fewer workflow divisions they have, and (4) the larger the group of which they form part. Set against size of organisation, however, these other variables provide only a marginal increment to the prediction of supportive employment.

Of the 54 manufacturing companies studied, 51 provided Child (1973c) with a full set of accurate statistics on 16 specialised nonworkflow functions. The numbers in maintenance, finance, sales and service, transport and dispatch, quality control and public relations each have over 50 per cent of their variance accounted for by variation in total employment. In contrast, under 20 per cent of the variance of numbers in legal and insurance, market research, training, and employment — personnel — is accounted for by total size, while the figure drops below 10 per cent for the numbers in design and development, office services — statistics, filing, post, and so forth - and production methods. Two conclusions are suggested by these results: (1) factors other than size of organisation are associated with the numbers in certain specialised staff functions and (2) the relative size of each specialised functions is unlikely to be correlated highly in this sample of organisations with the relative sizes of all the other functions.

In the second place, Child (1973c) argues that complexity indicated by measures of horizontal and vertical differentiation and of specialist expertise is found to increase with a larger size of organisation. It is argued that the greater the complexity of relationships and tasks, the larger will be the relative investment in supportive personnel required to cope with consequentially increased difficulties of co-ordination and control. In particular, if bureaucratic solutions are applied to problems of co-ordination and control and new or more elaborate procedures are instituted, presumably more support staff will be required to operate those procedures.

In summary, Child's (1973c) findings point out that total employment in supportive activities displays a strong linear relationship to total organisational employment. Complexity, spatial dispersion, and the number of workflow divisions also provide a basis for predicting total employment in supportive activities.

Cresap (1988)

On the basis of interview and mail surveys with 45 leading British firms, Cresap (1988), a management consultancy firm, found four different types of corporate centre reflecting modes of directing the organisation — *targeting, guiding, directing, and running* corporate centres.

The fundamental management philosophy of the *targeting* corporate centre is 'hands off', underpinned by a belief that this philosophy has motivational benefits for the operating units. A very clear separation between the role of the corporate centre and the activities of the subsidiaries is emphasised. The corporate centre is viewed as 'an overhead'. The size of the centre is 0.1 to 0.7 per cent of the total employees.

The guiding corporate centre espouses a 'hands off' philosophy, but its approach is marked by a greater degree of involvement with the operating units. So while operating decisions are largely left to the individual business units, which are set financial targets and held accountable for their performance relative to these, the philosophy is supported by a desire to 'help' the operating units, and to foster the sense of belonging to an 'active corporate group'. The size of the centre is 0.2 to 3.4 per cent of the total employees.

The *directing* corporate centre takes more of a 'hands on' approach in management philosophy. It not only helps the operating units to develop strategic plans, but also provides directional initiatives from the centre, and gets involved in the implementation of business strategies and the integration of major business decisions. The centre has 1.8 to 9.1 per cent of the total employees.

The *running* corporate centre is the most involved of the four types of corporate centre. It is very much 'hands on' in orientation. There is often little effective separation between the corporate centre and the business. The role of the corporate centre is integral to the firm's activities. Plans, policies, and guidelines are developed and monitored from the corporate centre, major operating decisions are taken and, to all intents and purposes, the firm is run from the corporate centre. The traditional distinction between "line" and staff is often neither clear nor relevant. The provision of services to the operating units is a key function for such a corporate centre. The centre's size varies from 3.7 to 27.5 per cent of the total employees.

Cresap (1985) contends that the 'hands-on' approach in management philosophies reflected in a greater number of tasks leads to a larger size of the corporate centre. As the 'hands-on' approach suggests a higher vertical information flow between the centre and lower management, Cresap's contention can be reduced to the relationship between the size of the corporate centre and vertical information flow.

Young (1995)

Young (1995) compared the corporate centres in Germany, Japan and UK drawing data from Passau University's 45 management holding companies in early 1990s, from this research, and from Young and Goold (1993), respectively.

Young (1995) found that the Japanese sample has fewer larger companies than German and UK samples¹⁶. Whereas in the Japanese sample only two per cent represented firms with over 25,000 employees, the German and UK samples have 36 per cent and 35 per cent, respectively. Japanese corporate centres are substantially larger than those of German and UK multi-business firms. Median corporate centre staff in Japanese multi-business firms ranges from nearly 300 for firms with 5,000 employees to 500 for firms with 5-25,000 employees. Median corporate centre staff in German and UK multi-business firms ranges from 20 to 30 for firms with 5,000 employees to 80 for firms with 5-25,000 employees. It means that, in the lower range, Japanese firms have nearly 10 times as many centre as their German and UK counterparts, and, in the higher range, 4 times as many.

That the Japanese corporate centres are larger than their counterparts in Germany and UK has two implications. First, the "largeness" of the Japanese centres may be explained by the number of the functions performed by them, which leads to more need for information flow between the centre and lower

¹⁶ According to a t-test, the sample firms of this research come from the same population as the sampling frame. Therefore, the employee number of the population firms are not expected to be divergent from the sample firms'.

management. Young (1995) found that all three countries have the finance and control functions at the centre, but other functions like marketing, technical advice, information system, logistics and property management are rarely performed by the centre in Germany and UK firms.

Second, the "largeness" of Japanese centres may shed some light on the discussion detailed in Section 2. 3. 2 concerning alternative explanations of the centre's size. Such a vast difference in the size of the corporate centre in Japanese firms in comparison with German and UK counterparts makes it difficult to believe that it is due to the political explanation advanced by Parkinson (1958). His explanation is that departments grow because managers prefer to multiply subordinates not rivals; and that this empire building is more easily achieved in areas away from direct production, since in production a growth in numbers would often be constrained by technology and by established controls on work load, stock levels, and manning. Such a huge difference in the size as mentioned above among Japanese, German and UK corporate centres needs more "managerial" than political explanation, unless Japanese managers were more biased towards the exercise of political influence than their German and UK counterparts.

Discussions

Child's (1973c) and Cresap's (1988) were cited in this section to support the findings in this research that the centre's size is a function of information flow (linkages). They pointed to the possible relationship between the size of the corporate centre and information flow or the hands-on and hands-off management styles (which lead to the higher or lower degree of information flow). Young's (1995) study underlined the largeness of the Japanese centres compared with their German and UK counterparts.

5

The relationship which exists between the centre's size and information flow may be explained as a result of either managerial actions or "sociological" elements discussed in Subsection 2. 3. 2. The case materials of Japanese firms described in Section 7. 5., shed light on this question. They illustrate that the centre's size is a product of managerial actions taken for the centralisation of coordination of linkages and the implementation of the features of high planning influence. It means that for the implementation of high planning influence, the centre's size is essential.

The corollary of the centre's size being essential is that down-sizing and process re-engineering, when they affect the centre's size, may result in some difficulties in the flow of information in the firm. This corollary has a practical implication and, as such, will be discussed in Chapter 8.

7. 7. Summary

This chapter has presented the results of the data analysis in two stages. First, a descriptive analysis of the respondent firms was presented. The analysis included the size of the corporate centre, business environment variables (stability of business environment, diversification, and production technologies), and organisational structure variables (divisional form, functions performed by divisions and functions performed by the corporate centre).

The second stage of analysis consisted in testing the hypotheses formulated in Chapter 5.

Hypothesis 1 was confirmed by means of the paired t-test for the planning and control influences, which found that these two sets of influences were significantly different from one another and that the planning influence was significantly higher than the control influence at .01 level. A regression analysis was carried out to determine the relationship between the planning and the control influences.

To test Hypotheses 2 and 3, multiple regression analyses were made for the size of the corporate centre, the planning influence, and the control influence. Pearson's correlation moment coefficients were computed for these three sets of variables. All these statistical tests showed no significant results. Thus, Hypotheses 2 and 3 should both be rejected.

On the other hand, it was found in this research that (1) the sample Japanese firms have a large corporate centre representing an average 9.1 per cent of the total employees and (2) the sample firms exert from their centre high planning and high control influences.

For Hypothesis 4, the multiple regression was first analysed for all respondent firms. Dummy variables were used, which confirmed the significant effect (at .05 level) of employee size on the relationship between the size of the corporate centre and linkages. Afterwards, a multiple regression analysis was conducted which showed a significant (at .05 level) relationship between the size of the corporate centre and linkages for the size of the size of the corporate centre and linkages for the firms with over 5,000 employees.

In brief, of the four hypotheses, Hypothesis 1 and Hypothesis 4 were found to prove significant, albeit with some qualifications. The Matsushita case relates to Hypothesis 1, whereas the Kawasaki, Sharp and Kao cases correspond to Hypothesis 4.

For the illustration of these findings, some case materials were provided :

 on the central role the corporate centre plays in the promotion of new business, creating a linkage in the form of shared customers — Kawasaki (with over 5,000 employees);

(2) on the high planning and high control influences exerted by *large* Japanese corporate centres — Matsushita; and

(3) on how one of the linkages (shared product development) is coped with by the corporate centres of the firms with over 5,000 employees — Sharp and Kao.

A comparison of the findings with Japanese firms was made with other empirical studies on the Western corporate centre. Reference was made to three empirical studies of the corporate centre in Western firms.

These results and findings will lead to conclusions which will be presented in the next and last chapter.

CHAPTER 8: CONCLUSIONS AND IMPLICATIONS

8. 1. Introduction

This last chapter addresses itself to drawing conclusions and implications, based on the analysis of data in Chapter 7.

Section 8. 2 presents and analyse the three main conclusions which are derived from the four sets of findings mentioned in Section 7. 4.

Section 8. 3 connects the results of the present research with the principal works existing in the literature. Afterwards, it assesses the contribution of the present research to this area of management studies.

Section 8. 4 analyses the practical implications which the conclusions suggest for corporate management, and formulates, in accordance with these implications, a series of prescriptions for practising managers.

Section 8. 5 concludes with the recognition of the main limitations of the research. It is followed, in Section 8. 6., by some suggestions for studies which might be undertaken in future, taking advantage of experience gained in the course of the development of the present dissertation.

8. 2. Main Conclusions

As already mentioned, this section contains the central part of the chapter — the drawing and in-depth analysis of the main conclusions based on the four sets of findings described in Chapter 7. (Alternative explanations for these findings will be discussed, although the relationships between planning influence and the size of the corporate centre and those between horizontal information flow owing to linkages and the size of the corporate centre will be justified as the best explanations.)

In the first place, the association of the size of the corporate centre with linkage variables will be discussed.

Second, other variables, which did not form part of the hypotheses formulated in Chapter 5, but were found to have bearing on the size of the corporate centre will be treated. These variables were diversification, the number of divisional functions and the number of corporate functions.

Third, planning and control influences will be discussed in the light of results confirming or rejecting Hypotheses 1, 2 and 3.
8. 2. 1. The size of the corporate centre and linkages

The first conclusion of the research is that the size of the corporate centre measured in terms of the number of regular full-time personnel is positively associated with linkages among divisions or functional departments, when the firm has over 5,000 regular full-time employees. These linkages result from sharing 1) customers, 2) purchase of products, 3) production technologies, 4) marketing skills, and 5) product development among divisions or functional departments. From these linkages it was found that sharing customers and common product development are positively associated with the size of the corporate centre. (This conclusion results from the analysis presented in Section 7. 3. 2.)

Two issues related to the first conclusion: centre size and linkages, and the nature of linkages

There are two issues related to the foregoing conclusion that ought to be further worked out. The first is why the centre size showed a statistically significant relationship with the linkages, when the employee size was over 5,000. The second point refers to the linkages. Why are product development and shared customers associated with the size of the corporate centre? What is the mechanism by which the size of the corporate centre relates to the linkages?

1) The centre's size and linkages

Concerning the first issue, the total employee size might influence both the linkages and the size of the corporate centre. As the company size in terms of the total employee number increases, the number of divisions and functional departments will increase. Accordingly, the need will also grow for institutionalised mechanisms, such as "keiei kaigi", the committee of senior board members, to monitor and coordinate the larger horizontal information flow. The number of the staff at the corporate centre will, as a consequence, have to be increased to support these mechanisms.

Alternatively, it ought to be recalled that Equation 7. 1. in Chapter 7 evinced a statistically significant positive relationship between the size of the corporate centre and the number of employees. The growth in the number of employees, then, leads to increase in the size of the corporate centre. It is likely that there is a threshold of corporate size beyond which the supporting role of the corporate centre for the horizontal information flow among divisions or functional departments becomes instrumental. It is probable that, as the use of dummy variables showed, this threshold falls somewhere between 1,500 and 5,000 employees.

In connection to this, Child (1984) reports that, in a survey conducted by him, the more profitable and faster growing companies of 2,000 employees and above were those that developed bureaucracy in the form of highly compartmentalised jobs and areas of work, detailed

procedural and paperwork systems, long hierarchies, and delegation of routine decisions to lower level managers within precise discretionary limits. On the other hand, Child (1973c) argues that total employment in supportive activities displays a strong linear relationship to total organisational employment. The combination of these findings may at least partly shed light on this point.

Further research on this point will be needed, which should take into account the opinion of line managers on horizontal information flow as well as their assessment of the value added by the corporate centre. Section 8. 6. will further discuss this point.

2) Linkages: shared product development and shared customers

The second issue concerns the linkages. The sharing of product development and of customers showed a positive association with the size of the corporate centre at a significance level of p < .05 (See Table 7. 18.).

The close relationship between sharing customers and the centre's size may have something to do with the resolution of conflicting interests among divisions vying for the same customers. Co-ordination by the centre and intervention in differences and conflicts among divisions is needed.

The Kawasaki case mentioned in Section 7. 5 illustrates the mechanism by which the centre's promotion of new business for various divisions resulted in the increase in its size. The centre is more involved in integration than in mere conflict resolution.

The positive association between shared product development and the size of the corporate centre could be explained by the need to allocate the staff and financial resources of the firm, which can be done better by a centralised approach.

8. 2. 2. Diversification, divisional functions and corporate functions

The second conclusion that may be drawn is that the diversification of business and the number of divisional and corporate functions are positively associated with the size of the corporate centre.

Three issues related to the second conclusion (diversification, number of corporate functions, and number of divisional functions)

Some discussion will be held below on this conclusion. In the first place, the diversification issue, second, the number of corporate functions and, lastly, the number of divisional functions will be discussed.

1) Diversification

First, it was found that the sharing of common customers (market) was positively associated with the size of the corporate centre at p = .05level (See Equation 7. 2 and Section 8. 2. 1). An increase in diversification means a decrease in shared markets or customers (by related products or product lines)¹. It should follow, then, that an increase in diversification leads to a decrease in the size of the corporate centre, but what was observed in the data analysis suggested the opposite.

One possible explanation is that, as both diversification and the centre's size are closely associated with the company size, this common factor leads to a positive relationship between diversification and the centre's size.

An alternative explanation is as follows. Diversification may mean reduced linkage both in terms of shared customers and common product development, which has a reverse effect on horizontal information flow and reduces the size of the corporate centre. Nevertheless, diversification into new areas, industries, *etc.*, removed in their nature from the company's core business may require higher planning influence and hands-on approach, especially in its initial stage. This will result in a larger corporate centre compensating even

¹For the definition of diversification, see Question 2 (2) of the questionnaire: "Please consider your firm not diversified, if related products or product lines sharing the same common market or technology represent over 95 per cent of the total turnover".

for the size reduction because of smaller linkage and, therefore, in a net increase in its size.

Further study should focus on the diversification strategy of Japanese firms and their scale of economy, especially how different new businesses tend to compare with existing ones in terms of product and client base.

2) Number of corporate functions

Second, the increase in number of the functions performed by the corporate centre was positively associated with its size.

This seems to stand to reason. If the number of corporate functions increases, the size of the corporate centre must increase proportionately even if not to the level of 100 per cent (on the condition that, as the number of corporate functions increases, the number of staff required for each function does not decrease, which cannot be tested from the questionnaire data).

3) Number of divisional functions

Third, the number of divisional functions was positively associated with the size of the corporate centre. This is an interesting finding, because, if an increase in the number of divisional functions meant devolution from the corporate centre, the increase in the number of divisional functions should have led to reducing the size of the corporate centre. In aid of our efforts to explain this finding, some examples will be cited below.

The first examples concern the "co-existence" of large corporate centres and "strong" divisions. These were found among respondent firms. Those firms whose responses indicated that their divisions were "complete" in the sense that they performed major functions like production, sales, finance, personnel, etc., had nearly 10 per cent of their staff working at the corporate centre². Table 8. 1. shows that ten firms having relatively complete functions in their divisions still maintain a large corporate centre and centralised functions at it.

Another example is a Japanese automobile manufacturer³. Although it does not concern the centre-divisions relationship, it still seems interesting, because it sheds some light on the issue under consideration. Some years ago this Japanese automobile manufacturer transferred part of its corporate staff in charge of North American operations to USA. Its intention was to reduce the number of corporate

4) in-company interest rate

- 7) reporting system to the corporate centre
- 8) appraisal system of divisions performance

² Imanishi (1988) cited the following as necessary conditions for the divisional structure to be complete:

¹⁾ definition of the basic principles for the management of divisions

²⁾ management control system

³⁾ corporate overhead cost assignment to divisions

⁵⁾ assignment of capital to divisions

⁶⁾ in-company transfer pricing system and right to buy from other outside suppliers

³ An interview held in Barcelona in December, 1993 with a planning manager of a Japanese automobile manufacturer.

staff based in Tokyo and, at the same time, to place its staff near the car users' market. The transferred staff were considered to form part of the corporate centre in Tokyo, and not part of the North American subsidiary.

Table 8. 1.: Strong divisional functions and the size of the corporate centre

	Divisional functions								Corporate functions																		
(6)	,	2	3	4	5	6	7		9	12	1	2	3		5	6	7	8	9	10	11	12	13	н	15	16	(4)
14	1	0	1	1	1	1	1	1	1	89	1	1	1	1	0	1	1	1	1	0	0	1	0	1	1	1	75
14	0	1	1	1	0	1	1	1	1	78	1	1	1	1	o	1	0	1	1	0	1	0	1	0	0	1	63
11	1	1	0	1	1	1	1	1	1	89	1	1	1	1	1	1	1	1	1	ö	0	0	0	1	1	1	75
11	1	1	1	1	1	o	0	1	0	67	1	1	1	1	1	1	1	1	1	. 0	1	0	0	0	0	1	69
10	1	1	1	1	1	1	0	1	1	89	1	1	1	1	1	1	0	0	1	0	1	0	0	1	0	0	56
9	1	1	1	1	1	1	1	1	1	100	1	1	1	1	1	1	1	1	1	1	1	i	1	1	1	1	100
5	1	1	1	1	1	1	1	1	1	100	1	1	1	1	1	0	0	1	1	o	0	0	0	0	1	1	50
5	1	1	1	1	1	0	o	0	1	67	1	1	1	1	1	1	0	1	1	1	1	0	0	1	1	0	75
5	1	1	1	1	1	1	1	i	1	100	1	1	1	0	1	1	1	1	o	0	0	o	0	1	0	0	50
	1	1	1	1	1	0	0	1	0	67	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	1	100

Note: 1 = yes, 0 = no

7: internal audit

8: information systems

(a) Firms with over 60 per cent divisional functions

(b) (Centre's size/Total employees)*100

(c) (Number of divisional functions/9)*100

(d) (Number of corporate functions/16)*100

Divisional functions	
1: production 2: sales 3: marketing planning 4: personnel 5: control	6: finance 7: basic research - 8: applied research and development 9: purchasing
Corporate functions	
1: finance, accounting 2: secretarial and legal 3: personnel, human resources	9: management services 10: research and development 11: engineering services

5. personner, numan resources	TT. Engineering services
4: government, public relations	12: production, manufacturing
5: strategic/business planning	13: purchasing

- 14: property services
- 6: marketing/commercial services 15; administrative services
 - 16: others

The size of the corporate centre in Tokyo in charge of the North American market shrank, but not in proportion to the reduction brought about by the transfer of staff to USA. Over time, the corporate centre in Japan felt the need to make up for the gap created by the transfer of staff to USA, by assigning a certain number of its personnel to communication with the transferees in USA. The information inflow from USA needed to be processed and replied by Japan, which led to an increase in the number of the corporate staff in Tokyo in charge of communication with the staff in USA. On the surface, the cost of personnel has increased, but the firm itself feels benefited from this (not deliberate) arrangement in so far as it now enjoys richer information and higher information-processing capabilities concerning the North American market.

Both these examples imply that the increase in the number of divisional functions is not a result of the devolution of the centre's functions to divisions. The corporate centre in both of these cases transferred functions outside its physical presence with the aim of enriching information.

8. 2. 3. Planning and control influences

The third conclusion of this research is that Japanese firms have higher planning influence than control influence, but that, at the same time, they exert in relative terms both high planning and control influences.

Two issues related to the third conclusion

Two observations will be made on this conclusion: the close involvement of the centre in operations, and the trade-off between planning and control influences.

1) The involvement of the centre in operations

The first concerns the close involvement of the corporate centre in operations. The findings in the data analysis rejected the hypotheses that planning and control influences are associated with the size of the corporate centre, but they also maintained, as the third conclusion suggests, that the centre is closely involved in the management of divisions or functional departments. In other words, planning influence is important for Japanese firms, but so is control influence.

However, this control influence seems to be of a different nature from that of Western firms. There are two sets of circumstances which lead us to believe so. First, in Western firms, the pressure from control influence comes by way of salary and promotion prospects which depend on short-term results. In Japanese firms, on the other hand, control influence does not seem to be directly connected to these two kinds of pressure. Salary does not substantially differ from one person to another on account of short-term performance, if they have the

same hierarchical rank. Promotion prospects are not directly connected to a division's short-term performance, either.

Second, the data from the respondent firms showed a difference at the p < .0001 significance level between the incentives variable (division's performance and manager's career) and other control variables. As shown in Table 6. 13., the former's mean score was 4.49 as against the latter's 2.35. It may be deduced from this that the incentives variable is "at odds" with other variables of control influence (*i.e.*, setting objectives and monitoring results). Setting objectives and monitoring results are "tight", but career prospects are less constrained by the division's short-term performance.

It may be concluded from the above that, whereas in Western firms control influence hinges on "punitive" incentives to motivate people on the basis of "the carrot and the stick," in Japanese firms control influence is less "punitive" and less dependent on the individual's "linear" reaction to incentives, both positive (the carrot) and negative (the stick), and is more designed for the improvement of business performance.

2) Trade-off between planning and control influences

Table 7. 14. in Chapter 7 showed that there was a negative association at p < .01 level of significance between planning influence and control influence. This means that, as planning influence grew, control influence declined, and *vice versa*, though always within the high planning and high control quadrant (See Figure 7. 2.).

Lacking data, and in particular longitudinal data, it may be risky to speculate upon the reasons which lie behind this trade-off. It could be that there is a trend among Japanese firms, over time, either to lean more on planning influence and less on control influence, or to lean more on control influence and less on planning influence. Further study will be needed to throw light on this.

Recapitulating this section, the first conclusion was about the association of the size of the corporate centre and linkage variables.

The second conclusion referred to the variables which, though they did not form part of the hypotheses formulated in Chapter 5, were found to have bearing on the size of the corporate centre. These variables were diversification, divisional function numbers and corporate function numbers.

The third conclusion was about planning and control influences. It argued that Japanese firms have higher planning than control influence, but that, at the same time, Japanese firms exert in relative terms both high planning and control influences. Two observations were made on this conclusion.

8. 3. Connection of the Results of this Research with the Literature

As mentioned in the introduction to this chapter, this section will first analyse how the results of the present research relate to the principal works in the literature, including the reassessment of the alternative theoretical approaches discussed in Chapter 2 (Section 2. 3. 2.). Afterwards, an evaluation will be made of the contribution of this dissertation to the state of the art in this area of management studies.

8. 3. 1. Linkages and the size of the corporate centre

To begin with, it is interesting to relate the empirical finding of the present research, which points out that the size of the corporate centre is associated with linkages to the views of some authors on linkages.

Authors have argued as follows. Linkages exist among divisions or functional departments (Kagono, 1992) resulting from the sharing of common customers and product development (Goold and Campbell, 1987). These linkages require co-ordination. In Japanese firms coordination is horizontal (Aoki, 1988; Kagono, 1992; Nonaka, 1990), but such structure requires a centralised approach to administration (Kono, 1984; Aoki, 1988). In the present research it was found, however, that not all types of linkage are associated with the size of the corporate centre. In our sample, it was only the sharing of common customers and product development that showed positive association with the size of the corporate centre.

It was also found that the association between the size of the corporate centre and linkages becomes discernible only when the size of the firm becomes large (in the sample, over 5,000 employees)⁴.

Thus, a blind down-sizing of staff would not help firms to improve their competitive position. In carrying out process re-engineering, it is advisable to examine what linkages would be affected, what kind of information flow would be brought about, and what capability would be needed to handle the new information structure. The size of the corporate centre should be considered in the above context.

8. 3. 2. Strategic management in Japanese firms

In the second place, the results of this research link up with Goold and Campbell's study (1987) of strategic management styles. These British management researchers found that these styles represent a

⁴It implies that the size of the corporate centre is related to the total employee number. In order to determine the relationship between the information flow and the centre's size, the level of the company size (measured in its total employee number) was controlled by the dummy variables as shown in Table 7. 19.

different balance of emphasis between the fundamental tensions existing in corporate management. These tensions are (i) clear responsibilities versus multiple perspectives; (ii) detailed planning versus entrepreneurial decisions; (iii) strong leadership versus business autonomy; (iv) long-term strategic versus short-term financial objectives; and (v) tight controls versus flexible pursuit of long-term objectives. To resolve these tensions the corporate centre uses two dimensions: planning and control influences. In the sample Japanese firms of the research, there was a negative association between these two influences. When planning influence was stronger, control influence was weaker in relative terms. This negative association reflects the way Japanese firms cope with the fundamental tensions that Goold and Campbell encountered.

The results of the present research in some way contrast with Goold and Campbell's finding. As shown in the results of the test of Hypothesis 1, the average values of all the sample Japanese firms were higher on planning influence than on control influence. These results diverge from Goold and Campbell's finding that strategic management styles depend on the nature of the businesses and the resources available in the corporation as a whole. The sample firms were all located in the high-planning and high-control Quadrant (See Figure 7. 2.), which means that their strategic management style, if classified on the basis of planning and control influences, was more or less similar, even though these sample firms belonged to different industries and probably had different resource availability. The similarity of strategic management style among Japanese firms may be related to Bartlett and Ghoshal's view (1989) that Japanese firms tend to build their strategy primarily on manufacturing scale economies, to the detriment of the "conditional fit" between the key success factor in their industry and the strategic and organisational capability to achieve it. Does this mean that the manufacturing scale economies better accommodate the needs of information flow than do other strategies observed by Bartlett and Ghoshal among non-Japanese firms? Further study will be needed to address the question.

8. 3. 3. Information-processing and the corporate centre

In the third place, the results of the research concern informationprocessing and the corporate centre.

In this research empirical evidence was not found to support a relationship between the vertical information-processing needs and the size of the corporate centre, but the result should not be interpreted as showing that the centre's size has nothing to do with the need for and capabilities to process vertical information flow.

The result might rather indicate that other equally important factors such as environmental or organisational ones influence the centre's size. One possibility is that, even if firms do not have a large information flow, they might have large staff at their corporate centres for historical or competitive reasons. This might explain the overall impression that the corporate centres of Japanese firms are usually large, regardless of industry. Further study comparing Japanese and Western firms will shed light on this question.

Another possibility is that, because the sample firms were relatively large companies listed in the stock market, they tend to be more homogeneous regarding the structure and size of the corporate centre. This bias may be corrected if, in further study, small as well as large firms are taken as samples.

8. 3. 4. Considerations of sociological explanations

In the fourth place, the results of the research lead us to reconsider the alternative theoretical explanations to contingency theory.

In Chapter 2 it was contended that contingency theory is open to criticism because of its conservative assessment of the possibilities of operating socially preferred forms of organisation without incurring significant economic costs because of such aspects of organisational behaviour as political process, social constraints and dysfunctional uses of bureaucracy.

One such criticism is that organisational design is not neutral over the distribution of power and status along hierarchy (Child, 1984). Some arrangements imply a greater concentration of power and status at the top than do others. This raises the question of which organisational arrangements are acceptable to management and other groups, and

suggests that unacceptable ones will be resisted and therefore inefficient. The power to influence and/or resist policies on organisational design is of central interest to this approach. It concerns the relative power of management, workers and specialist groups within business companies. The distribution of this power will shift largely according to conditions in labour and product markets, which determine the supply and demand for employment of the specific groups which have exclusive rights to carry out certain tasks and to control training and recruitment.

On his part, Parkinson (1958; 1980) attributed to dysfunctional uses of bureaucracy the rise of the proportion of indirect to direct employees within industry. Departments grow because managers prefer to multiply subordinates not rivals. In turn, more subordinates make more work for each other. This empire building is more easily achieved in areas away from direct production, since in production a growth in numbers would often be constrained by technology and by established controls on work load, stock levels, and manning. Drake and Mitchell (1977) maintain that power dimensions are important for understanding people's behaviours in the decision process.

Based on the results of this research, however, it may be possible to qualify the argument in favour of socially preferred forms of organisations, at least in the case of Japanese manufacturing companies. Three reasons are to be adduced below. Before proceeding to them, however, it should be emphasised (1) that the argument in favour of socially preferred forms of organisations has the virtue of exposing the significance of individual actors in preserving and altering the status quo (Pennings, 1992), and (2) that, although the evidence of this research leads to an explanation in favour of contingency factor, such as information flows, the plausibility of the argument based on socially preferred forms could not be ruled out completely concerning the size of the Japanese corporate centres.

First, the size of the corporate centre in the sample Japanese firms is too large (9.1 per cent on the average) to be explained away only on the basis of political process, social constraints and dysfunctional uses of bureaucracy. (In terms of indirect cost such a large size should represent a significant weight and could be justified only if the value added of the centre compensates for it.)

It may be argued, however, that the cultural traits and traditional values of the Japanese society shape the mentality of Japanese managers in such a way that they prefer the large size of the centre.

If the relative size of the Japanese centres were not too deviant from the size of competing Western corporate centers, it could be accepted that the large size derived from cultural traits and traditional values does not run the risk of incurring significant economic costs.

However, Young's (1995) data on German and UK firms reveal that Japanese centres are 5 to 10 times as large as their German and UK counterparts).

Young's data must be qualified, since the comparison base is not uniform between Germany and UK, and Japan. However, such a large difference certainly begs more explanation than political process, social constraints and dysfunctional uses of bureaucracy.

Second, the business environment of the sample firms was found to be unstable, as mentioned in Section 7. 2. 2. In such an environment, how long and to what extent can a less effective but preferred form of organisation be sustained without incurring threats to the survival of the firm or to the incumbency of its leader? Child (1984) argues that there is scarce scope to maintain a preferred but inefficient form of organisation in a firm operating in competitive markets.

Third, the conditions under which management, or indeed any other powerful group, can sustain the organisational design it prefers constitute a contingent factor in themselves (Child, 1984).

8. 3. 5. Contribution of this research to management study

Finally, it is necessary to evaluate the contribution of this dissertation to the literature on strategic management. Its contribution may be divided into four parts.

First, although there is a growing concern about indirect costs among Japanese firms (e.g., Hori, 1993), literature on the value added by indirect cost departments does not abound. Among these indirect

costs, this study focused on the corporate centre. The study is, therefore, one of the first to attempt to offer some explanation on the value added by indirect cost departments.

The second contribution of the present dissertation is of an empirical nature. Several authors reviewed in the literature — such as Aoki, Kagono, Kono, Nonaka —, have suggested that Japanese firms are characterised by horizontal information flow among divisions or functional departments, and that such horizontal information flow relates to the centralised administration, affecting the size of the corporate centre. However, no empirical study has been conducted to test such relationship. This research is the first to do so. At the same time, as Chapter 4 synthesises this relationship between horizontal information flow and the size of the corporate centre, this research helps to fill the gap in theoretical models.

The third contribution of the present dissertation consists in shedding some light on the strategic management style of Japanese firms on the basis of two dimensions — planning and control influences. Campbell *et al* (1990) broadly classified Japanese firms as pursuing a strategic planning style, namely, high on planning but low on control, according to Goold and Campbell's work (1987) on strategic management styles. The present research modifies Campbell *et al*'s classification in that it found that Japanese firms were high on planning influence but also high on control influence.

In the fourth place, through its focus on information flow this research helps to clarify the mechanism of strategy as an emerging process. It relates to the growing awareness in the strategic management literature that strategy formation is not necessarily a consciously controlled mental process, but rather an emergent process, which is a process of learning "up and across a hierarchy" accompanied by vertical and horizontal information flows (Mintzberg, 1990a, 1990b; Ansoff, 1991). The former line of thinking corresponds to *prescriptive* orientation of schools of strategic thought, the latter to *descriptive*.

Based on either the *prescriptive* or *descriptive* orientation, ten schools of strategic thought were identified by Mintzberg (1990a and 1990b). Three are *prescriptive*, treating strategy formation as a process of conceptual design, of formal planning, and of analytical positioning. Six other schools deal with specific aspects of the process in a descriptive way. The most representative of the *descriptive* orientations is the *learning school*, which treats strategy formation as an emergent process.

Only deliberate strategies based on explicit intentions are recognised in the *prescriptive* schools. However, Mintzberg (1990a) contends that strategy formation can be deliberate or emergent. Therefore, strategists cannot always rely on procedure and technique, nor can they always design strategic perspectives in some analytically sequential or personal visionary process. According to the *learning school*, strategies emerge as strategists, sometimes individually but more often collectively, come to know a context and their organisation's capability of dealing with it. The role of leadership is not to preconceive deliberate strategies, but to manage the process of strategic learning.

Mintzberg (1990b) cites Pascale's (1988) story of Honda as a good example of emerging strategy. Honda performs strategic thinking in terms of "strategic accommodation," or "adaptive persistence," underlining its belief that corporate direction evolves from an incremental adjustment to unfolding events. Rarely does one leader (or a strategic planning group) produce a bold strategy that guides a firm unerringly. Far more frequently, the input is from below. It is in this ability of an organisation to move information and ideas from the bottom to the top and back again in continuous dialogue that the Japanese excel (Pascale, 1988). Kagono *et al*'s (1985) findings also subscribe to this view and emphasise Japanese firms' inductive and incremental reasoning approach to strategy based upon intense information flow⁵.

⁵Loveridge (1982) relates the decision-making modes or strategy schools to manpower planning, and contends that in the emerging school manpower plans may have tactical value in preparing for and concluding negotiated agreements for future changes in occupational structures.

8. 4. Implications and Prescriptions for Corporate Management

This section analyses the practical implications which the three conclusions in Section 8. 3 suggest for corporate management, and formulates, in accordance with these implications, a series of prescriptions for practising managers.

8. 4. 1. Implications for structural change

One of the implications of the research results is that, in evaluating the organisational structure, the right size of the corporate centre should not be determined in isolation. It must be decided in consideration of its relationship with the horizontal information processing needs which are derived from linkages, diversification, and the number of corporate and divisional functions. It should be borne in mind that the significance of the centre's influence on decisions is much more than its basic cost (Goold, Campbell and Alexander, 1994).

If linkages among divisions or functional departments are large, some "slack" human resources should be provided for so that the size of the corporate centre can adjust itself to information processing needs.

In the same way, diversification and the number of corporate and divisional functions ought to be taken into account in determining the size of the corporate centre.

To determine the size of the corporate centre, a methodology for measuring the information-processing needs should be developed. The present research will provide some clues to devise the methodology.

This implication relates to the following two prescriptions.

Prescription for process re-engineering

The findings of the research concerning linkages enable us to gain an insight into what is called process re-engineering (Hammer, 1990; 1993) or process innovation (Davenport, 1993). Process is a structured, measured set of activities designed to produce a specified output for a particular customer or market. In process re-engineering, business is viewed in terms of key processes rather in terms of functions, divisions, or products. Co-ordination and management of functional interdependency are achieved. Through process re-engineering, firms seek to achieve major reductions in process cost or time, or major improvements in quality.

The adoption of process re-engineering implies an absolute reduction of staff, especially at the corporate centre (Hammer, 1990; 1993). However, process re-engineering actually brings about an increase in the horizontal information flow, which, as the present research has found, may lead to a new increase in the central staff, if the structure resulting from process re-engineering is to fit in with the new information-processing needs. Likewise, such re-engineering requires strong direction from senior management, because only those in positions overlooking multiple functions can see opportunities for innovation. Therefore, in process re-engineering the number of functions at the corporate centre might increase. This might imply a need to increase its size, if the structure resulting from process re-engineering is to be accommodated to the new situation.

Prescription for lean management

The conclusions of the research suggest that firms may run a risk, if, in an effort to cut overheads, they make an indiscriminate reduction in the size of the corporate centre without prior in-depth assessment of its role. The research highlighted the fact that there was some association between the size of the corporate centre and the way it managed the firm. Linkages, diversification, and the numbers of corporate and divisional functions were discussed, tested and found to be related to the size of the corporate centre.

Therefore, an indiscriminate reduction in the size of the corporate centre might impair its capability to cope with the horizontal information flow arising from linkages, diversification, and the numbers of corporate and divisional functions. Changes in organisational design bring about changes in decision-making, information flow, and management style (Cooper, 1990). Therefore, such a reduction could

even jeopardise the competitive edge which results from the firm's information processing capability.

That down-sized firms lose their competitive edge in innovation due to the lack of horizontal information flow among different business units was researched by Dougherty and Bowman (1995). (It may not be wide of the mark to contend that down-sizing more often than not affects indirect staff units, such as the corporate centre.)

Dougherty and Bowman (1995) divide the business of innovation into three distinct activities. First, a firm must develop and perfect a design. Second, to turn a product into a commercial reality, the company must co-ordinate the work of various departments such as R&D and engineering. Finally, the firm must undertake what the study call "strategic linking" derived from horizontal information flow. If it is to succeed, a new product must become part of a firm's corporate strategy, fit into its organisational structure, and successfully compete for resources, such as cash and people.

Down-sized firms were no better or worse than their unshrinking counterpart at the first two phases of product development. But they were far worse at strategic linking. Down-sizing seems to interfere with the web of informal relationships that innovators use to win support and resources for new products, and which helps mesh innovative activities with those of the firm as a whole (Reichheld, 1996). Dougherty and Bowman (1995) argue that down-sized companies tend to lack such "entrepreneurial networking".

These findings help to explain why some companies perform well in the immediate wake of a down-sizing, but then run into difficulties later.

8. 4. 2. Implications for the interpretation of Japanese management

Following the arguments mentioned in Section 1. 3., and developed in Section 4. 6., the results of this research may contribute two insights into Japanese management: the internationalisation of Japanese firms and the transfer of Japanese lean production systems.

Prescription for internationalisation

Concerning its internationalisation process, the results suggest that Japanese firms will need Japanese managers in certain key positions of their overseas units for the handling of information flow with the head office. The reason is that a common background fostered by the ranking hierarchy and contextual skills is essential for the information-processing structure of Japanese firms. Thus, it will be difficult to replace Japanese managers with local ones in this.

Kao is a case in point. The Economist (1996a) reports that Kao has failed to incorporate its new foreign employees into the formal network that works so well for its Japanese ones. Despite its democratic pretensions, there are no foreigners on Kao's board even though half of its employees now work abroad. Most of Kao's ideas still come from its Japanese staff.

Another case is New United Motors Manufacturing (NUMMI), a joint venture between General Motors and Toyota that began operation in Fremont, California in 1984. To establish a close co-ordination of information flow between the new joint venture and the corporate centre in Japan, Toyota needed to send a core staff of between 30 and 35 managers and production co-ordinators to staff the venture (Adler, 1992). They stayed on for three to five years. The size of the Japanese expatriate core staff is large, when compared with NUMMI's total salaried employees who numbered 100.

Prescription for the transfer of Japanese lean production system

For the implementation abroad of lean production management, they suggest that the priority should be given to those aspects of production technology which do not hinge on contextual skills. They include, for example, the layout of equipment, *e.g.*, U-cells, the creation of production flow, smoothing of production, etc. The appropriateness of such priority is evidenced by the experience of Price Waterhouse, which has successfully helped to implement lean production management in Spain and UK.

After all, the internationalisation of firms is neither a simple matter of transferring managerial technologies to foreign subsidiaries, and thereby transforming "backward" economies, nor a case of firms adapting their practices to those dominant in each host country; it is rather a complex and variegated set of relationships between economies, institutions and firms (Whitley, 1992).

Last, planning and control influences are used for the close supervision of lower management by the corporate centre. In this context, control influence is used not to provide incentives to individual employees but rather to set objectives and to monitor closely the performance of divisions or functional departments. This implies that too much emphasis on individual incentives in place of the current more "egalitarian" handling of individuals' remuneration may impair the balance which exists between planning and control influences.

8. 5. Limitations of the Present Research

After pointing out in the preceding section the main contributions of the present research to the literature, it is proper to recognise its limitations. In the first place, the research focused on the corporate centre, and, therefore, did not take into account the perceptions of "line" managers in divisions or functional departments. All responses were filled in by staff departments such as corporate planning, general affairs, personnel, president's office, and public relations. Accordingly, even though they may have tried to be as objective as possible in their perceptions concerning the questionnaire items, it is impossible to be sure that bias in the responses was minimum. This is a limitation inherent in the design of the research. Unfortunately, time constraint and the work involved in contrasting views between the corporate centre, and divisions and functional departments, made such contrasting of views impracticable.

The second limitation of the research, in a way similar to the first, consists in having conducted the survey from the viewpoint of the head office in Japan, and not from the perspective of the firms as a whole, namely, the head office in Japan and overseas operations in cases where there is an international office network. In a world increasingly integrated, a survey involving the entire office network would have enabled us to appreciate the role of the corporate centre in a broader context. It is comforting to know that studies on multinational firms by other authors (Schollhammer, 1971; Ghoshal, 1986) found that the perceptions of the managers at the head office and in subsidiaries were usually quite consistent; that the difference among them was minimal.

The third limitation relates to the fact that the present research was confined to Japanese firms. Hypotheses 2 and 3 relating the size of the corporate centre to planning and control influences were rejected in the present research. On the other hand, Young (1993) reported that British firms adopting Strategic Planning style, namely, firms with high planning influence and low control influence, had a larger corporate centre than financial control firms, namely, firms with low planning influence and high control influence. Further comparative study between Japanese and Western firms may cast clarifying light on this discrepancy between Young's study and the present research.

8. 6. Suggestions for Future Research

In the light of the limitations of the present research described in the foregoing section, it is possible to make some suggestions for future studies on the subject.

It is desirable that all those questions which involve the perceptions of respondents include the responses not only of the representatives of the corporate centre but also of line managers. Despite the fact that the perceptions of the two groups of respondents are usually coincident, it is advisable to avoid possible bias in responses. The inclusion of responses from line managers may enable us to discover some interesting aspects which might not be noticed in the data gathered from the managers at the corporate centre.

Second, a more ambitious study than the present one should include the relationships between the head office of Japanese MNCs (multinational corporations) and their subsidiaries and offices in the world. As highlighted by Hedlund (1993), Gupta and Govindarajan (1991) and others, MNCs increasingly confront the need for leveraging the resources and capabilities not just of head offices, but also of their national subsidiaries on a world basis. Japanese firms are no exception. Therefore, in a world in which Japanese firms are increasingly becoming global competitors, such a study will allow us to

gather more complete information on the functioning of the Japanese corporate centre in a wider context.

Finally, similar research should be conducted comparing Japanese and Western firms, which will help us to view the characteristics of the Japanese corporate centre "in the round" in the aspects addressed in the present research — namely, the relationship of the size of the corporate centre with the horizontal as well as the vertical information flow.

Bibliography

Aaker, David A. and George S. Day. *Marketing Research*. Second ed., New York, NY: John Wiley & Sons, 1983.

Abegglen, James C. The Japanese Factory: Aspects of its Social Organization. Glencoe, IL: The Free Press, 1958.

Abegglen, James C. Management and Worker: The Japanese Solution. Tokyo: Kodansha International, 1973.

Abegglen, James C. and George Jr. Stalker. Kaisha, The Japanese Corporation. New York: Basic Books, 1985.

Abell, Derek F. Defining the Business: The Strategic Point of Strategic Planning. Englewood Cliffs, NJ: Prentice-Hall, 1980.

Abernathy, William J., Kim B. Clark, and Alan M. Kantrow. "The New Industrial Competition." *Harvard Business Review* (September-October 1981): 68-81.

Adler, Paul S. "The 'learning bureaucracy': New United Manufacturing, Inc." In *Research in Organizational Behavior*, edited by L.L. Cummings and Barry M. Staw, 111-194. London: JAI Press, 1992.

Aiken, Michael and Jerald Hage. "Organizational interdependence and intra-organizational structure." *American Sociological Review* 33 (1968): 912-930.

Aldrich, Howard E. Organizations and Environment. Englewood Cliffs, NJ: Prentice-Hall, 1979.

Allen, Louis A. "The line-staff relationship." In *Readings in Business Policy*, ed. Edmund R Gray. 374-386. New York, NY: Appleton-Century-Crofts, 1968.

Andrews, Kenneth R. "The Concept of Corporate Strategy." In *The Strategy Process: Concepts, Contexts, and Cases,* eds. James Brian Quinn, Henry Mintzberg, and Robert M James. 43-50. Revised ed., Vol. Englewood Cliffs, NJ: Prentice-Hall International, 1980.

Andrews, Kenneth R. "Director's Responsibility for corporate strategy." In *Corporate Planning—Addresses, essays, lectures,* ed. Richard G Hamermesh. 512-521. John Wiley & Sons, 1983.

Ansoff, H Igor. Corporate Strategy. Revised ed., London: Penguin Books, 1987.

Ansoff, H. Igor. "Critique of Henry Mintzberg's 'The Design School: Reconsidering the basic premises of strategic management'." *Strategic Management Journal* 12 (1991): 449-461.

Anthony, Robert N, John Dearden, and Norton M Bedford. Management Control Systems. 6th ed., Homewood, IL: Irwin, 1989.

Aoki, Masahiko, Kazuo Koike, and Iwao Nakatani. *Nihon Kigyo no Keizaigaku* (The Economics of Japanese Enterprises). Tokyo: TBS Britannica, 1986.

Aoki, Masahiko. "Horizontal vs. vertical information structure of the firm." *American Economic Review* 76 (5 1986): 971-983.

Aoki, Masahiko. Information, Incentives, and Bargaining in the Japanese Economy. Cambridge: Cambridge University Press, 1988.

Aoki, Masahiko, Kazuo Koike, and Iwao Nakatani. *Nihon Kigyo Global-ka no Kenkyu: Joho System, Kenkyu Kaihatsu, Jinzai Ikusei* (Research into the Globalisation of Japanese Enterprises: Information System, Research & Development, Human Resources). Kyoto: PHP, 1989.

Argyris, Chris. "The individual and organization: some problems of mutual adjustment." In *The Great Writings in Management and*
Organizational Behavior, edited by Louis E. Boone and Donald D. Bowen, 141-155. New York: MacGraw-Hill, 1987.

Ariga, Kenn, Giorgio Brunello, Yasushi Ohkusa, and Yoshihiko Nishiyama. "Corporate Hierarchy, Promotion, and Firm Growth: Japanese Internal Labor Market in Transition." Journal of the Japanese and International Economies 6 (1992): 440-471 (1992).

Armour, H. O. and David J. Teece. "Organisational structure and economic performance: A test of the multidivisional hypothesis." *Bell Journal of Economics* 9 (1 1978): 106-122.

Asch, David. "Strategic control: A problem looking for a solution." Long Range Planning 25 (2 1992): 105-110.

Azumi, K. "Japanese society: a sociological view." In An Introduction to Japanese Civilization, ed. A E Tiedemann. New York: Columbia University Press, 1974.

Bailey, Kenneth D. Methods of Social Research. Second ed., New York, NY: The Free Press, 1982.

Baritz, L. The Servants of Power: A History of the Use of Social Science in American Industry. Middletown, CT: Wesleyan University Press, 1960.

Barnard, Chester I. *The Functions of the Executive*. Cambridge, MA: Harvard University Press, 1938.

Bartlett, Christopher A. and Hideki Yoshihara. "New challenges for Japanese multinationals: Is organization adaptation their Achilles' heel?" *HumanResource Management*, 27 (1 1988): 19-43.

Bartlett, Christopher, A., and Sumantra Ghoshal. "Matsushita Electric Industrial (MEI) in 1987." : Harvard Business School, 1988.

Bartlett, Christopher A. and Sumantra Ghoshal. *Managing Across Borders*. London: Hutchinson Business Books, 1989.

Bartlett, Christopher A. and Sumantra Ghoshal. "Matrix management: not a structure, a frame of mind." *Harvard Business Review* (July-August 1990): 138-145.

Bell, Daniel. The Coming of Post-Industrial Society: A Venture in Social Forecasting. New York: Basic Books, 1973.

Blau, Peter M. "The hierarchy of authority in organizations." The American Journal of Sociology 73 (1968): 453-467.

Blau, Peter M. "A formal theory of differentiation in organizations." *American Sociological Review* 35 (2 1970): 201-218.

Blau, Peter M. On the Nature of Organizations. New York, NY: John Wiley & Sons, 1974.

Blinder, Alan S. "A simple note on the Japanese firm." Journal of the Japanese and International Economies 7 (1993): 238-255.

Blumenthal, Tuvia. "Some Reflections on the Japanese Motivation System." In *The Economic Analysis of the Japanese Firm*, ed. Masahiko Aoki. 413-416. Amsterdam: Elsevier, 1984.

Bourgeois, Lionel J and David R Brodwin. "Strategic implementation: Five approaches to an elusive phenomenon." *Strategic Management Journal* 5 (3 1984): 241-264.

Bower, Joseph L. "Planning and control: bottom up or top down?" *Journal of General Management* 1 (3 1974): 20-31.

Bower, Joseph L. and Yves Doz. "Strategy formulation: A social and political process." In *Strategic Management*, eds. Dan E. Schendel and Charles W. Hofer. 152-166. Boston, MA: Little, Brown, 1979.

Boyd, Brian. "Corporate linkages and organizational environment: A test of the resource dependence model." *Strategic Management Journal* 11 (1990): 419-430.

Braybrooke, D., and C. E. Lindblom. A Strategy of Decision. New York: Free Press, 1963.

Broadbent, Marianne and Danny Samson. Business and information strategy alignment: Ensuring outcomes of value to the organization. The University of Melbourne, The Graduate School of Management, 1990. Working Paper No 10,.

Brown, Paul, Virginia E. Soybel, and Clyde P. Stickney. "Comparing US and Japanese corporate-level operating performance using financial statement data." *Strategic Management Journal* 15 (1994): 75-83.

Burgelman, Robert A. "A model of the interaction of strategic behavior, corporate context, and the concept of strategy." Academy of ManagementReview 8 (1 1983): 61-70.

Burgelman, Robert A. "A process model of internal corporate venturing in the diversified major firm." Administrative Science Quarterly 28 (1983): 223-244.

Burns, Tom and Graham M. Stalker. *The Management of Innovation*. London: Tavistock, 1961.

Burns, Tom "Industry in a new age." In *Organization Theory*, ed. D. S. Pugh. 40-51. Second ed., Harmondsworth: Penguin Books, 1963.

Burns, Lawton R. and Douglas R. Wholey. "Adoption and abandonment of matrix management programs: Effects of organizational characteristics and interorganizational networks." *Academy of Management Journal* 36 (1 1993): 106-138.

Burton, Richard M. "Variety in strategic planning: An alternative to the problem solving approach." *Columbia Journal of World Business* (Winter 1984): 92-98. Cable, John and Hirohiko Yasuki. "Internal organization, business groups, and corporate performance: An empirical test of the multidivisional hypothesis in Japan." International Journal of Industrial Organization 3, (1985) (1985): 401-420.

Camillus, John C. and John H. Grant. "Operational planning: The integration of programming and budgeting." Academy of Management Review 5 (3 1980): 369-379.

Campbell, Nigel, Michael Goold, and Kimio Kase. "The role of the centre in managing large diversified companies in Japan." 49. Working paper. Manchester Business School, Ashridge Strategic Management Centre, 1990.

Carlson, Howard C. "The parallel organization structure at General Motors." *Personnel* 55 (4 1978): 64-69.

Carney, T. F. "Organizational communication: Emerging trends, problems, and opportunities." In Organization, Communication: Emerging Perspective I, ed. Lee Thayer. Norwood, N. J.: Ablex Publishing Corporation, 1986.

Caves, Richard E. American Industry: Structure, Conduct, Performance. Fourth ed., Englewood Cliffs, NJ: Prentice-Hall, Inc., 1977.

Chakravarthy, Balaji and Peter Lorange. Strategic adaptation in multi-business firms. INSEAD, 1989. Working Paper No. 89/44.

Chalmers, A. F. What is This Thing called Science. Milton Keynes: Open University Press, 1978.

Chandler, Alfred D. Strategy and Structure: Chapters in the History of the Industrial Enterprise. Cambridge, MA: The MIT Press, 1962.

Chandler, Alfred D. "The functions of the HQ unit in the multibusiness firm." Strategic Management Journal 12 (1991): 31-50.

Chase, Richard, B., and Eric L. Prentis. "Operations management: a field rediscovered." *Journal of Management* 13, no. 2 (1987): 351-366.

Child, John. "Organizational structure, environment and performance: the role of strategic choice." Sociology 6 (1 1972): 1-22.

Child, John. "Predicting and understanding organization structure." Administrative Science Quarterly 18 (1973a): 168-185.

Child, John. "Organization: a choice for man." In Man and Organization: The Search for Explanation and Social Relevance, ed. John Child. 234-257. London: George Allen & Unwin, 1973b.

Child, John. "Parkinson's progress: accounting for the number of specialists in organizations." *Academic Science Quarterly* 18, no. 3 (1973c): 328-348.

Child, John. Organization: A Guide to Problems and Practice. London: Harper & Row, 1984.

Churchman, C. W., R. L. Ackoff, and E. L. Arnoff. Introduction to Operations Research. New York: Wiley, 1957.

Clark, Rodney. The Japanese Company. New Haven: Yale University Press, 1979.

Clark, Kim B. and Takahiro Fujimoto. "Product development and competitiveness." *Journal of the Japanese and International Economies* (6 1992): 101-143.

Cohen, Michael D., James G. March, and Johan P. Olsen. "A garbage can model of organisational choice." *Administrative Science Quarterly* (17 1972): 1-25.

Cole, Robert E. Japanese Blue Collar. Berkley, CA: University of California Press, 1971.

Cole, Robert E. "Functional alternatives and economic development: An empirical example of permanent employment in Japan." *American Sociological Review* (38 1973): 424-438.

Cole, Robert E. "Learning from the Japanese: Prospects and pitfalls." *ManagementReview* 69 (9 1980): 22-28, 38-43.

Cooper, Arnold and Albert V. Bruno. "Success among high-technology firms." *Business Horizon* 20 (2 1977): 16-22.

Cooper, Michael R. "Managing corporate culture to achieve growth and renewal." In *Corporate Restructuring*, edited by Milton L. Rock and Robert H. Rock, 237-259. New York: McGraw-Hill, 1990.

Cox, Eli P. "The Optimal Number of Response Alternatives for a Scale: A Review." *Journal of Marketing Research* 17 (November 1980): 407-422.

Cresap. The Effective Head Office: Data Report. London: 1988.

Cummings, L. L., G. P. Huber, and E. Arendt. "Effects of team size and spatial arrangements on group decision making." Academy of Management Journal (17 1974): 460-475.

Daft, Richard L. and Karl E. Weick. "Toward a model of organization as interpretation systems." *Academy of Management Review* 9 (2 1984): 284-295.

Daft, Richard L. and Robert H. Lengel. "Organizational information requirements, media richness and structural design." *Management Science* 32 (5 1986): 554-571.

Daft, Richard L., Juhani Sormunen, and Don Parks. "Chief executive scanning, environmental characteristics, and company performance: An empirical study." *Strategic Management Journal* 9 (2 1988): 96-104. Daley, Lane, James Jiambalvo, Gary L. Sunden, and Yasumasa Kondo. "Attitudes toward financial control systems in the United States and Japan." *Journal of International Business Studies* (Fall 1985): 91-110.

Daniel, Shirly and Wolf D. Reitsperger. "Management control systems for JIT: An empirical comparison of Japan and the US." *Journal of International Business Studies* (Fourth Quarter 1991): 603-617.

Daniel, A. Lynn. "Strategic planning—The role of the chief executive." Long Range Planning 25 (2 1992): 97-104.

Davenport, Thomas H. Process Innovation: Reengineering Work through Information Technology. Boston, MA: Harvard Business School Press, 1993.

Davis, Stanley M. and Paul R. Lawrence. *Matrix*. Reading, MA: Addison-Wesley Publishing Company, 1977.

de Vaus, D. A. Surveys in Social Research. second ed., London: Unwin Hyman, 1990.

Deshpandé, Rohit, John U. Farley, and Frederick E. Jr. Webster. "Corporate culture, customer orientation, and innovativeness in Japanese Firms: A quadrad analysis." *Journal of Marketing* 57 (January 1993): 23-27.

Dess, Gregory G. and D. Beard. "Dimensions of organizational task environments." *Administrative Science Quarterly* 29 (1984): 52-73.

Dess, Gregory G. "Consensus on strategy formulation and organizational performance: competitors in a fragmented industry." *Strategic Management Journal* 8 (1987): 259-277.

DiMaggio, Paul J. and Walter W. Powell. "The iron cage revisited: Institutional isomorphism and collective rationality in organizational fields." American Sociological Review 48 (April 1983): 160.

Doi, Takeo. *The Anatomy of Dependence*. New York, NY: Kodansha International, 1981.

Dore, Ronald. British Factory Japanese Factory: The Origins of National Diversity in Industrial Relation. Berkeley, CA: University of California Press, 1973.

Dougherty, Deborah, and Edward H. Bowman. "The effects of organizational downsizing on product innovation." *California ManagementReview* 37, no. 4 (1995): 28-44.

Doz, Yves L. and C. K. Prahalad. "A process model of strategic redirection in large complex firms: the case of multinational corporations." In *The Management of Strategic Change*, ed. Pettigrew Andrew M. 63-83. Oxford: Basil Blackwell, 1987.

Drake, Bruce, and Terence Mitchell. "The effects of vertical and horizontal power on individual motivation and satisfaction." Academy of Management Journal 20, no. 4 (1977): 573-591.

Dubin, R. "Business behaviour behaviourally viewed." In Social Science Approach to Business Behaviour, ed. G. B. Strother. London: Tavistock, 1962.

Duncan, Robert B. "Characteristics of organizational environments." Administrative Science Quarterly 17 (1972): 313-327.

Duncan, Robert B. "Multiple decision-making structures in adapting to environmental uncertainty: The impact of organizational effectiveness." *Human Relations* 26 (1973): 273-291.

Dunphy, Dexter. "Convergence/divergence: A temporal review of the Japanese enterprise and its management." Academy of Management Review 12 (3 1987): 445-459.

Dutton, Jane E. and Susan J. Ashford. "Selling issues to top management." Academy of Management Review 18 (3 1993): 397-428.

Dyer, Jeffrey H. and William G. Ouchi. "Japanese-style partnerships: Giving companies a competitive edge." *Sloan ManagementReview* (Fall 1993): 51-63.

Economist, The. "Wheels within wheels.", March 12th 1988, 72-75.

Economist, The. "They lost their shirt." November 19th 1994, 88.

Economist, The. "Japanese consumer goods: should we kow-tow to Kao?", March 30th 1996a, 60-61.

Economist, The. "Fire and forget?", April 20th 1996b, 51-52.

Economist, The. "How Japan remembers.", April 20th 1996c, 52.

Eisenhardt, Kathleen M. "Agency theory: An assessment and review." Academy of Management Review 14 (1 1989): 57-74.

Emery, F. E. and E. L. Trist. "The causal texture of organizational environments." *Human Relations* 18 (1965): 21-32.

Engelhoff, William G. "Strategy and structure in multinational corporations: An information-processing approach." *Administrative Science Quarterly* 27 (1982): 435-458.

Etzioni, Amitai. A Comparative Analysis of Complex Organization: On Power, Involvement, and Their Correlates. New York: The Free Press, 1961.

Etzioni, Amitai. Modern Organizations. Englewood Cliffs, NJ: Prentice-Hall Inc, 1964.

Etzioni, Amitai. "Choose America must—between 'reindustrialization' and 'quality of life'." Across the Board 17, no. 10 (1980): 42-49. Fayol, Henri. General and Industrial Management. London: Pitman, 1949.

Fiedler, Fred E. A Theory of Leadership Effectiveness. New York: McGraw-Hill, 1967.

Fikelstein, Sydney and Donald C. Hambrick. "Chief executive compensation: A study of the intersection of markets and political processes." *Strategic Management Journal* 10 (1989): 121-134.

Filley, A., and R. House. Managerial Process and Organizational Behavior. Glenview: Scott-Foresman, 1969.

Fleishman, E. A. and E. F. Harris. "Patterns of leadership behavior related to employee grievance and turnover." *Personnel Psychology* 15 (1962): 43-56.

Forrester, J. W. "Counterintuitive behavior of social systems." *Technology Review* 73, no. January (1971): 52-68.

Fruin, W. Mark. "The family as a firm and the firm as a family in Japan: The case of Kikkoman Shoyu Company Limited." *Journal of Family History* (5 1980): 432-449.

Fruin, W. Mark. The Japanese Enterprise System: Competitive Strategies and Cooperative Structures. Oxford: Oxford University Press, 1992.

Galbraith, Jay R. Designing Complex Organizations. Reading, MA: Addison Wesley, 1973.

Galbraith, Jay R. "Organization Design: An information processing view." Interfaces (May 1974): 28-36.

Galbraith, Jay R. Organization Design. Reading, MA: Addison-Wesley, 1977.

Galbraith, Jay R. and Robert K. Kazanjian. Strategy Implementation: Structure, Systems, and Process. St Paul, MN: West Publishing Company, 1978.

Galbraith, Jay R. "Strategy and organization planning." In *The Strategy Process*, eds. James Brian Quinn, Henry Mintzberg, and Robert M. James. 304-311. Englewood Cliffs, NJ: Prentice-Hall International, 1983.

Galbraith, Jay R. and Robert K. Kazanjian. Strategy Implementation: Structure, System, and Process. St. Paul, MN: West Publishing Company, 1986.

Gerlach, Michael. "Business alliances and the strategy of the Japanese firm." *California Management Review* (Fall 1987): 126-142.

Gerlach, Michael L. Alliance Capitalism: The Social Organization of Japanese Business. Berkley, CA: University of California Press, 1992.

Ghoshal, Sumantra. "The Innovative Multinational: A Differentiated Network of Organizational Roles and Management Processes." Unpublished doctoral dissertation, Harvard Business School, 1986.

Ghoshal, Sumantra and Nitin Nohria. Requisite variety versus shared values: Managing corporate-division relationships in the M-form organization. INSEAD, 1989. Working Paper No 89/34.

Ghoshal, Sumantra, Harry Korine, and Gabriel Szulanski. "Interunit communication in multinational corporations." *Management Science* 40 (1 1994): 96-109.

Giglioni, G B, and A G Bedeian. "A conspectus of management control theory: 1900–1972." Academy of Management Journal 17 (2 1974): 292-305.

Gilbreth, Frank. Primer of Scientific Management. New York: Harper & Brothers, 1912.

Gioia, Dennis A., and Evelyn Pitre. "Multiparadigm perspectives on theory building." *Academy of Management Review* 15 (1990): 584-602.

Gluck, Frederick W., Stephen P. Kaufman, and A. Steven Walleck. "Strategic management for competitive advantage." *Harvard Business Review* (July-August 1980): 154-161.

Glueck, William F. Business Policy and Strategic Management. Third ed., Tokyo: Mc-Graw Hill Kodansha, 1980.

Goold, Michael and Andrew Campbell. Strategies and Styles: The Role of the Centre in Managing Diversified Corporations. Oxford: Basil Blackwell, 1987.

Goold, Michael, Andrew Campbell, and Marcus Alexander. Corporate-Level Strategy: Creating Value in the Multibusiness Company. New York: John Wiley, 1994.

Goold, Michael and John J. Quinn. Strategic Control: Milestones for Long-Term Performance. London: The Economist Books, 1990.

Goold, Michael, Andrew Campbell, and Kathleen Luchs. "Strategies and styles revisited: Strategic planning and financial control." *Long Range Planning* 26 (5 1993): 49-60.

Gordon, David M. Fat and Mean: the Corporate Squeeze of Working Americans and the Myth of Managerial "Downsizing". New York: The Free Press, 1996.

Goto, A. "Business groups in a market economy." *EuropeanEconomic Review* 19 (1 1982): 53-70.

Green, Paul E. and Donald S. Tull. *Research for Marketing Decisions*. 4th ed., Englewood Cliffs, NJ: Prentice-Hall, 1978.

Greenwood, Ronald G., and Charles D. Wrege. "The Hawthorne Studies." In Papers Dedicated to the Development of Modern Management, edited by Daniel A. Wren, 24-35: Academy of Management, 1986.

Gresov, Christopher and Carroll Stephens. "The context of interunit influence attempts." *Administrative Science Quarterly* 38 (1993): 252-276.

Gupta, Anil K. and V. Govindarajan. "Business unit strategy, managerial characteristics, and business unit effectiveness at strategy implementation." *Academy of Management Journal* 27 (1 1984): 25-41.

Gupta, Anil K. and V. Govindarajan. "Knowledge flows and the structure of control within multinational corporations." Academy of Management Review 16 (4 1991): 768-792.

Hage, J. and M. Aiken. "The relationship of centralization to other structural properties"." *Administrative Science Quarterly* (14 1967): 366-376.

Haleblian, Jerayr and Sydney Finkelstein. "Top management team size, CEO dominance, and firm performance: The moderating roles of environmental turbulence and discretion." *Academy of Management Journal* 36 (4 1993): 844-863.

Hall, R. H. Organizations: Structure and Process. 2nd ed., Englewood Cliffs, NJ: Prentice-Hall, 1977.

Hambrick, Donald C. and Phyllis A. Mason. "Upper echelons: the organization as a reflection of its top management." Academy of ManagementReview 9 (2 1984): 193-206.

Hambrick, D. C. and S. Finkelstein. "Managerial discretion: A bridge between polar views on organizations." In *Research in Organizational Behavior*, eds. L. L. Cummings and B. M. Staw. 369-406. 9. Greenwich, CT: JAI Press, 1987.

Hambrick, Donald C. "The adolescence of strategic management, 1980-1985: Critical perceptions and reality." In *Perspectives on Strategic Management*, ed. James W. Fredrickson. 237-261. New York, NY: Harpers & Row Publishers, 1990.

Hambrick, Donald C. and R. A. D'Aveni. "Top team deterioration as part of the downward spiral of large corporate bankruptcies." *Management Science* (38 1992): 1445-1566.

Hamel, Gary and C. K. Prahalad. "Strategic intent." Harvard Business Review (May-June 1989): 63-76.

Hammer, Michael. "Reengineering work: Don't automate, obliterate." *Harvard Business Review* (July-August 1990): 104-112.

Hammer, Michael, and James Champy. Reengineering the Corporation: A Manifesto for Business Revolution. New York: Harper Collins, 1993.

Hanada, Mitsuyo. "Nihon no jinji seido ni okeru kyoso genri-Shoshin, shokaku system no jittai (Competition Principles in the Personnel System of Japan-Realities of personnel promotions)." In *Japanese Enterprise System*, eds. Hiroyuki Itami, Tadao Kagono, and Motoshige Itoh. 276-299. 3. Tokyo: Yuhikaku, 1993.

Hanami, T. Labor Relations in Japan Today. Tokyo: Kodansha International, 1979.

Hannan, M. T., and J. Freeman. "The population ecology of organizations." *American Journal of Sociology*, no. 52 (1977): 929-964.

Harrison, E. Frank. *The Managerial Decision-Making Process*. Boston, MA: Houghton Mifflin, 1975.

Harrison, E. Frank. "Strategic control at the CEO level." Long Range Planning 24 (6 1991): 78-87.

Hart, Stuart L. "An integrative framework for strategy-making processes." Academy of Management Review 17 (2 1992): 327-351.

Haspeslagh, Philippe. "Portfolio planning: uses and limits." Harvard Business Review (January-February 1982): 58-73.

Hatvany, Nina and Vladimir Pucik. "An integrated management system: Lessons from the Japanese experience." Academy of Management Review (6 1981): 469-480.

Hax, Arnoldo C. and Nicolás S. Majluf. The Strategy Concept and Process: A Pragmatic Approach. Englewood Cliffs, NJ: Prentice-Hall, 1991.

Hayashi, Kichiro. "Corporate planning practices in Japanese multinationals." Academy of Management Journal 21 (2 1978): 211-226.

Hayes, Robert H. and William Abernathy. "Managing our way to economic decline." *Harvard Business Review* (July-August 1980): 67-77.

Hayes, Robert H. "Why Japanese factories work." *Harvard Business Review* (November-December 1981): 57-66.

Hayes, Robert H. and Steven C. Wheelwright. *Restoring our Competitive Edge*. New York, NY: John Wiley and Sons, 1984.

Hedlund, G. "Assumptions of hierarchy and heterarchy — with applications to the management of the multinational corporation." In *Organization Theory and the Multinational Corporation*, eds. S. Ghoshal and E. Westney. London: Macmillan, 1993.

Heenan, David A. and Howard V. Perlmutter. *Multinational* OrganizationDevelopment. Reading, MA: Addison-Wesley, 1979.

Herzberg, Frederick. "One more time: how do you motivate employees?" In *The Great Writings in Management and Organizational Behavior*, edited by Louis E. Boone and Donald D. Bowen, 169-183. New York: MacGraw-Hill, 1987. Hickson, D. J. "A convergence in organization theory." Administrative Science Quarterly (12 1968): 224-237.

Hill, G. W. "Group versus individual performance: Are n+1 heads better than one?" *Psychological Bulletin* (91 1982): 517-539.

Hill, Richard. "The coming of post-industrial society." The Insurgent Sociologist, no. 4 (1974): 37-51.

Hofer, Charles W. and Dan Schendel. *Strategy Formulation: Analytical Concepts.* St. Paul, MN: West Publishing, 1978.

Hofstede, Geert. "Motivation, leadership, and organization: Do American theories apply abroad?" *Organizational Dynamics* (Summer 1980): 42-63.

Hofstede, Geert. Culture's Consequence: International Differences in Work Related Values. London: Sage Publications, 1980.

Holden, Nigel and Matt Burgess. Japanese-Led Companies: Understanding How to Make Them Your Customers. Berkshire: McGraw-Hill, 1994.

Honma, Masaaki. "Watakushino shin shihonshugiron—Nihon no kigyo shakai, 'Naibu shijo' ga hidaika (My new approach to the capitalism—Japan's enterprise society has too much fat on 'its internal market')." *Nihon Keizai*, July 7th 1992,

Hori, Shintaro. "Fixing Japan's white-collar economy: A personal view." *Harvard Business Review* (November-December 1993): 157-172.

Howard, Nigel and Yoshiya Teramoto. "The really important difference between Japanese and Western management." *Management International Review* 21 (3 1981): 19-30.

Hrebiniak, Lawrence G. "The organization and environment research program: overview and critique." In *Perspectives on Organization* Design and Behavior, eds. Andrew H. Van de Ven and William F. Joyce. 338-346. New York, NY: John Wiley & Sons, 1981.

Hrebiniak, Lawrence G. and William F. Joyce. "The strategic importance of managing myopia." *Sloan Management Review* (Fall 1986): 5-13.

Hungenberg, Harald. "How to ensure that headquarters add value." Long Range Planning 26 (6 1993): 62-73.

Ibarra, Hermina. "Network centrality, power, and innovation involvement: determinants of technical and administrative roles." *Academy of Management Journal* 36 (3 1993): 471-501.

Imanishi, Shinji. *Jigyobusei no Kaimei—Kigyo Seicho to Keiei Soshiki* (Analysis of the Divisional Organisation—Company Growth and Management System). Tokyo: Management-sha, 1988.

Itami, Hiroyuki. *Mobilizing Invisible Assets*. Cambridge, MA: Harvard University Press, 1987.

Itoh, Hideshi. "Information processing capacities of the firm." *Journal of the Japanese and International Economies* (1 1987): 299-326.

Jaeger, Alfred M. and B. R. Baliga. "Control systems and strategic adaptation: lessons from the Japanese experience." *Strategic Management Journal* 6 (1985): 115-134.

James, J. "A preliminary study of the size determinant in small group interaction." *American Sociological Review* (16 1951): 474-477.

Japan Development Bank. *Keiei Senryaku to Soshiki—Jigyobusei wo Chushin toshite* (Management Strategy and Organisation—with the Emphasis on the Divisionalisation). Japan Development Bank, 1989.

Johnson, Gerry. The application of grounded theory to a study of corporate growth. The University of Aston Management Centre, 1981. Working Paper Series No. 212. Johnson, Gerry. "Managing strategic change—Strategy, culture and action." *Long Range Planning* 25 (1 1992): 28-36.

Johnson, Chalmers. "Comparative capitalism: the Japanese difference." *California Management Review* (Summer 1993): 51-67.

Juran, J. M. "Japanese and Western quality: a contrast in methods and results." *ManagementReview* (November 1978): 27-45.

Kagono, Tadao , Ikujiro Nonaka, Akihiro Okumura, Kiyonori Sakakibara, Yoichi Komatsu, and Akinobu Sakashita. "Mechanistic vs organic management systems: A comparative study of adaptive patterns of US and Japanese firms." *The Annals of the School of Business Administration*, Kobe University 25 (1981): 115-145.

Kagono, Tadao, Ikujiro Nonaka, Kiyonori Sakakibara, and Akihiro Okumura. Evolutionary Management: A US-Japan Comparison of Strategy and Organization. Amsterdam: Elsevier Science Publishers B. V., 1985.

Kagono, Tadao. Soshiki Ninshikiron—Kigyo ni okeru Sozo to Kakushin no Kenkyu (Organisational Cognition Theory—Research into the Creativity and Innovation in Enterprises). Tokyo: Chikura Shobo, 1988.

Kagono, Tadao. "Nihon kigyo no jigyobusei (The divisional organisation of Japanese enterprises)." *Nihon Keizai*, July 11th—17th 1992,

Kamm, Judith B. An Integrative Approach to Managing Innovation. Lexington, MA: Lexington Books, 1987.

Kanai, Toshihiro. Henkakugata Middle no Kenkyu—Senryaku Kakushingata no Kanrisha Kodo (In Search of Transformational Middle—A Strategy and Innovation Driven Managerial Behaviour). Tokyo: Hakuto Shobo, 1991. Kanai, Toshihiro. "Soshiki no joho shori paradigm (Information processing paradigm of organisations)." In *Nihon no Kigyo System* (Japanese Enterprise System), eds. Hiroyuki Itami, Tadao Kagono, and Motoshige Itoh. 353-378. 1. Tokyo: Yuhikaku, 1993.

Kanter, Rosabeth Moss. The Change Masters: Corporate Entrepreneurs at Work. London: Unwin Hyman, 1983.

Kanter, Rosabeth Moss. When Giants Learn to Dance: Mastering the Challenge of Strategy, Management, and Careers in the 1990s. Routelage, 1990.

Kast, F. E., and J. E. Rosenzweig. "General systems theory: applications for organization and management." Academy of Management Journal, no. December (1972): 447-465.

Kawai, Tadahiko. "Generating innovation through strategic action programmes." *Long Range Planning* 25 (3 1992): 36-42.

Keats, Barbara W. and Michael A. Hitt. "A causal model of linkages among environmental dimensions, macro organizational characteristics, and performance." *Academy of Management Journal* 31 (3 1988): 570-598.

Keys, J. Bernard and Thomas R. Miller. "The Japanese management theory jungle." Academy of Management Review 9 (2 1984): 342-353.

Khandwalla, Pradip N. "Mass output orientation of operations technology and organizational structure." *Administrative Science Quarterly* 19 (1974): 74-98.

Koike, Kazuo. "Skill Formation System in the US and Japan: A Comparative Study." In *The Economic Analysis of the Japanese Firm*, ed. Masahiko Aoki. 47-75. Amsterdam: Elsevier Science Publishers, 1984.

Kono, Toyohiro. *Strategy and Structure of Japanese Enterprises*. London: The Macmillan Press, 1984. Kono, Toyohiro. *Choki Keikaku no Jitsurei* (Examples of Long-term Planning). Tokyo: Dobunkan, 1986.

Kono, Toyohiro. "Corporate culture and long-range planning." Long Range Planning 23 (4 1990): 9-19.

Koontz, Harold. "The management theory jungle." Academy of Management Journal, no. 4 (1961): 174-178.

Kotler, Philip, Liam Fahey, and Somkid Jatusripitak. *New Competition*. Englewood Cliffs, NJ: Prentice-Hall Inc., 1984.

Kotter, John P. The General Managers. New York, NY: The Free Press, 1982.

Lauenstein, Milton. "Strategy planning in Japan." The Journal of Business Strategy 2 (6 1985): 78-86.

Lawrence, Paul R. and Jay W. Lorsch. Organization and Environment: Managing Differentiation and Integration. Cambridge, MA: Harvard University Press, 1967.

Leibenstein, Harvey. "The Japanese management system: An Xefficiency-game theory analysis." In *The Economic Analysis of the Japanese Firm*, ed. Masahiko Aoki. 331-358. Amsterdam: Elsevier Science Publishers, 1984.

Leighton, Alexander H. "The functional point of view." In *Behavior in Organizations: A Diagnostic Approach*, edited by Jean Bartunek and Judith Gordon, 13-15. Lexington, MA: Xerox, 1978.

Leontiades, Milton. "The Japanese art of managing diversity." *The Journal of Business Strategy* (March/April 1991): 30-36.

Levine, S. B. and S. B. Kawada. Human Resources in Japanese Industrial Development. Princeton, NJ: Princeton University Press, 1980.

Likert, Rensis. "New patterns of management." In Organization Theory, ed. D. S. Pugh. 293-316. Harmondsworth, Middlesex: Penguin Books, 1961.

Likert, Rensis. "An integrating principle and an overview." In The Great Writings in *Management and Organizational Behavior*, edited by Louis E. Boone and Donald D. Bowen, 216-236. New York: MacGraw-Hill, 1987.

Lincoln, J. R. "Japanese organization and organization theory." *Research in Organisational Theory* 12:255-294.

Lincoln, James R., Jon Olson, and Mitsuyo Hanada. "Cultural effects on organizational structure: The case of Japanese firms in the United States." *American Sociological Review* 43 (December 1978): 829-847.

Lincoln, James R., Mitsuyo Hanada, and Kerry McBride. "Organizational structures in Japanese and U.S. manufacturing." *Administrative Science Quarterly* 31 (1986): 338-364.

Lincoln, James R. and Arne L. Kalleberg. Culture, Control, and Commitment: A Study of Work Organization and Work Attitudes in the United States and Japan. Cambridge: Cambridge University Press, 1990.

Lindblom, C. E. "The science of "muddling through"." In *Business Strategy*, ed. H. Igor Ansoff. 41-60. Middlesex: Penguin Books, 1959.

Lorenz, Christopher. "El mito de la central, siempre menguante: Recortar no siempre es solución." *Expansión*, June 18th 1993, 38.

Lorsch, Jay W. and Stephen A. Allen. Managing Diversity and Interdependence: An Organizational Study of Multidivisional Firms. Boston, MA: Harvard University Graduate Business Administration Division of Research, 1973. Lorsch, Jay W. and Haruo Takagi. "Keeping managers off the shelf." *Harvard Business Review* (July-August 1986): 60-65.

Loveridge, Ray. Contingency, control and risk: the utility of manpower planning in a risky environment. The University of Aston Management Centre, 1982. Working Paper Series Number 240.

Loveridge, Ray. "Sources of diversity in internal markets." *Sociology* 17 (1 1983): 44-64.

Lupton, T. Management and the Social Science. Harmondsworth: Penguin Books, 1971.

McClelland, David C. "That urge to achieve." In *The Great Writings in Management and Organizational Behavior*, edited by Louis E. Boone and Donald D. Bowen, 384-395. New York: McGraw-Hill, 1987.

McGregor, Douglas M. "The human side of enterprise." In *The Great Writings in Management and Organizational Behavior*, edited by Louis E. Boone and Donald D. Bowen, 126-136. New York: McGraw-Hill, 1987.

McMillan, Charles J. The Japanese Industrial System. Second ed., Berlin: Walter de Gruyter, 1985.

Mahoney, Joseph T. and J. Rajendran Pandian. "The resource-based view within the conversation of strategic management." *Strategic ManagementJournal* 13 (1992): 363-380.

Mann, Leon and Chris Ball. Information processing and decision making. The University of Melbourne, the Graduate School of Management, 1992. 13.

March, James G. "Decision making perspective: Decisions in organizations and theories of choice." In *Perspectives on Organization Design and Behavior*, eds. H. Van de Ven and William F. Joice. 205-244. New York, NY: John Wiley & Sons, 1981. Marsh, Robert M. and Hiroshi Mannari. "Technology and size as determinants of the organizational structures of Japanese factories." *Administrative Science Quarterly* 26 (March 1981): 33-57.

Marsh, Robert M. "A research note: centralization of decisionmaking in Japanese factories." *Organization Studies* 13 (2 1992): 261-274.

Marsland, S. and M. Beer. "The evolution of Japanese management: Lessons for US managers." *Organizational Dynamics* 11 (3 1983): 49-67.

Maslow, Abraham H. "A theory of human motivation." In *The Great Writings in Management and Organizational Behavior*, edited by Louis E. Boone and Donald D. Bowen, 106-119. New York: McGraw-Hill, 1987.

Masuda, Y. The Information Society as Post-Industrial Society. Washington, D. C.: World Future Society, 1981.

Matsumoto, Koji. Organizing for Higher Productivity: An Analysis of Japanese Systems and Practices. Tokyo: Asian Productivity Organization, 1982.

Mayo, Elton. The Human Problems of an Industrial Civilization. Salem, N. H.: Ayer Company, 1977.

Mayo, Elton. "Hawthorne and the Western Electric Company." In *The Great Writings in Management and Organizational Behavior*, edited by Louis E. Boone and Donald D. Bowen, 77-89. New York: McGraw-Hill, 1987.

Merchant, Kenneth A. "The control function of management." *Sloan ManagementReview* (Summer 1982): 43-55.

Meyer, Marshall W. "Size and the structure of organizations: a causal analysis." *American Sociological Review* 37 (August 1972): 434-441.

Meyer, Glenn E. SPSS: A Minimalist Approach. Fort Worth: Harcourt Brace Jovanovich College Publishers, 1993.

Michel, J. G. and D. C. Hambrick. "Diversification posture and top management team characteristics." Academy of Management Journal 35 (1992): 9-37.

Miles, Raymond E. Theories of Management: Implications for Organizational Behavior and Development. New York: McGraw-Hill, 1975.

Miles, Raymond and Charles C. Snow. *Environmental Strategy and Organization Structure*. New York: McGraw-Hill, 1978.

Miles, Raymond, Charles C. Snow, Alan D. Meyer, and Henry J. Coleman. "Organizational strategy, structure, and process, Academy of Management Review, July 1978, in." In *The Strategy Process: Concepts, Contexts, and Cases*, eds. James Brian Quinn, Henry Mintzberg, and Robert M. James. 524-530. Englewood Cliffs, NJ: Prentice-Hall International, 1983.

Miller, Danny. "Toward a new contingency approach: the search for organizational gestalts." *Journal of Management Studies* 18, no. 1 (1981): 1-26.

Miller, E. J. and A. K. Rice. System of Organization. London: Tavistock, 1967.

Millner, Edward L., and George E. Hoffer. "Do Japanese auto manufactureres behave as short-run profit maximizers in the US market?" *Applied Economics*, no. 28 (1996): 157-162.

Mink, Oscar G., James M. Shultz, and Barbara P. Mink. *Developing & Managing Open Organizations: A Model and Methods for Maximizing Organizational Potential*. Austin, Texas: Learning Concepts, 1979.

Mintzberg, Henry. "Strategy making in three modes." california ManagementReview 16, no. 4 (1973): 44-58.

Mintzberg, Henry and James A. Waters. "Of strategies, deliberate and emergent." *Strategic Management Journal* 6 (1985): 257-272.

Mintzberg, Henry. "Strategy formation: Schools of thoughts." In Perspectives on Strategic Management, ed. James W. Fredrickson. 105-236. New York: Harper & Row, 1990a.

Mintzberg, Henry. "The design school: Reconsidering the basic premises of strategic management." *Strategic Management Journal* 11 (1990b): 171-195.

Mintzberg, Henry. "The fall and rise of strategic planning." *Harvard Business Review* (January-February 1994): 107-114.

Mintzberg, Henry, Duru Raisinghani, and André Théorêt. "The structure of 'unstructured' decision processes." *Administrative Science Quarterly* 21 (1976): 246-275.

Mintzberg, Henry and Alexandra McHugh. "Strategy formation in an adhocracy." Administrative Science Quarterly 30 (1985): 160-197.

Misumi, Jyuji. "Decision-making in Japanese groups and organizations." In International Perspectives on Organizational Democracy, eds. B. Wilpert and A. Sorge. 525-539. John Wiley, 1984.

Monden, Yasuhiro. Toyota Production System. Atlanta, GA: Industrial Engineering and Management Press, Institute of Industrial Engineers, 1983.

Monden, Yasuhiro. "Japanese management control systems." In Innovations in Management: The Japanese Corporation, eds. Yasuhiro Monden, Rinya Shibakawa, Satoru Takayanagi, and Teruya Nagao. 41-58. Atlanta, GA: Industrial Engineering and Management Press, 1985. Munchus, G. "Employer-employee based quality circles in Japan: Human resource policy implications for American firms." Academy of ManagementReview 8 (1983): 255-261.

Murray, John A. "A concept of entrepreneurial strategy." *Strategic Management Journal* 5 (1 1984): 1-13.

Nadler, David, and Michael Tushman. "A diagnostic model for organizational behavior." In *Perspectives on Behavior in Organizations*, edited by J. R. Hackman, 85-101. New York: McGraw-Hill, 1977.

Nadler, David, and Micael Tushman. "A model for diagnosing organizational behavior." *Organizational Dynamics* Autumn (1980): 35-51.

Nagao, Teruya. "Japanese organizational behavior." In *Innovations in Management: The Japanese Corporation*, eds. Yasuhiro Monden, Rinya Shibakawa, Satoru Takayanagi, and Teruya Nagao. 23-40. Atlanta, GA: Industrial Engineering and Management Press, 1985.

Nakahara, Toshio and Yutaka Isono. "Strategic planning for Canon: The crisis and the new vision." Long Range Planning 25 (1 1992): 63-72.

Nakane, Chie. Japanese Society. Harmondsworth, Middlesex: Penguin Books, 1973.

Nippon Keizai Shimbun. Nikkei Kaisha Joho ('93-IV). Tokyo: Nippon Keizai Shimbun, 1993.

Noda, Kazuo. "Big business organization." In *Modern Japanese Organization and Decision-Making*, ed. Ezra F. Vogel. 115-145. Tokyo: Charles E. Tuttle, 1979.

Nonaka, Ikujiro. "Self-renewal of the Japanese firm and the human resource strategy." *Human Resource Management* 27 (1 1988): 45-62.

Nonaka, Ikujiro. "Toward middle-up-down management: accelerating information creation." *Sloan Management Review* 9 (Spring 1988): 9-18.

Nonaka, Ikujiro. "Redundant, overlapping organization: A Japanese approach to managing the innovation process." *California Management Review* (Spring 1990): 27-38.

Nonaka, Ikujiro. "The vitality of agile companies (Agile company de katsuryoku)." *Nihon Keizai Shimbun*, April 22nd 1996, 18.

Nonaka, Ikujiro, and Hirotaka Takeuchi. *The Knowledge-Creating Company*. New York: Oxford University Press, 1995.

Normann, R. "Organizational innovativeness: product variation and reorientation." *Administrative Science Quarterly* 16 (1971): 203-215.

Odagiri, Hiroyuki. Growth through Competition, Competition through Growth: Strategic Management and the Economy in Japan. Oxford: Oxford University Press, 1992.

Odaka, K. Japanese Management: A Forward Looking Analysis. Tokyo: Asian Productivity Organization, 1986.

Okuzumi, Hisao. "Taisei Corporation plans for the Year 2000." Long Range Planning 23 (1 1990): 53-65.

Olsen, M. E. The Process of Social Organization. New York, NY: Holt Rinehart & Winston, 1978.

Oppenheim, A. N. Questionnaire Design and Attitude Measurement. London: Heinemann, 1966.

Osgood, Charles E., George J. Suci, and Percy H. Tannenbaum. *The Measurement of Meaning*. Urbana, IL: University of Illinois Press, 1957. Ouchi, William G. and A. M. Jaeger. "Type Z organization: Stability in the midst of mobility." *Academy of Management Review* (3 1978): 305-314.

Ouchi, William G. Theory Z: How American Business Can Meet the Japanese Challenge. New York, NY: Avon Books, 1981.

Ouchi, William G. The M-Form Society: How American Teamwork Can Recapture the Competitive Edge. New York, NY: Avon Books, 1984.

Oyama, Michihiro. "Nihongata keiei no kozai (Advantages and disadvantages of Japanese management)." *Nihon Keizai*, June 26th—July 2nd 1992,

Palmer, Donald A., P. Devereaux Jennings, and Xuenguang Zhou. "Late adoption of the multidivisional form by large U. S. corporations: Institutional, political, and economic accounts." *Administrative Science Quarterly* (38 1993): 100-131.

Parkinson, C. Northcote. The Fur-lined Mousetrap. London: Leviathan House

Parkinson, C. Northcote. Parkinson' Law, or the Pursuit of Progress. London: Murray, 1958.

Parkinson, C. Northcote. *Parkinson: The Law*. Boston, MA: Houghton Mifflin, 1980.

Parthasarthy, Raghavan and S. Prakash Sethi. "Relating strategy and structure to flexible automation: a test of fit and performance implications." *Strategic Management Journal* 14 (1993): 529-549.

Pascale, Richard T. and Mary Ann Maguire. "Comparison of selected work factors in Japan and the United States." *Human Relations* 33 (7 1980): 433-455.

Pascale, Richard T. and A. G. Athos. *The Art of Japanese Management*. New York: Simon & Schuster, 1981. Pascale, Richard T. "The Honda effect." In *The Strategy Process: Concepts, Contexts, and Cases,* eds. James Brian Quinn, Henry Mintzberg, and Robert M. James. 105-113. Englewood Cliffs, NJ: Prentice-Hall International, 1988.

Pennings, Johannes M. "Structural contingecy theory: a reappraisal." *Research in Organizational Behavior* 14 (1992): 267-309.

Penrose, Edith T. The Theory of the Growth of the Firm. Second ed., Oxford: Basil Blackwell, 1980.

Perrow, Charles. "A framework for the comparative analysis of organizations." *American Sociological Review* 32 (1967): 194-208.

Perrow, Charles. "Markets, Hierarchies and Hegemony." In Perspectives on Organization Design and Behavior, eds. Andrew H. Van de Ven and William F. Joyce. 371-386. 1981.

Peters, Thomas J. and Robert H. Waterman. In Search of Excellence: Lessons from America's Best-Run Companies. New York, NY: Harper & Row, 1982.

Peterson, P. B. "Fighting for a better Navy: an attempt at Scientific Management (1905-1912)." Journal of Management 16, no. 1 (1990): 151-166.

Porter, Michael E. "Competition in global industries: A conceptual framework." In *Competition in Global Industries*, ed. Michael E. Porter. 15-60. Boston, MA: Harvard Business School, 1986.

Prendergast, Canice. "Career development and specific human capital collection." *Journal of the Japanese and International Economies* 6 (1992): 207-227.

Pucik, Vladimir and Nina Hatvany. "Management practices in Japan and their impact on business strategy." In *The Strategy Process: Concepts, Contexts and Cases*, eds. James Brian Quinn, Henry Mintzberg, and Robert M. James. 351-363. Englewood Cliffs, NJ: Prentice-Hall International, 1988.

Pucik, Vladimir, Mitsuyo Hanada, and George Field. Management Culture and the Effectiveness of Local Executives in Japanese-Owned US Corporations. New York, NY: Egon Zehnder International, 1989.

Pugh, D. S., D. J. Hickson, C. R. Hinings, and C. Turner. "Dimensions of organization structure." Administrative Science Quarterly 14 (1968): 65-105.

Pugh, D. S., D. J. Hickson, C. R. Hinnings, and C. C. Turner. "The context of organization structure." *Administrative Science Quarterly*, 14 (1969): 115-126.

Pugh, D. S. "Introduction." In Organization Theory: Selected Readings, ed. D. S. Pugh. Second ed., Vol. Harmondsworth: Penguin Books, 1984.

Quinn, James Brian. Strategies for Change: Logical Incrementalism. Homewood, IL: Irwin, 1980.

Quinn, James Brian. "Managing innovation: controlled chaos." Harvard Business Review, May-June (May-June 1985): 73-84.

Randall, R. "Influence of environmental support and policy space on organisational behavior." *Administrative Science Quarterly* 18 (1973): 236-248.

Reich, R. B. The Work of Nations: Preparing Ourselves for 21st Century Capitalism. New York: Alfred A. Knopf, 1991.

Reichheld, Frederick F. The Loyalty Effect: the Hidden Force behind Growth, Profits, and lasting Value. Boston, MA: Harvard Business School Press, 1996.

Reitsperger, Wolf D. "Japanese management: Coping with British industrial relations." *Journal of Management Studies* 23 (1 1986): 72-87.

Rice, A. K. Productivity and Social Organization. London: Tavistock, 1958.

Roethlisberger, Fritz J. and William J. Dickson. *Management and the Worker*. Cambridge, MA: Harvard University Press, 1939.

Rohlen, T. P. For Harmony and Strength: Japanese White-collar Organization in Anthropological Perspective. Berkeley, CA: University of California Press, 1974.

Rohlen, Thomas P. "The company work group." In *Modern Japanese* Organization and Decision-Making, ed. Ezra F. Vogel. 185-209. Tokyo: Charles E. Tuttle, 1979.

Rumelt, Richard P. "Diversification strategy and profitability." *Strategic Management Journal* 3 (1982): 359-369.

Sakuma, Masaru. "A proposal for transferring a Japanese management system overseas: Applying the information-sharing system approach in the UK." In Japanese and European Management: Their International Adaptability, eds. Kazuo Shibagaki, Malcolm Trevor, and Tetsuo Abo. Tokyo: The University of Tokyo Press, 1989.

Sasaki, Toru. "How the Japanese accelerated new car development." Long Range Planning 24 (1 1991): 15-25.

Schatzman, Leonard and Anselm L. Strauss. Field Research: Strategies for a Natural Sociology. Englewood Cliffs, NJ: Prentice-Hall International, 1973.

Schein, Edgar H. "Does Japanese management style have a message for American managers?" In *Planning Strategies that Work*, ed. Arnoldo C. Hax. 72-90. New York, NY: Oxford University Press, 1987. Schendel, Dan and Charles W. Hofer. Strategic Management: A New View of Business Policy and Planning. Boston, MA: Little Brown, 1979.

Schollhammer, H. "Organization structures of multinational corporations." Academy of Management Journal (September 1971): 345-365.

Schreyögg, Georg and Horst Steinmann. "Strategic control: A new perspective." Academy of Management Review 12 (1 1987): 91-103.

Scott, William G. "Organization theory: an overview and an appraisal." Academy of Management Journal, no. 4 (1961): 7-26.

Seiler, John A. A Systems Analysis in Organizational Behavior. Homewood, IL: Richard D. Irwin, 1967.

Shibakawa, Rinya. "Management strategies of Japanese companies." In *Innovations in Management*, eds. Yasuhiro Monden, Rinya Shibakawa, Satoru Takayanagi, and Teruya Nagao. 13-22. Atlanta, GA: Industrial Engineering and Management Press, 1985.

Shimada, Haruo. "Nihongata humanware gijutsu (Japanese-style humanware technology)." In *Nihon Kigyo no System* (Japanese Enterprise System), eds. Hiroyuki Itami, Tadao Kagono, and Motoshige Itoh. 126-143. 3. Tokyo: Yuhikaku, 1993.

Shimizu, Ryuhei. Top Management in Japanese Firms. Tokyo: Chikura Shobo, 1986.

Shirai, Taishiro. "Decision-making in Japanese labor unions." In Modern Japanese Organization and Decision-Making, ed. Ezra F. Vogel. 167-184. Tokyo: Charles E. Tuttle, 1979.

Siew, Meng Leong and Tiong Tan Chin. "Managing across borders: An empirical test of the Bartlett and Ghoshal (1989) organizational typology." *Journal of International Business Studies*, (Third Quarter 1993): 449-464.

Simons, Robert. "Strategic orientation and top management attention to control systems." *Strategic Management Journal* 12 (1991): 49-62.

Smith, Peter B. "Organizational behaviour and national cultures." British Journal of Management 3 (1992): 39-51.

Smothers, Norman P. "Patterns of Japanese strategy: strategic combinations of strategies." *Strategic Management Journal* 11 (1990): 521-533.

Snodgrass, Carl. "Strategic control and cultural homogeneity: An empirical investigation of Japanese strategic control systems." In the Association of Japanese Business Studies Annual Meeting in Philadelphia, PA, 1982.

Snow, Charles C. and Donald C. Hambrick. "Measuring organizational strategies: Some theoretical and methodological problems." *Academy of Management Review* 5 (4 1980): 527-538.

Spender, J-C. "Some frontier activities around strategy theorizing." *Journal of Management Studies* 30 (1 1993): 11-30.

Stalk, George, Jr. and Alan M. Webber. "Japan's dark side of time." *Harvard Business Review* (July-August 1993): 93-102.

Stein, Barry A. and Rosabeth Moss Kanter. "Building the parallel organization: Creating mechanisms for permanent quality of work life." *The Journal of Applied Behavioral Science* 16 (3 1980): 371-388.

Steiner, George A. "The critical role of top management in long-range planning." In *Readings in Business Policy*, ed. Edmund R. Gray. 119-130. New York, NY: Appleton-Century-Crofts, 1968.

Stone, Eugene F. Research Methods in Organizational Behavior. Santa Monica, CA: Goodyear Publishing Company, 1978. Sullivan, Jeremiah. "A critique of theory Z." Academy of Management Review 8 (1983): 132-142.

Sullivan, Jeremiah and Richard B. Peterson. "A test of theories underlying the Japanese lifetime employment system." *Journal of International Business Studies* (First Quarter 1991): 79-97.

Suzuki, Y. "The strategy and structure of top 100 Japanese industrial enterprises 1950-1970." *Strategic Management Journal* 1 (1980): 265-291.

Sykes, A.J.M. "Economic interest and the Hawthorne researches." *Human Relations* 18 (1965): 253-263.

Tabachnick, Barbara G. and Linda S. Fidell. Using Multivariate Statistics. New York: Harper & Row, Publishers, 1983.

Tachibanaki, Toshiaki. "Labor mobility and job tenure." In *The Economic Analysis of the Japanese Firm*, ed. Masahiko Aoki. 77-102. Amsterdam: Elsevier Science Publishers, 1984.

Takahashi, K. "Matsushita Electric." : Case # 9-481-146, Harvard Business School 1981.

Takahashi, Nobuo and S. Takayanagi. "Decision procedure models and empirical research: The Japanese experience." *Human Relations* 38 (8 1985): 767-780.

Takahashi, Nobuo. "Sequential analysis of organization design: A model and a case of Japanese firms." *European Journal of Operational Research* 36 (1988): 297-310.

Takamiya, Susumu. What is Modern Management All About? Theory and Practice. Tokyo: Management-sha, 1980.

Takamiya, Susumu and Keith Thurley, eds. Japan's Emerging Multinationals: An International Comparison of Policies and Practices. Tokyo: The University of Tokyo Press, 1985. Takeuchi, Hirotaka, Kiyonori Sakakibara, Tadao Kagono, Akihiro Okumura, and Ikujiro Nonaka. *Kigyo no Jiko Kakushin—Chaos to Sozo no Management* (Self-Renewal of Enterprise—The Management of Chaos and Creativity). Tokyo: Chuo Koron, 1986.

Taylor, Frederick. "The Principles of Scientific Management." In *The Great Writings in Management and Organizational Behavior*, edited by Louis E. Boone and Donald D. Bowen. New York: McGraw-Hill, 1987.

Teece, David J. "Economies of scope and the scope of the enterprise." Journal of Economic Behaviour and Organisation 1 (1980): 223-247.

Teece, David J. "Contributions and impediments of economic analysis to the study of strategic management." In *Perspective on Strategic*, ed. James W. Fredrickson. 39-80. New York, NY: Harper & Row Publishers, 1990.

Thompson, James. Organizations in Action. New York: McGraw-Hill, 1967.

Tomita, Teruhiko. *Relevance of intra-organizational transferability of Japanese-style management*. Faculty of Economics, Shiga University, 1990.

Trevor, Malcolm, Jochen Schendel, and Bernhard Wilpert. The Japanese Management Development System. London: Frances Pinter, 1986.

Trevor, Malcolm. Toshiba's New British Company: Competitiveness through Innovation in Industry. London: Policy Studies Institute, 1988.

Trist, E. L., G. W. Higgin, H. Murray, and A. B. Pollock. *Organizational Choice*. London: Tavistock, 1963.

Tsuda, Masumi. *Nihonteki Keiei no Jinji Senryaku* (The Personnel Strategy of Japanese Management). Tokyo: Dobunkan, 1987.

Tsurumi, Y. "Productivity: The Japanese approach." *Pacific Basin Quarterly* 6 (1981): 7-11.

Tushman, Michael L. and David A. Nadler. "Information processing as an integrating concept in organisational design." Academy of ManagementReview (3 1978): 613-624.

Tushman, Michael L. "Technical communication in research and development laboratories: Impact of project work characteristics." *Academy of Management Journal* 21 (1978b): 624-645.

Tushman, Michael L. "Work characteristics and subunit communication structure: A contingency analysis." *Administrative Science Quarterly* 24 (1979): 82-97.

Ueno, Susumu and Uma Sakaran. "The influence of culture on budget control practices in the USA and Japan: an empirical study." *Journal of International Business Studies* (Fourth Quarter 1992): 659-674.

Urwick, Lyndall F. *The Theory of Organization*. New York: American Management Association, 1952.

Van de Ven, Andrew H., Andre Delbecq, and Richard Koenig Jr. "Determinants of coordination modes within organizations." *American Sociological Review* 41 (April 1976): 322-338.

van Wolferen, Karel. The Enigma of Japanese Power. London: Macmillan, 1989.
Vancil, Richard F. "Implementing strategy: the role of top management." : Harvard Business School, 1979.

Vancil, Richard F. "General Electric Comapny: background note on management systems 1981." : Harvard Business School, 1981.

von Bertralanffy, L. General System Theory: Foundations, Development, and Applications. New York: Braziller, 1967.

Weber, Max. The Theory of Social and Economic Organization. Edited by Talcott Parsons. New York: The Free Press, 1975

Weick, Karl E. "Educational organizations as loosely coupled systems." Administrative Science Quarterly 21 (March 1976): 1-19.

Weick, Karl E. "Cognitive process in organization." In Research in Organizational Behavior, ed. B. M. Staw. 41-74. 1. Greenwich, Conn.: JAI Press, 1979.

Wheelwright, S. C. "Japan - Where operations really are strategic." *Harvard Business Review* 59 (July-August 1981): 67-74.

Whitley, Richard. Firms, Markets and Societies. London: Sage Publication, 1992.

Womack, James P., Daniel T. Jones, and Daniel Roos. *The Machine that Changed the World*. New York, NY: Harper Collins Publishers, 1990.

Womack, James P. and Daniel T. Jones. "From lean production to the lean enterprise." *Harvard Business Review* (March-April 1994): 93-103.

Woodward, J. Management and Technology. London: HMSO, 1958.

Woodward, J. Industrial Organizations: Theory and Practice. London: Oxford University Press, 1965.

Wooldridge, Bill and Steven W. Floyd. "Strategic process effects on consensus." *Strategic Management Journal* 10 (1989): 295-302.

Wren, Daniel. The Evolution of Management Thought. New York: Wiley, 1979.

Yang, Charles Y. "Management styles: American vis-à-vis Japanese." Columbia Journal of World Business 31 (23 1977): 23-31.

Yoshihara, Hideki, Akimitsu Sakuma, Hiroyuki Itami, and Tadao Kagono. *Nihon Kigyo no Takakuka Senryaku: Keiei Shigen Approach* (The Diversification Strategy of Japanese Companies: A Managerial Resource Approach). Tokyo: Nihon Keizai Shinbun, 1981.

Yoshino, Michael Y. Japan's Managerial System: Tradition and Innovation. Cambridge, MA: The MIT Press, 1968.

Yoshino, Michael Y. "Emerging Japanese multinational enterprises." In Modern Japanese Organization and Decision-Making, ed. Ezra F. Vogel. 146-166. Tokyo: Charles E. Tuttle, 1979.

Young, David. "Brief case: Headquarters staff—Products of history or sources of distinctive skills?" *Long Range Planning* 26 (5 1993): 139-141.

Young, David and Michael Goold. *Effective Headquarters Staff: A guide* to the Size, Structure and Role of Corporate Headquarters Staff. London: Ashridge Strategic Management Centre, 1993. Young, David. The Headquarters Fact Book. London: Ashridge Strategic Management Centre, 1993.

Young, David. "Corporate Headquarters: A comparison of Germany, Japan and the UK." . London: Ashridge Strategic Management Centre, 1995.

Zimmerman, Mark. How to do business with the Japanese. Tokyo: Charles E. Tuttle, 1987.

Annex 1a:

Covering Letter for the Questionnaire

. . . .

Manchester, 23rd November, 1993

Dear Sirs,

We at the Manchester Business School belonging to the University of Manchester are conducting a survey about the corporate centre.

The objective of the survey is to learn about the role of the corporate centre and its relationship with management style in Japanese firms. We would be grateful if you could fill in the enclosed questionnaire.

All responses will be treated as strictly confidential. The analysis of the responses will be presented in the aggregate and in a codified form. As a consequence, it will be impossible to identify individual responses.

If possible, please return this questionnaire with:

(1) a copy of organisational chart showing basic structure of your firm, and

(2) a copy of the most recent annual report.

Thank you in advance for your co-operation.

Yours faithfully

Kimio Kase Doctoral Programme Manchester Business School

Annex 1b:

Covering Letter for the Questionnaire (Japanese)

1993年11月23日

前略

現在、マンチェスター大学経営大学院であるマンチェスター・ビジネス・ス クールでは企業の本部組織(つまり、本社ジェネラル・スタッフ部門または 本社管理部門)についての調査を行っています。

調査の目的は、日本企業において、本部組織の役割と、企業の経営スタイル とがいかに関連しているかを調べることにあります。質問調査票を同封いた しますので、ご回答願えれば幸いです。

いただいたご回答はコンフィデンシャルな情報として扱われることは勿論、 いただいたご回答のデータの分析は、個々の企業のデーターとしてではなく、 産業別、規模別などによる傾向を示すデーターとしてコード化して行われま す。したがい、個々の回答企業の名前は、分からないようになっております。

ご回答をいただく際、下記の資料もご送付いただければ幸甚です。

(1) 貴社の組織図

(2) 一番最近の財務諸表

よろしくお願いいたします。

草々

マンチェスター大学経営大学院博士課程 (マンチェスター・ビジネス・スクール)

加瀬公夫

Annex 2a:

Questionnaire

.

THE CORPORATE CENTRE IN JAPANESE MANAGEMENT

Company name:

Respondent's name:

Title:

Name of contact for queries (if different from above): Telephone number: Fax number:

1 **General Information**

(1) which of the following most closely describes the nature of your company's business? (Multiple answers are allowed)

Please circle all significant activities (more than 25 percent of turnover or assets). If there are other such activities, please complete section (i).

		•	
		<u>i</u>	Others: please state
đ	Machinery and parts	h	Miscellaneous manufacturing
С	Metal products	g	Motor vehicles
Ъ	Chemicals and petroleum products	f	Shipbuilding and rolling stock
a	Foods and beverages	e	Communications and electronics

(2)Please inform the following data

Turnover in last financial year

Total number of full-time employees

Total expenditure on corporate functions an (if available)

nd	management

÷

•	

2 **Environment Factors**

(1) How stable is the environment surrounding your firm (in consideration, for example, of the market competitiveness, frequency of new products, entrance barrier, etc.)? Please circle the number which best represent the situation at your firm.

extremely stable			midpoint			extremely unstable
1	2	3	4	5	6	7

369

(2) How diversified is your firm? (Please consider it not diversified, if related products or product lines sharing the same common market or technology represent over 95% of the total turnover.) Please circle the number which best represent the situation at your firm.

extremely diversified			midpoint			not diversified
						at all
1	2	· 3	4	5	6	7

(3) What degree of routineness have the production technologies of your firm.? Please consider them to be less routine, if custom technology is the principal production technology; highly routine, if continuous process technology (*eg*, petroleum refinery) is the principal production technology. Please circle the number which best represent the situation at your firm.

extremely			midpoint			extremely
high rou-			. –			low rou-
tineness						tineness
1	2	3	4	5	6	7

3 Organisational Structure and Systems

- (1) Is your firm's organisation divided into autonomous business units on geographic and/or product bases (*ie*, geographic divisions, product divisions, etc)?
 - 1 YES—Please answer question (2) and (3) below.
 - 2 No —Please proceed to question 4
- (2) What kinds of function are performed by typical divisions? Please circle the numbers of functions that are performed by these business units or divisions. (*multiple answers are allowed*)

1	production	6	finance
2	sales	7	basic research
3	marketing planning	8	applied research and development
4 5	personnel control	9	purchasing

(3) How many divisions are identified in your firm? Please restrict your calculation to division that have some degree of autonomy in operating decisions. Please circle the appropriate number.

1 2-5 divisions

more than 30 divisions

- 2 6 10 divisions
- 3 11 20 divisions
- 4 21 30 divisions

5

4 Corporate Centre (General staff and headquarters administrative departments)

(1) To what extent is the corporate centre¹ of the firm involved in the decisions and strategies of the divisions or functional departments²? circle the number which best represents the situation at your firm.

Please

extremely involved			midpoint			no involve- ment at all
1	2	3	4	5	6	7

(2) To what extent does the corporate centre provide a strategy and direction as guidance for the divisions or functional departments? Please circle the number which best represents the situation at your firm.

a very great deal of strategy and direction			midpoint			no strategy and direction at all
1	2	3	4	5	6	7

(3) How important is formal strategic planning system for the relationships between the corporate centre, and divisions or functional departments? Please circle the number which best represent the situation at your firm.

extremely important			midpoint			not important in the least
1	2	3	4	5	6	7

(4) Are major investment ideas that affect the division or functional departments initiated by corporate centre level?

1 YES—Please answer the next question.

2 No —Please proceed to question (5).

Please circle the number which best represent the situation at your firm.

100% of cases	over 75% but less than	over 50% but less than 75%	over 25% but less than 50%	over 0% but less than 25%
	100% of cases	of cases	of cases	
1	2	3	4	5

¹ General staff and headquarters administrative departments

² For example, production department, sales department, etc.

(5) How important are strategy criteria in corporate centre's approval of large capital requests? Please circle the number which best represent the situation at your firm.

extremely			midpoint			not
important						important
						in the
1	2	3	4	5	6	7

(6) How much importance do financial returns have in corporate centre's approval of large capital requests? Please circle the number which best represent the situation at your firm.

a very great deal of impor-			Midpoint			no impor- tance at all
tance 1	2	3	4	5	6	7

(7) How important are non-financial and strategic objectives set by the corporate centre for the division or functional departments at the beginning of each year? Please circle the number which best represent the situation at your firm.

extremely			midpoint			not
mportant						in the
1	2	3	4	5	6	Teast 7

(8) How much importance is placed on financial targets as opposed to strategic objectives in performance reviews of division or functional department managers? Please circle the number which best represent the situation at your firm.

a very great deal of impor-			midpoint			no impor- tance at all
tance						
1	2	3	4	5	6	7

(9) How important are financial targets agreed between divisions/functional departments and corporate centre for division or department head? Please circle the number which best represent the situation at your firm.

extremely			cannot			not
important			say one	· .	÷	important
			way or			in the
			the other			least
1	2	· 3	4	5	6	7

(10) Do financial targets agreed between divisions/functional departments and corporate centre amount to a promise by division/functional departments that has to be met? Please circle the number which best represent the situation at your firm.

definitely			cannot			definitely
true			say one			incorrect
			way or			
			the other			
	2	3	4	5	6	7

(11) How intensely is the division's or functional department's financial performance monitored by the corporate centre during the year? Please circle the number which best represent the situation at your firm.

extremely			cannot			not
intensely			say one			monitored
monitored			way or			in the
			the other			least
1	2	3	4	5	6	7

(12) What happens to a division/functional department manager who fails to achieve his or her annual financial target? Is the division/functional department manager in severe danger of jeopardising his career? Please circle the number which best represent the situation at your firm.

definitely true			cannot say one way or			definitely incorrect
1	2	3	the other 4	5	6	7

5 Linkages: Please circle the number which best represents the situation at your firm.

(1) How important are common customers for divisions/functional departments? Please circle the number which best represent the situation at your firm.

extremely important			midpoint			not important in the eleast
1	2	3	4	5	6	7

(2) How important is the purchase of products from each other have for divisions/functional departments? Please circle the number which best represent the situation at your firm.

extremely important			midpoint			not important in the
						least
1	2	3	· 4	5	6	7

(3) Do divisions/functional departments have a common production technology? Please circle the number which best represent the situation at your firm.

definitely			cannot			definitely
true			say one			incorrect
			way or the other			
1	2	3	4	5	6	7

(4) Do divisions/functional departments share an essential marketing skill? Please circle the number which best represent the situation at your firm.

definitely true			cannot say one			definitely
uue			way or			Inconcet
1	2	3	une otner 4	5	6	7

(5) Do divisions/functional departments share an essential product development skill? If neither extreme represents your judgement, circle an appropriate number in between, considering the distance for the description at both extremes, and, if any, at midpoint 3.

definitely			cannot			definitely
true			say one			incorrect
			way or			1
	!		the other			1
1	2	3 ·	4	5	6	7

6 Scope and Manning of the Corporate Centre: Please inform the number of the corporate staff. Please also circle the number of Please circle the number if corporate function. If so, please inform the number of the corporate staff.

10

11

12

Total Number of Corporate Personnel: Please state.

Please circle the number, if corporate function.

- 1 Finance, Accounting
- Secretarial and Legal $\mathbf{2}$ 3
- Personnel, Human Resources
 - Government, Public Relations 13
- 4 5 Strategic/Business Planning
- 6 Marketing/Commercial Services
- 7 Internal Audit
- 8
- Information Systems 9 Management Services

- **Research and Development**
- **Engineering Services**
- Production, Manufacturing,
- Physical Distribution Services
- Purchasing Property Services
- 14 Administrative Services
- 15 Other Services: Please state. 16

Annex 2b:

Ì

Questionnaire (Japanese)

日本的経営における本部組織について

記入者御氏名:	
役職:	
電話番号: ファックフ発品・	
プラックス書々。 ご回答に関する質問問合先(上記と異なる場合のみご記入ください):	

1 貴社に関する一般的データ

(1) 以下の記述のうち、どれが責社の業種に該当しますか。該当すると思われるもので、売上または資産の25%を代表する業種全てに〇印を付けてください。(いくつでも)

a	食品、飲料	е	通信、電子機器
b	化学、石油製品	f	造船、車両
С	金属	g	自動車、運搬機器
d	機械	h	その他製造業
•		i	その他。枠の中にご記述ください。

(2) 以下のデータについてご記入ください。

ー番最近の事業年度における売上高	. (年度)
正規従業員総数		
一般管理費		· ·

2 環境要因

(1) 貴社の事業環境はどの程度安定的ですか(例えば、市場の競争性、新製品の頻度、参入障壁度など)。 両極に示した記述を参考にしながら、貴社の現状にもっとも妥当すると思われる数字に〇印をご記 入ください。

きわめて安定的 である			どちらともいえ ない			きわめて不安定 である
1	2 ·	3	4	· 5	6	7

(2) 貴社の経営はどの程度多角化されていますか。(市場、技術等を共有する関連製品の売上構成に占める割合が95%以上であれば多角化されていないとして)両極に示した記述を参考にしながら貴社の現状にもっとも妥当すると思われる数字に〇印をご記入ください。

きわめて多角化 の度合が高い			中程度			まったく多角化 されていない
1	2	3	4	5	6	7

(3) もし個別受注生産の割合が高ければ生産のルーティン度が低い、反対に連続的な装置生産(例えば 石油精製)の割合が高ければ生産のルーティン度が高いとして、両極に示した記述を参考にしなが ら、貴社の現状にもっとも妥当すると思われる数字に〇印をご記入ください。

きわめてルーテ ィン度が高い			中程度		•	きわめてルーテ ィン度が低い
1	2	3	4	5	6	7

3 組織構造およびシステム

(1) 貴社の組織は製品別あるいは地域別等の基準にもとづき、事業ユニット(たとえば製品事業部、地域事業部)に細分されていますか。

1 はい 一下の問(2)~(3)にお答えください。 2 いいえ 一問4にお進みください。

(2) 事業部はどのような職能をもっていますか。以下の職能のうち、事業部によって遂行されている職能に〇印をご記入ください。(いくつでも)

1	生産	6	財務
2	販売	7	基礎研究
3	マーケティング企画	8	応用研究と開発
4	本人	9	購買
5	会計・コントロール		

(3) どの程度の数の事業部に分けられていますか。業務運営上の決定に関して自立性をもっている事業 部の数をお答えください。

 1 2~5
 4 21~30

 2 6~10
 5 30以上

 3 11~20

4

本部組織(一般に本社ジェネラル・スタッフ部門、本社管理部門と呼ばれるもの)

(1) 本部組織^{注1}は事業部または職能部門^{注2}の決定および戦略にどの程度関与していますか。両極に示した記述を参考にしながら、責社の現状にもっとも妥当すると思われる数字に○印をご記入ください。

きわめて多くの 関与			どちらともいえ ない			まったく関与し ない
_ 1 <u>_</u>	2	3	4	5	6	7

(2) 本部組織は事業部または職能部門に対してどの程度の戦略および方向づけを与えていますか。両極に示した記述を参考にしながら、責社の現状にもっとも妥当すると思われる数字に〇印をご記入ください。

きわめて多く与 えている			どちらともいえ ない			まったく与えな い
1	2	3	4	5	6	7

(3) 公式的な戦略計画システム(例えば、長期事業計画)は、本部組織と事業部または職能部門との関係においてどの程度の重要性をもっていますか。両極に示した記述を参考にしながら、貴社の現状にもっとも妥当すると思われる数字に〇印をご記入ください。

きわめて多くの 重要性をもつ			どちらともいえ ない			まったく重要性 をもたない
1	2	3	4	5	6	7

一般に本社ジェネラル・スタッフ、本社監理部門と呼ばれるもの。
 御 例えば、生産部、販売部など。

(4) 事業部または職能部門に影響をあたえる重要な投資案件は本部組織により提案されますか。

- 1 はい 一下の問にお答えください。
- 2 いいえ 一問(5)にお進みください。

貴社の現状にもっとも妥当すると思われる数字に○印を記入してお答えください。

すべて(100%とも)	75%以上100%未	50%以上75%未満	25%以上50%未満	0%以上25%未満が
本部提案	満が本部提案	が本部提案	が本部提案	本部提案
1	2	3	4	5

(5) 多額の投資案に対する本部組織の承認において、戦略的な判断基準は、どの程度の重要性をもっていますか。両極に示した記述を参考にしながら、貴社の現状にもっとも妥当すると思われる数字に 〇印をご記入ください。

きわめて多くの 重要性をもつ			どちらともいえ ない			まったく重要性 をもたない
1 .	2	3	. 4	5	6	7

(6) 多額の投資案に対する本部組織の承認において、財務的収益率(例えば、ROI、ROA、ROE、 ROS)はどの程度の重要性をもっていますか。両極に示した記述を参考にしながら、貴社の現状 にもっとも妥当すると思われる数字に〇印をご記入ください。

きわめて多くの 重要性をもつ			どちらともいえ ない			まったく重要性 をもたない
1	2	3	. 4	5	6	7

(7) 毎年度年初に本部組織により事業部または職能部門に与えられる、財務的な数字以外の、戦略的な 目標はどの程度の重要性をもっていますか。両極に示した記述を参考にしながら、貴社の現状にも っとも妥当すると思われる数字に〇印をご記入ください。

きわめて多くの 重要性をもつ			どちらともいえ ない			まったく重要性 をもたない
1	2	3	4	5	6	7

(8) 事業部または職能部門責任者の人事考課において、財務的目標達成度は、どの程度の重要性をもっていますか。両極に示した記述を参考にしながら、貴社の現状にもっとも妥当すると思われる数字に○印をご記入ください。

きわめて多くの 重要性をもつ			どちらともいえ ない			まったく重要性 をもたない
1	2	3	4	5	6	7

(9) 事業部または職能部門と本部組織との間で合意された財務的数字は、事業部または職能部門責任者にとり、どの程度の重要性をもっていますか。両極に示した記述を参考にしながら、責社の現状にもっとも妥当すると思われる数字に○印をご記入ください。

きわめて多くの 重要性をもつ			どちらともいえ ない			まったく重要性 をもたない
1	2	3	4	5	6	7

(10)事業部または職能部門と本部組織との間で合意された財務的数字目標は事業部または職能部門により、守らなければならない約束とみなされますか。両極に示した記述を参考にしながら、貴社の現状にもっとも妥当すると思われる数字に○印をご記入ください。

まったくそのと おり			どちらともいえ ない			まったくちがう
1	2	3	4	5	6	7

(11)事業部または職能部門の業績の本部組織による監督は、どの程度綿密にされますか。両極に示した 記述を参考にしながら、貴社の現状にもっとも妥当すると思われる数字に○印を記入してお答えく ださい。

きわめて綿密に 業績監督される			中程度			まったく業績監 督されない
1	2	3	4	5	- 6	7

(12)事業部または職能部門が年度予算を達成しない場合、部門責任者の交代が行われますか。両極に示した記述を参考にしながら、責社の現状にもっとも妥当すると思われる数字に○印をご記入ください。

まったくそのと おり			どちらともいえ ない			まったくちがう
. 1	2	3	4	5	6	7

5 部門または事業部間のつながり:両極に示した記述を参考にしながら、貴社の現状にも っとも妥当すると思われる数字に〇印をご記入ください。

⁽¹⁾ 部門または事業部間で多くの顧客を共有していますか。両極に示した記述を参考にしなが ら、責社の現状にもっとも妥当すると思われる数字に○印をご記入ください

きわめて多く共 有			中程度		v .	まったく共有し ない
1	2	3	4	5	6	7

(2) 部門または事業部間での製品または部品の相互取り引きの程度はどれくらいですか。両極に示した 記述を参考にしながら、貴社の現状にもっとも妥当すると思われる数字に〇印をご記入ください

きわめて多くの 相互取り引きを 行っている			中程度			まったく相互取 り引きを行って いない
1	2	3	4	5	6	7

(3) 部門または事業部間で、どの程度、生産技術を共有していますか。両極に示した記述を参考にしな がら、貴社の現状にもっとも妥当すると思われる数字に〇印をご記入ください

きわめて多く共 有			中程度			まったく共有し ない
1	2	3	4	5	6	7

(4) 部門または事業部間で、基本的なマーケティング・スキルを共有していますか。両極に示した記述 を参考にしながら、貴社の現状にもっとも妥当すると思われる数字に〇印をご記入ください

きわめて多く共 有			中程度			まったく共有し ない
1	2	3	4	5	6	7

(5) 部門または事業部間で、基礎的製品開発能力を共有していますか。両極に示した記述を参考にしな がら、貴社の現状にもっとも妥当すると思われる数字に○印をご記入ください

きわめて多く共 有		· · ·	中程度		•	まったく共有し ない
_ 1	2	3	4	5	6	7

6

本部組織(本社スタッフ部門)の役割および人員構成:

本部組織人員総数:枠の中にご記入ください。

次に述べる職能が本部組織に属する場合には、〇印をつけてください。

1	財務・経理	1 (0	マネジメント・サービス(企業内コ ンサルタント等)
2	.総務•法務	1	1	研究開発
3	人 事/ ヒューマンリソース	1 2	2 ·	エンジニアリング・サービス
4	涉外	1 :	3	製造、生産、流通サービス
5	広報	1 4	4	購買
6	戦略計画	1 !	5	不動産関連サービス(資産管理、不 動産の法的管理、等)
7	マーケティング/販売関連サービス	1 (6	事務サービス(電話、郵便、食堂、 医療、等)
8	内部監査			
9	情報システム	1	7	その他サービス:枠のなかにご記入く ださい。

382